

Vol 14 | No 1 | SPRING 2016

web site: http://www.asecu.gr ISSN 1109-8597 ISSN On line: 1792-3115



FUNDING INSTITUTION UNIVERSITY OF MACEDONIA THESSALONIKI, GREECE

# SEEJE

Editorial Office: ASECU, University of Macedonia 156 Egnatia Str., 546 36 Thessaloniki, Greece tel: +30 2310 891 793, fax: +30 2310 891 748 e-mail: asecu@uom.edu.gr web site: http://www.asecu.gr

# © ASECU

ISSN 1109-8597 ISSN On line: 1792-3115 Thessaloniki, Spring 2016

# contents

# Articles

in the Transition Economies of the European Union	
YILMAZ BAYAR	7
Political Risks of Western Companies in a Transforming Country. Evidence from Russia 2014 EWA MARTYNA-DAVID	21
<b>The Dynamics of China's Export Growth: An Intertemporal Analysis</b> FRANCIS TUAN, AGAPI SOMWARU, SUN LING WANG, EFTHIMIA TSAKIRIDOU	37
<b>A Review of Online Crowdsourcing Platforms</b> EVANGELOS MOURELATOS, MANOLIS TZAGARAKIS, EFTHALIA DIMARA	59
<b>Turkish Non-Core Bank Liabilities</b> ENGİN YILMAZ, BORA SÜSLÜ	75
Book Reviews	
<i>Economic and Policy Foundations for Growth in South East Europe:</i> <i>Remarking the Balkan Economy</i> edited by Adam Benett, G. Russel Kincaid, Peter Sanfey, Max Watson reviewed by GEORGE MAKRIS	93
Should Britain Leave the EU? An Economic Analysis of a Troubled Relationship edited by Patrick Minford, Sakshi Gupta, Vo Phuong Mai Le, Vidya Mahambare and Yongdeng Xu reviewed by CHRISTOPHER TSOUKIS	97
<i>Full Industry Equilibrium</i> <i>A Theory of the Industrial Long Run</i> by Arrigo Opocher and Ian Steedman reviewed by LEFTERIS TSOULFIDIS	103

# aim and scope of ASECU

ASECU was founded in 1996 as Association of South-Eastern Europe Economic Universities with the general aim of promoting the interests of those economic universities in South-Eastern Europe which are public, recognized or financed by the state of origin.

By decision taken in General Assembly of 2007, it was modified in Association of Economic Universities of South and Eastern Europe and the Black Sea Region. Presently, Universities and Research Centres are included to the active full members of ASECU from Albania, Armenia, Bulgaria, Bosnia-Herzegovina, FYROM, Greece, Montenegro, Poland, Romania, Russia, Slovakia, Serbia, Turkey and Ukraine. Also, as associated members, Universities form Egypt and Lebanon.

The specific aims of the Association are:

To promote cooperation between Economic Universities, Faculties, Departments; i.e., especially: a) to exchange views and information about syllabi, b) to exchange undergraduate and postgraduate students and c) to exchange teaching and research staff.

To provide members with the opportunity exchange information, opinions etc. by publishing a relevant scientific journal or by cooperation in elaborating scientific studies in relation to the future development of higher education and research as well as to improve their quality in the field of economic studies and business administration.

To undertake initiatives for the protection of the interests of members and their institutions, so as to be supported by international organizations and in particular by the higher education institutions of the European Union.

To encourage cooperation between universities inside and outside the countries referred to in the Association.

To pursue cooperation in the field of higher education with the consolidation of close relations with other organisations having similar aims, e.g. E.U.A.; To provide opportunities for harmonising the degrees of faculties and departments of the universities participating in the Association;

To promote cooperation between economic universities, faculties, departments in the field of research for the benefit of the economy, the society, peace and the cultural development of the countries referred to the Association.

# Presidency and Board Members

The following Board was elected for a four years period, during the 6th General Assembly of ASECU, on the 6th of March 2015.

**Bozidar Cerović**, President University of Belgrade, Belgrade, Serbia

**Grigorios Zarotiadis,** Vice President Aristotle University of Thessaloniki, Greece

Atanas Damyanov, General Secretary University "D. Tsenov", Svishtov, Bulgaria

Vesna Karadzić, Member University of Montenegro, Podgorica, Montenegro

**Fatmir Memaj,** Member University of Tirana, Tirana, Albania

**Dejan Mikerević,** Member University of Banja Luka, Bosnia- Herzegovina

Leonid Nakov, Member "Ss Cyril and Methodious" University, Skopje, FYROM

**Dorel Paraschiv**, Member Bucharest University of Economic Studies, Romania

**Bogdan Wierzbinski**, Member University of Rzeszów, Rzeszów, Poland



# IMPACT OF OPENNESS AND ECONOMIC FREEDOM ON ECONOMIC GROWTH IN THE TRANSITION ECONOMIES OF THE EUROPEAN UNION

# YILMAZ BAYAR\* Usak University

#### Abstract

The globalization process has accelerated, particularly as of the 1980s, and countries began to remove obstacles on the flows of goods, services and capital. Hence, substantial increases have resulted in both global trade volume and crossborder capital flows. Moreover, countries have improved their institutional and legal infrastructure to achieve sustainable economic growth and attract foreign capital. This study examines the impact of openness and economic freedom on the economic growth of the transition economies in the European Union during the 1996-2012 period, through the use of panel data analysis. We have found there is a long-run relationship among the variables and both economic freedom and trade openness have a positive impact on economic growth, while financial openness has a negative impact on economic growth.

JEL Classification: C32, F43, O10, O40 Key Words: Openness, Economic Freedom, Economic Growth, Panel Data Analysis

<sup>\*</sup>*Corresponding Address:* Yilmaz Bayar, Usak University, Department of Economics, 1 Eylul Campus, Usak-Turkey. e-mail address: yilmaz.bayar@usak.edu.tr

# 1. Introduction

Countries began to remove constraints on the movement of goods, services and capital with the increasing globalization after the Second World War. Trade and financial liberalization contributed to increasing world trade volume and cross-border capital flows. World trade volume, as a percentage of GDP, increased from 25.62% in 1960 to about 60% in 2013 (World Bank, 2015a). Furthermore, cross-border capital flows increased to about 20% of the world GDP in 2007, but then decreased to 5% of the world GDP in 2012 (James *et al.*, 2014).

Transition economies of the European Union (EU) belatedly integrated into the liberalization process in the world after the collapse of Communism. They implemented transition from centrally planned economies to market economies as of the late 1980s and then joined the EU in the 2000s. During this transition process, the countries liberalized their trade, integrated into global financial markets and improved their quality of institutional infrastructure through structural reforms.

In this regard, new growth theories have emphasized that openness and institutional quality have had a positive impact on economic growth (See McKinnon (1973), Shaw (1973), Bencivenga and Smith (1991), King and Levine (1993), Fedderke (2002), Andersen and Babula (2008), Hye and Lau (2015)). This study investigated the impact of openness and economic freedom on the economic growth in transition economies of the EU. Our empirical findings also verified that increases in the level of both institutional quality and trade openness raised economic growth. However, we found that financial openness had a negative impact on economic growth and we came to the conclusion that this can result from the fact that the financial sector of these countries has not reached the necessary threshold level during their development process.

The rest of the paper is organized as follows: The next section presents an overview of the theoretical and empirical literature on our topic. Section 3 presents data and the econometric methodology; section 4 conducts the empirical analysis and presents major findings. Finally, the study is completed with the Conclusion.

## 2. Literature Review

There has been a wide range of theoretical and empirical studies on the impact of trade openness, financial openness and economic freedom on economic growth in relevant literature. The literature review led us to select the method and variables used in the study. We also found that there have been extensive empirical studies on the topic of this article, but there have been few studies specifically on the transition economies of the EU. Therefore, this study will bridge the gap of current extensive literature by examining the impact of openness and economic freedom on the economic growth in this group of countries. Finally, we are going to use econometric tests, which consider structural breaks contrary to most empirical studies in the litera-

ture, because traditional econometric tests with no structural breaks may yield biased results in cases of structural breaks.

#### 2.1. Trade Openness and Economic Growth

Trade openness is one of the important variables of sustainable economic growth in the globalized world. Endogenous growth theories provide a theoretical basis for the relationship between trade openness and economic growth. In the context of these theories, trade openness possibly has an impact on economic growth via knowledge spillovers, capital accumulation, and factor price equalization (See Romer (1990) and Hye and Lau (2015)).

Extensive empirical studies have been conducted so as to determine the impact of trade openness on economic growth and they have reached mixed findings on the relationship between two variables. Some studies, such as those by Marelli and Signorelli (2011), Sakyi *et al* (2012), Mercan *et al* (2013), Zakaria and Ahmed (2013) and Razmi and Refaei (2013), found that trade openness has a positive impact on economic growth, while some studies, such as those by Menyah *et al* (2014) and Ulasan (2015), have found that trade openness has no significant impact on economic growth. On the other hand, other studies, such as those by Kim (2011) and Hye and Lau (2015), found that the relationship between trade openness and economic growth could be different depending on the level and duration of development.

#### 2.2. Financial Openness and Economic Growth

There are two major theoretical views on the relationship between financial openness and economic growth. One view states that financial openness affects economic growth positively by efficiently allocating resources and providing better access to foreign capital, by improving risk sharing and contributing to the stabilization of the economy and through the development of the financial sector. The other view supports that the benefits of financial openness may not be realized or may be realized in a limited manner and that financial openness makes a national economy more vulnerable to crises (Kim *et al*, 2014).

Empirical studies on the relationship between financial openness and economic growth have also reached mixed findings. Some studies, such as those by Bekaert *et al* (2005), Ranciere *et al*. (2006), Garita (2009), Levchenko *et al*. (2009) and Kim *et al*. (2014), have found that financial openness has had a positive impact on economic growth, while relatively few studies, such as those by Gine and Townsend (2004), Fratzscher and Bussiere (2004), Tswamuno *et al*. (2007), have found that financial openness has had a negative or no significant impact on economic growth. Furthermore, some studies have investigated the causality between financial openness and economic growth. Other studies, such as those by Yapraklı (2007), found that there was unidirectional causality from financial openness to economic growth, while other

studies, such as those by Kar and Pentecos (2000), Bas-Dinar *et al.* (2015), found that there was unidirectional causality from economic growth to financial openness.

# 2.3. Economic Freedom and Economic Growth

Economic freedom may be accepted as a quality indicator of institutions and a legal structure which countries have. Institutional and legal structure becomes very important both for creating an investment environment and, also, for attracting foreign investment and capital in a globalized world. Discussions on economic freedom go back to Adam Smith, but the concept of economic freedom has different meanings depending on various economic theories and approaches. In this study, we are using the economic freedom index calculated by The Heritage Foundation. This index is based on four pillars, namely, the rule of law (property rights, freedom from corruption), limited government (fiscal freedom, government spending), regulatory efficiency (business freedom, labour freedom, monetary freedom) and open markets (trade freedom, investment freedom, financial freedom) (The Heritage Foundation, 2015).

There have been a large number of empirical studies on the relationship between economic freedom and economic growth, especially in the last two decades. Most of the studies have found that economic freedom has generally had a positive impact on economic growth (See Nelson and Singh (1998), Gwartney *et al* (2004), Yun-Peng and Tuan-Yuen (2009), Paakkonen (2010), Peev and Mueller (2012), Piątek *et al* (2013), Razmi and Refaei (2013) and Akıncı *et al* (2014)).

#### 3. Data and Econometric Methodology

We examined the long run relationship between economic growth, openness and economic freedom in transition economies of the EU during the 1996-2012 period using the Basher and Westerlund (2009) cointegration test.

#### *3.1. Data*

In the study we have used the real GDP per capita growth as a proxy for economic growth (dependent variable). We also used the sum of export and import as a percentage of the GDP as a proxy for trade openness and Chinn-Ito index (KAOPEN) as a proxy for financial openness and economic freedom index, as calculated by The Heritage Foundation (2015). The data of economic growth and trade openness were obtained from the World Bank (2015a & 2015b), the data of financial openness from Chinn and Ito (2015) and the data of economic freedom from The Heritage Foundation (2015). Our sample and study period were dictated by data availability. Variables used in the econometric analysis and their symbols are presented in Table 1.

Variable	Symbol	Source
Real GDP per capita growth	GROWTH	World Bank (2015b)
Trade openness (export and import of	TRAOP	World Bank (2015a)
goods and services as a percentage of		
GDP)		
Financial openness	FINOP	Chinn and Ito (2015)
Economic freedom index	EFR	The Heritage Foundation
		(2015)

#### Table 1. Variables Used in the Study

E-views 8.0, WinRATS Pro. 8.0 and Gauss 11.0 software packages were used for analyses in the study.

# 3.2. Econometric Methodology

In this study, we investigated the impact of trade openness, financial openness and economic freedom on economic growth in transition economies of the EU. Firstly we tested cross-sectional dependence with the bias-adjusted LM (Lagrange Multiplier) test of Pesaran *et al* (2008) and conducted the stationarity testing of the series using the PANKPSS (Panel Kwiatkowski, Phillips, Schmidt and Shin) test by Carrion-i-Silvestre *et al* (2005). Then we analyzed long run relationships between variables using the Basher and Westerlund (2009) method and cointegrating coefficients were estimated with the panel Augmented Mean Group (AMG) analysis by Eberhardt and Bond (2009).

# 3.2.1. Cross-sectional Dependence Test

Cross-sectional dependence of variables is very important for determining further econometric tests used in the study. Therefore, we should test whether there is cross-sectional dependence in the series and the cointegrating equation. When the time dimension of the panel is higher than the cross-section dimension, the Breusch-Pagan (1980) LM test was used. Otherwise, the cross-section dependence (CD) LM test by Pesaran (2004) was used. Later the *CD LM<sub>adj</sub>* test was developed by Pesaran *et al* (2008) through adding variance and mean to test statistics in order to adjust its bias, because the *CD LM<sub>adj</sub>* test yields biased results when the group mean is zero and the individual mean is not zero. The *CD LM<sub>adj</sub>* test statistics developed by Pesaran *et al* (2008) is calculated as follows:

$$CD \ LM_{adj} = \left(\frac{2}{N(N-1)}\right)^{1/2} \sum_{i=1}^{N-1} \sum_{j=i+1}^{N} \left[\hat{\rho}_{ij}^2 \left(\frac{(T-K-1)\hat{\rho}_{ij-\hat{\mu}_{Tij}}}{v_{Tij}}\right)\right] \sim N(0,1)$$
(1)

where  $\hat{\mu}_{Tij}\hat{\mu}_{Tij}$  represents the mean and  $v_{Tij}v_{Tij}$  represents the variance in equation 2. The test statistics from equation 2 exhibits an asymptotically standard normal distribution. The null hypothesis of the test is that there is no cross-sectional dependence, while the alternative hypothesis is that there is cross-sectional dependence.

#### 3.2.2. PANKPSS Unit Root Test

The PANKPSS unit root test developed by Carrion-i-Silvestre *et al* (2005) considers the cross-sectional dependence and the possible multiple structural breaks in the dataset. The test model is as follows:

$$Y_{it} = \alpha_{it} + \beta_{it} + \varepsilon_{it} \qquad i = 1, 2, ..., N \text{ and } t = 1, 2, ..., T$$
(2)  
$$\alpha_{it} = \sum_{k=1}^{m} (\theta_{ik} K 1_{it}) + \sum_{k=1}^{m} (\gamma_{ik} K 2_{it}) + \alpha_{it-1} + u_{it}$$
  
$$\beta_{it} = \sum_{k=1}^{n} (\varphi_{ik} K 1_{it}) + \sum_{k=1}^{n} (\delta_{ik} K 2_{it}) + \beta_{it-1} + v_{it}$$

where K1 and K2 are dummy variables defined as follows:

$$K1 = \begin{cases} 1 & t = T_B + 1 \\ 0 & other \ cases \end{cases} \qquad K2 = \begin{cases} 1 & t > T_B + 1 \\ 0 & other \ cases \end{cases}$$

where  $T_B$  represents the structural break point in equation 3 and it allows m structural breaks in the constant term and n structural breaks in the trend. The PANKPSS unit root test allows a maximum of 5 structural breaks. The null hypothesis of the test is that the series is stationary, while the alternative hypothesis of the test is that the series is not stationary.

# 3.2.3. Basher and Westerlund (2009) Cointegration Test

The Basher and Westerlund (2009) cointegration test considers cross-sectional dependence and multiple structural breaks and allows a maximum of three structural breaks, while testing the long run relationship among the variables. The test statistics of the model developed by Basher and Westerlund (2009) is as follows:

$$Z(M) = \frac{1}{N} \sum_{i=1}^{N} \sum_{j=1}^{M_i+1} \sum_{t=T_{ij-1}+1}^{T_{ij}} \left( \frac{S_{it}^2}{\left(T_{ij} - T_{ij-1}\right)^2 \hat{\sigma}_i^2} \right)$$
(3)

where  $S_{it} = \sum_{s=T_{ij-1}+1}^{t} \widehat{W}_{st}$  and  $\widehat{W}_{it}$  is a residual vector obtained from an efficient estimator; for example, fully modified least squares,  $\widehat{\sigma}_i^2$  is a variance estimator based on  $\widehat{W}_{it}$ . Test statistics exhibits a standard normal distribution. The null hypothesis of the test is that there is cointegration of variables for all cross-sections, while the alternative hypothesis is that there is no cointegration of variables for some of the cross-sections.

## 3.2.4. Panel Augmented Mean Group (AMG)

The Panel AMG method estimates cointegrating coefficients by considering crosssectional dependence; it also calculates the average group effect by weighting the overall panel results and individual coefficients. Therefore, it is more reliable than the common correlated effects method developed by Pesaran (2006) for estimating cointegrating coefficients (Eberhardt and Bond, 2009). Therefore, we estimated cointegrating coefficients with the panel AMG developed by Eberhardt and Bond (2009). In this estimation method variables are decomposed in the following manner:

$$y_{it} = \beta'_i x_{it} + u_{it}; \qquad u_{it} = u_{it} + \lambda'_t f_t + \varepsilon_{it}$$
(4)

$$x_{mit} = \pi_{mi} + \delta'_{mi}g_{mt} + \rho_{1mi}f_{1mt} + \dots + \rho_{nmi}f_{nmt} + v_{mit}$$
(5)

$$f_t = \varphi' f_{t-1} + \epsilon_{it} \text{ and } g_t = \aleph' g_{t-1} + w_t \tag{6}$$

where  $f_t$  represents unobservable common factors, while  $g_t$  represents country specific factors.

### 4. Empirical Analysis

#### 4.1. Cross-Sectional Dependence Test

We tested cross-sectional dependence of variables and the cointegrating equation using the  $CD \ LM_{adj}$  test by Pesaran *et al* (2008) and the results of the test are presented in Table 2. Results showed that the null hypothesis (cross-sectional independence) was rejected, because probability values are lower than 1%. In this case, the remaining countries were affected by a shock in one of the countries in the panel. Therefore, we should select econometric tests taking into consideration cross-sectional dependence.

Variable	Test Statistics	Probability
GROWTH	6.334	0.000
FINOP	4.268	0.019
TRAOP	3.782	0.004
EFR	5.331	0.000

Table 2	. Results	of Adjusted	CD	$LM_{ad}$	, Test
---------	-----------	-------------	----	-----------	--------

# 4.2. PANKPSS Unit Root Test

We used the PANKPSS unit root test to determine the stationarity of the variables in our study. We selected the model which allows for structural breaks in both the constant term and trend, when applying the test. Critical values were obtained by Monte Carlo simulations with 1,000 simulations. Results of the PANKPSS unit root test are presented in Table 3. These results indicate that variables were not stationary at their level, but became stationary after the first differencing. We also present the dates of structural breaks in Table 3 and results showed that the test determined structural breaks successfully, as well. In the dates of structural breaks, the Russian crisis, the global financial crisis and the Eurozone sovereign debt crisis, respectively, emerged in 1998, 2008 and 2009, during our study period.

Countries	DGR	OWTH	WTH DFINOP		DTRAOP		DEFR	
	<i>p</i> -	S. break	<i>p</i> -	S. break	<i>p</i> -	S. break	<i>p</i> -	S. break
	value		value		value		value	
Bulgaria	0.157*	1999	0.132*	1998	0.231*	1999	0.136	1998
		2008		2008		2009		2009
Croatia	0.162*	1998	0.238*	1998	0.134*	1999	0.182	1998
		2009		2009		2009		2009
Czech	0.216*	1998	0.261*	1999	0.119*	1999	0.109	1999
Republic		2009		2009		2009		2009
Estonia	0.194*	2008	0.205*	2009	0.226*	2009	0.231	2008
Hungary	0.289*	2009	0.266*	2008	0.137*	2009	0.226	2008
Latvia	0.215*	2008	0.392*	2009	0.141*	2009	0.248	2009
Lithuania	0.138*	2009	0.246	2008	0.102*	2009	0.159	2009
Romania	0.275*	1997,	0.207	1998	0.173*	1999	0.144	1998
		1998, 2009	*	2009		2008		2009
Slovak	0.119*	2009	0.178*	2008	0.168*	2009	0.152	2009
Republic				2009				
Slovenia	0.185*	2009	0.108	2009	0.217*	2009	0.180	2009
Panel	0.235*		0.268*		0.195*		0.163	

# Table 3. Results of PANKPSS Unit Root Test

\*Stationary at 5% significance level

# 4.3. Basher and Westerlund (2009) Cointegration Test

We used the Basher and Westerlund (2009) cointegration test for testing long run relationship between variables. We selected the model that allows structural breaks in both the constant term and trend for the cointegration test and results are presented in Table 4. Critical values were obtained by Monte Carlo simulations with 1,000 simulations. Results showed there was a cointegration relationship between variables when structural breaks were taken into consideration.

# Table 4. Results of Basher and Westerlund (2009) Cointegration Test

	Test Statistics	Probability	Decision
Exclusion of structural breaks in the constant term and trend	3.887	0.021	There is no cointegration
Consideration of structural breaks in the constant term and trend	34.678	0.349	There is cointegration

# 4.4. Estimation of Cointegrating Coefficients

We applied the panel AMG method for estimating cointegrating coefficients and the results are presented in Table 5. Autocorrelation and heteroscedasticity problems were eliminated using the Newey-West method. Findings demonstrated that trade openness and economic freedom have a positive impact on economic growth at 95%, while financial openness has a negative impact on economic growth at 95%.

Variables	riables Coefficient	
FINOP	-0.214	0.015*
TRAOP	0.326	0.002*
EFR	0.289	0.017*

Table 5. Results of Panel AMG Estimation

\*Statistically significant at 5% level

The positive impact of trade openness and economic freedom on economic growth is supported by endogenous growth theories (See Acemoglu *et al* (2004), Hye and Lau (2015)) and the findings of most empirical studies in the literature. However, the negative impact of financial openness on economic growth is not consistent with the propositions of endogenous growth theories. Kim *et al* (2014) asserted that

the impact of financial openness on economic growth depends on country specific factors, including the level of economic development, macroeconomic development and stability. Consequently, our findings could have arisen due to underdeveloped financial structures and institutional structure as well as the insufficient and unstable economic performance of transition countries.

Short run relationships between variables were estimated by the panel AMG and results are presented in Table 6. We found that the coefficients of error correction terms were negative and statistically significant. This demonstrated that deviations among series in the short run were eliminated and series converged to their long run equilibrium values. This finding also verified that our variables were cointegrated. On the other hand, the small coefficients of error correction terms showed that the equilibrating velocity of variables was low.

Variables	Coefficient	Prob.	Coefficient of Error Correction Terms
FINOP	-0.193*	0.031	-0.083*
TRAOP	0.294*	0.004	-0.107*
EFR	0.286*	0.007	-0.091*

 Table 6. Short Run Analysis

\*Statistically significant at 5% level

# 5. Conclusion

The transition economies of the EU transited from centrally planned economies to market economies concurrently with the fall of the Berlin Wall in 1989 and then these countries were integrated into the EU. During this process, these countries liberalized their economies and improved the quality of their institutional infrastructure. This study examined the impact of openness and economic freedom on economic growth in the transition economies of the European Union, including Bulgaria, Croatia, the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Poland, Romania, Slovakia and Slovenia during the 1996-2012 period. Our findings indicate that trade openness and economic freedom had a positive impact on economic growth in the long run, while financial openness had a negative impact on economic growth in the long run. The propositions of endogenous growth theories and empirical studies in the literature support the positive relationship between economic growth, trade openness and economic freedom.

The quality of the institutions has an impact on the arrangement of economic institutions, which, in turn, affects economic growth by channelling the allocation of resources into the economy. So the findings of the study verified that institutional

quality is one of the important components behind economic growth, as proposed by new endogenous growth theories. On the other hand, trade openness may also affect economic growth through various channels, such as technological spillovers and increasing productivity. In this regard, the economic performances of our sample were positively affected by trade openness. Finally, our expectations had been that financial openness would have a positive impact on economic growth, considering the prevailing positive relationship between financial development and economic growth in the extensive theoretical and empirical literature. However, we found that there was a negative relationship between economic growth and financial openness and we concluded that this may have resulted from the fact that the financial sectors of these countries have not sufficiently developed to make a positive contribution to economic growth.

The findings of the study imply that trade openness and economic freedom foster economic growth, while financial openness slows it down. Therefore, it is important for less developed countries that they improve their institutional quality and liberalize their trade gradually. Institutional quality is a prerequisite condition for long run growth, but the impact of trade liberalization on economic growth depends on the ability of countries to adjust and endogenize technological spillovers and to improve their productivity and competitiveness. Finally, countries can benefit from financial openness, if they have adequate financial infrastructure. Therefore, it is possible that economic growth was negatively affected by financial liberalization during its first stages. Further empirical studies can be conducted to determine the channels through which openness affects economic growth; this will be useful for policymakers so as to boost the impact of such factors in regard to economic growth.

#### References

- Acemoglu, D., Johnson, S., Robinson, J., 2004, Institutions as the Fundamental Cause of Long-run Growth, Retrieved from http://www.nber.org/papers/w10481.pdf (17.02.2015).
- Akıncı, M., Yüce, G., Yılmaz, Ö., 2014, "Ekonomik Özgürlüklerin İktisadi Büyüme Üzerindeki Etkileri: Bir Panel Veri Analizi", *Anadolu Üniversitesi Sosyal Bilimler Dergisi*, 14(2), 81-96.
- Andersen, L., Babula, R., 2008, The Link between Openness and Long-run Economic Growth, Retrieved from http://www.usitc.gov/publications/332/journals/openness\_growth\_link.pdf (15.02.2015).
- Bas-Dinar, G., Cakar-Dalgic, B., Varol-İyidogan, P. 2015, "Financial Liberalization and Economic Growth in Turkey: A Reexamination", *Hacettepe University Journal of Economics and Administrative Sciences*, 33(1), 19-43.
- Basher, S.A., Westerlund, J., 2009, "Panel Cointegration and the Monetary Exchange Rate Model", *Economic Modelling*, 26(2), 506-513.
- Bekaert, G., Harvey, C.R., Lundblad, C., 2005, "Does Financial Liberalization Spur Growth?", *Journal of Financial Economics*, 77, 3-55.
- Bencivenga, V.R., Smith., B.D., 1991, "Financial Intermediation and Endogenous Growth", *Review of Economic Studies*, 58, 195-209.
- Breusch, T.S., Pagan, A.R., 1980, "The Lagrange Multiplier Test and Its Applications to Model Specification Tests in Econometrics", *Review of Economic Studies*, 47(1), 239-253.

- Carrion-i-Silvestre, J.L., del Barrio, T., López-Bazo, E., 2005, "Breaking the Panels: An Application to the GDP per capita", *The Econometrics Journal*, 8(2), 159–175.
- Chinn, M., Ito, H., 2015, The Chinn-Ito index A de jure Measure of Financial Openness –, Retrieved from http://web.pdx.edu/~ito/Chinn-Ito website.htm (10.02.2015)
- Eberhardt, M., Bond, S., 2009, Cross-section Dependence in Nonstationary Panel Models: A Novel Estimator, Retrieved from http://mpra.ub.uni-muenchen.de/17692/ (17.02.2015)
- Fedderke, J., 2002, "Technology, Human Capital, Growth and Institutional Development: Lessons from Endogenous Growth Theory?", *Theoria: A Journal of Social and Political Theory*, 100, 1-26.
- Fratzscher, M., Bussiere, M., 2004, "Financial Openness and Growth: Short-run Gain, Long-run pain", *ECB Working Paper*, No.348.
- Garita, G., 2009, How does Financial Openness Affect Economic Growth and Its Components? Retrieved from http://mpra.ub.uni-muenchen.de/20099/ (14.02.2015)
- Gine, X., Townsend, R.M., 2004, "Evaluation of Financial Liberalization: A General Equilibrium Model with Constrained Occupation Choice", *Journal of Development Economics*, 74(2), 269– 307.
- Gwartney, J., Holcombe, R.G., Lawson, R.A., 2004, "Economic Freedom, Institutional Quality and Cross-country Differences in Income and Growth", *Cato Journal*, 24(3), 205-233.
- Heritage Foundation, 2015, Index of Economic Freedom, Retrieved from http://www.heritage.org/ index/explore?view=by-region-country-year (10.02.2015)
- Hye, Q.M.A., Lau, W.Y., 2015, "Trade Openness and Economic Growth: Empirical Evidence from India", Journal of Business Economics and Management, 16(1), 188-205.
- James, E., McLoughlin, K., Rankin, E., 2014, Cross-border Capital Flows since the Global Financial Crisis, Retrieved from http://www.rba.gov.au/publications/bulletin/2014/jun/pdf/bu-0614-8. pdf (05.02.2015)
- Kar, M., Pentecos, E.J., 2000, "Financial Development and Economic Growth in Turkey: Further Evidence on the Causality Issue", Loughborough University Economic Research Paper No.00/27
- Kim, D.H., 2011, "Trade, Growth and Income", The Journal of International Trade & Economic Development: an International and Comparative Review, 20(5), 677–709.
- Kim, D.H., Lin, S.C., Suen, Y.B., 2014, "Dynamic Effects of Financial Openness on Economic Growth and Macroeconomic Uncertainty", *Emerging Markets Finance and Trade*, 48(1), 25-54.
- King, R.G., Levine, R., 1993, "Finance and Growth: Schumpter might be Right", *Quarterly Journal of Economics*, 108(3), 717-737.
- Levchenko, A.A., Ranciere, R., Thoenig, M., 2009, "Growth and Risk at the Industry Level: The Real Effects of Financial Liberalization", *Journal of Development Economics*, 89(2), 210–222.
- Marelli, E., Signorelli, M., 2011, "China and India: Openness, Trade and Effects on Economic Growth", *The European Journal of Comparative Economics*, 8(1), 129–154.
- McKinnon, R.I., 1973, *Money and Capital in Economic Development*, Washington D.C.: The Brookings Institution.
- Menyah, K., Nazlioglu, S, Wolde-Rufael, Y., 2014, "Financial Development, Trade Openness and Economic Growth in African Countries: New Insights from a Panel Causality Approach", *Economic Modelling*, 37, 386–394.
- Mercan, M., Gocer, I., Bulut, S., Dam, M., 2013. "The Effect of Openness on Economic Growth for BRIC-T Countries: Panel Data Analysis", *Eurasian Journal of Business and Economics*, 6(11), 1-14.
- Nelson, M.A., Singh, R.D., 1998, "Democracy, Economic Freedom, Fiscal Policy and Growth in LDCs: A Fresh Look", *Economic Development and Cultural Change*, 46(4), 677-696.
- Paakkonen, J., 2010, "Economic Freedom as Driver of Growth in Transition. *Economic Systems*, 34(4), 469–479.

- Peev, E., Mueller, D.C., 2012, "Democracy, Economic Freedom and Growth in Transition Economies", *Kyklos*, 65(3), 371–407.
- Pesaran, M. H., 2004, "General Diagnostic Tests for Cross Section Dependence in Panels", University of Cambridge, Faculty of Economics, *Cambridge Working Papers in Economics* No. 0435.
- Peseran, M.H., 2006, "A Simple Panel Unit Root Test in the Presence of Cross-section Dependecy", *Cambridge Working Papers in Economics* No. 0346.
- Pesaran, M.H., Ullah, A., Yamagata, T., 2008, "A Bias-adjusted LM Test of Error Cross-section Independence", *The Econometrics Journal*, 11(1), 105-127.
- Piątek, D., Szarzec, K., Pilc, M., 2013, "Economic Freedom, Democracy and Economic Growth: A Causal Investigation in Transition Countries", *Post-Communist Economies*, 25(3), 267-288.
- Ranciere, R., Tornell, A., Westermann, F., 2006, "Decomposing the Effects of Financial Liberalization: Crises vs. Growth", *NBER Working Paper*, 12806.
- Razmi, M.J., Refaei, R., 2013, "The Effect of Trade Openness and Economic Freedom on Economic Growth: The Case of Middle East and East Asian Countries", *International Journal of Economics* and Financial Issues, 3(2), 376-385.
- Romer, P. M., 1990, "The Problem of Development: A Conference of the Institute for the Study of Free Enterprise Systems", *Journal of Political Economy*, 98(1), 1–11.
- Sakyi, D., Villaverde, J., Maza, A., Chittedi, K.R., 2012, "Trade Openness, Growth and Development: Evidence from Heterogeneous Panel Cointegration Analysis for Middle-Income Countries", *Cuadernos de Economía*, 31(57), 21-40.
- Shaw, E.S., 1973, *Financial Deepening in Economic Development*, Cambridge: Harvard University Press.
- Tswamuno, D.T., Pardee, S., Wunnava, P.V., 2007, "Financial Liberalization and Economic Growth: Lessons from the South African Experience", *International Journal of Applied Economics*, 4(2), 75-89.
- Ulasan, B., 2015, "Trade Openness and Economic Growth: Panel Evidence", Applied Economics Letters, 22(2), 163-167.
- World Bank, 2015a, Trade (% of GDP), Retrieved from http://data.worldbank.org/indicator/ NE.TRD.GNFS.ZS (10.02.2015)
- World Bank, 2015b, GDP per capita growth (annual %). Retrieved from http://data.worldbank.org/ indicator/NY.GDP.PCAP.KD.ZG (10.02.2015)
- Yapraklı, S., 2007, "Ticari ve Finansal Dışa Açıklık ile Ekonomik Büyüme Arasındaki İlişki: Türkiye Üzerine Bir Uygulama", *Ekonometri ve İstatistik*, 5(1), 67-89.
- Yun-Peng, C., Tuan-Yuen, K., 2009, "The Impact of the Political Liberalization-Economic Freedom Nexus on Economic Growth, 1970-2000", *Applied Econometrics and International Development*, 9(2), 21-30.
- Zakaria, M., Ahmed, E., 2013, "Openness–Growth Nexus in Pakistan: A Macro–econometric Analysis", Argumenta Oeconomica, 1(30), 47-83.



# POLITICAL RISKS OF WESTERN COMPANIES IN A TRANSFORMING COUNTRY EVIDENCE FROM RUSSIA 2014

# EWA MARTYNA-DAVID\*

University of Applied Sciences BFI Vienna, Austria

#### Abstract

Foreign companies form an integral part of the economic landscape in their respective host countries. Western enterprises have to face specific risks and obstacles related to the current local democratic and economic development stage and they are confronted with political risks such as corruption, favouritism or legal uncertainty. This paper is based on the evidence collected during a field study conducted between March and September 2014 in Austria and Russia. The author conducted 23 qualitative interviews with Austrian business representatives operating in Russia and Russian experts concerning various aspects of the business environment. The results of this study are intended to be used for the improvement of the understanding of specific business conditions prevailing in Russia. This paper should be considered a valuable foundation for further in-depth research both on political risks and individual factors.

#### JEL Classification: FO 17, FP 48

**Key Words:** Business Environment, Coping Strategies, International Business, Political Risk, Russian Federation.

The first version of this paper was presented at the 7th International Conference "The Economies of Balkan and Eastern Europe Countries in the changed World" which took place in Kavala, Greece on 8 - 10 May 2015.

<sup>\*</sup>*Corresponding Addresses:* Ewa Martyna-David, University of Applied Sciences BFI Vienna, Austria, Competence Center Black Sea Region, Wohlmutstrasse 22, 1020 Vienna, Austria. e-mail: Ewa.Martyna-David@fh-vie.ac.at

# 1. Introduction

After the breakup of the Soviet Union, the Russian Federation, as its main successor, seemed for many scientists to be on a path towards Western-style democracy and free market economy. Two decades of transformation have shown that many informal institutions inherited from the Soviet period are still present in political and economic life. The transformation process, however, has changed their character and scale.

This article discusses the political risks faced by foreign businesses in Russia. My purpose is to demonstrate the impact of informal institutions embedded in the political-institutional framework of a host country on the business development of Western enterprises. The aim is to look for answers to the following research questions: What are the manners in which the Russian business environment can be characterized by neopatrimonial practices and how do Western enterprises react to it? The latter question concerns coping strategies introduced by them in order to adjust to Russian market conditions.

There is a wide range of studies on informal individual factors which characterize a neopatrimonial system, such as patron-client relationship, various forms of corruption or the absence of rule of law. The impact of informal institutions within such a system on international business has not been addressed yet. This article attempts to bridge this gap by offering a comprehensive multidisciplinary approach.

The first section presents a literature review and methodology. The second part consists of a detailed case analysis of the consequences of political risks for Western firms in Russia. The final part outlines the coping strategies adopted by these companies when trying to deal with the business environment they operate in. At the end of this paper, I will try to summarise the most relevant findings.

## 2. Conceptual framework: Political Risks and Neopatrimonialism

So far, no universally accepted definition of political risks in business administration has been established, let alone a list of indicators. Despite the proliferation of definitions and theoretical concepts, however, there is at least a common understanding. Agarwal and Feils (2007) describe political risk as the consequence of possible interference by external agents with business operations. The results of changes are risky only if they have a potentially profound effect on profits or other objectives of a particular enterprise (Robock, 1971: 7).

Political risk is omnipresent; it can occur both in high-industrialised as well as Less Developed Countries (LDCs). The distinction between them lies in the manifestation of risk factors and the level of institutional development. Typically, LDCs have less developed institutional frameworks, and, consequently, various informal institutions substitute formal institutions and impact business activities (Meyer & Peng, 2016: 5-6).

When searching for matching ideas which may be applicable for transition countries, one encounters the concept of neopatrimonialism, which is mainly used in political science. It analyses the coexistence of formal and informal institutions within a political system (Bratton & Van de Walle, 1997). Neopatrimonialism describes a type of system in which the tradition of patron-client networks is interrelated to some kind of rational-legal bureaucratic structure. Formal structures and rules do exist, even if the distinction between the private and public sphere is not always detectable in practice (Erdmann & Engel, 2007: 105). This situation creates uncertainty for stakeholders. Under a neopatrimonial organisation "people have a certain degree of choice as to which logic they want to employ to achieve their goals and to best realise their interests" (Erdmann & Engel, 2007: 105).

Informal institutions are "mainly unwritten, socially shared rules and procedures, which are put into social live, distributed and implemented "outside of officially sanctioned channels" (Helmke & Levitsky, 2004: 725). They might include corruption, personal networks, clientelism and patronalism (Helmke & Levitsky, 2004: 727-729).

Informal institutions exist regardless of properly or improperly functioning formal institutions. If informal institutions coexist with ineffective formal institutions, the former are supposed to substitute or compete with the latter (Helmke & Levitsky, 2004: 729).

Although the concept of neopatrimonialism is still underdeveloped and ill-defined due to lack of coherence, some factors, such as corruption, rent seeking, patron-client relationships and legal uncertainty are widely used to describe this phenomenon. These informal factors are embedded in the political system but, due to the interplay between politics and economics, they mainly tend to influence the latter.

This paper focuses on informal institutions in a neopatrimonial system and their consequent impact on the operational environment of Western companies. Corruption and favouritism based on clientelism and patronage can be regarded as two risk factors in a broader set of political risk factors with direct consequences on companies' performance.

Corruption is an umbrella term which covers various aspects, such as collusion, cronyism and nepotism, fraud, gifts, hospitality, lobbying, abuse of power or office and influence peddling (Shekshnia *et al.*, 2014: 5). This broad understanding can be found in the widely cited definition by Nye. According to this author, corruption is a behaviour that deviates from formal duties or rules of conduct governing the actions of an officeholder, because of their private motives, namely, wealth, power, or status. This pattern can be detected in various forms, i.e. bribery, nepotism and misappropriation. The first practice is always of a monetary nature, the last two may also have a non-monetary character (Nye, 1967: 419).

The definition has a very broad scope in terms of types of corruption and gains. For this paper, I have adopted a narrow understanding, which is limited to the monetary nature of corruption acts. Systemic corruption means that politicians and officials use their authority in order to sustain their private wealth.

Taking into account the importance of non-monetary aspects of relations within informal institutions, the term "systemic favouritism" is to be introduced and consequently applied. Systematic favouritism describes the use of public office to foster the interests of the patron himself and/or his clients, while impeding the interests of actors who are no part of this network.

# 3. Literature review

Economic and financial literature on political risks has been developing since the 1960s. Much of the research work has been dedicated to expounding on concepts and definitions.

A comprehensive model of two dimensions and thee factors of political risks has been developed over time. Robock (1971) commences by drawing a distinction between macro and micro political risks. While macro risks are typical for all firms in a host country, micro risks impact only a select group. The micro level is important for analysis due to the fact that what may be a political risk for one company may not be the same for another (Robock, 1971: 8). Haner (1979) added internal and external dimensions to this model (Alon & Martin, 1998: 12). Internal risks are caused by the host country while external ones are generated by the home country, the international environment or the global scene. Simon (1982) combined both approaches and divided the factors into society-related and government-related. Alon (1996) added a third factor: economy-related political risks. Simon (1994) extended the model of internal and external dimensions to include direct and indirect risks. Torre and Neckar developed a model of exposure to political risks based on two dimensions: loss as a result of governmental actions or actions caused by non-governmental actors and loss contingencies or value contingencies. These four types of risk exist at the micro and macro levels.

Jensen (2003 and 2008) has analysed the impact of political regimes on risks for international investors. He (2003) explored the casual link between democratic governance and foreign direct investment (FDI). Results show a positive relationship between democracy and a country's ability to attract foreign investors. In another article Jensen (2008) focused on the relations between political regimes and political risk and found out that democratic institutions tend to reduce risks for international investors. This happens especially through increasing constraints imposed on executives.

The character of risks has changed over time. Typical risks of the 1960s and the 1970s, such as expropriation of business assets, have vanished or have become a

minor problem for companies and new risks came to the forefront. According to Markwick, corruption belongs to the five important new risks, which have dominated the relevant scene since the 1990s (Marwick, 1998: 45). This was not a newcomer in the business world but there is evidence that the significance of this risk started increasing at that time.

Liuhto (cf. Liuhto 2010) analyses Russian political risks by applying Alan and Martin's macro risk model on Russia. He assesses developments in a firm's environment in the first decade of the 2000s. Liuhto came to the conclusion that the stability of the current political regime, the protracted instability in the Caucasus region, rising nationalism and the deteriorating opinion of the world concerning Russia, as well as the significant increase in unemployment rates in the region are key-determinants, which increase macro risks in Russia.

There are a few relatively recent publications that pay focus on the interrelationship between political and economic spheres in Russia viewed through a neopatrimonial prism (cf. Ledeneva, 2013a; Robinson, 2013). The concept of patrimonial capitalism applied by Robinson can be seen as a variant of Weber's patrimonialism and neo-patrimonialism in the economic sphere (Robinson, 2011: 4). Ledeneva analysed the nature, norms and methods of informal governance under Putin based on this concept (Ledeneva, 2013a: 1136).

If we take a broader look at relevant literature on the interplay between politics and economics in Russia, without direct reference to the neopatrimonial concept, we may observe different lines of research. Scientists elaborating on transition or area studies have noted the existence of a variety of informal practices, including the corruption of the public sector and the informal networks of the ruling elites.

Much has been written about one particular informal institution, namely, corruption. When we look at Russia, this phenomenon is widely discussed as a practice inherent in bureaucracy and the political system (Cheloukine *at al.*, 2007; Leslie, 2012; Orttung 2006; Rozov, 2013). Leslie claims that: "corruption has been, is and will continue for some time" to be an important problem in Russia. According to Cheloukhine, forms of corruption are "constantly changing and getting more sophisticated" (Cheloukine *at al.*, 2007: 53). As for the agents involved, Rozov explains that each of the three tiers of the corruption pyramid (i.e., politicians, bureaucrats and citizens/businessmen), has its own informal but relatively stable and highly respected corporate rules (Rozov, 2013: 47). When we look at geographic highlights, the most corrupted locations are Russia's major cities, due to the strong concentration of red tape there (Orttung, 2006: 2).

Corruption networks within political and economic elites have been another popular topic for research on informality and informal institutions (Hanson, 2013; Ledeneva, 2013). One of the pioneers in this area is Ledeneva. She describes how bribes are shared within power networks (Ledeneva, 2013). Her perspective is

based on in-depth interviews with insiders and enhanced by evidence of workings brought to light in court cases. According to her, there are four key types of networks, namely, the inner circle, useful friends, core contacts and more diffuse connections, functioning under Vladimir Putin's system of governance – referred to as *sistema*. Collection and redistribution of bribes is a key factor of this system (Ledeneva, 2013: 106). Hanson writes about increasing involvement of politicians in Russian business and points a finger to the extractive industry and high-tech sectors as core areas of close cooperation between senior officials and businesspeople (Hanson, 2011: 119).

There are also a few quantitative studies on informal practices related to corruption in domestic and foreign business in Russia (c.f. Fey & Shekshnia, 2011, Ledeneva & Shekshnia 2011). Ledeneva and Shekshnia (2011) asked 33 representatives of domestic and foreign companies to describe the extent to which their businesses were confronted with informal practices at the regional level. Based on the results, the authors categorised informal practices into five groups: 1. "Dinosaur" practices, i.e. ever or rarely used practices, which include extorting favours from job candidates or briberies to officials. 2. The most commonly used practices are tagged as "predator" practices, which describe, for instance, informal money flows between state and business and include voluntary or constrained payments related to fire inspections, customs tax or financial engagement in regional projects and programmes. 3. "Black cash" practices, which are in decline and include payments for a court decision, to a prosecutor or the police. 4. The term "rat" practice describes internal informal corporate practices. It means, *inter alia*, the use of the company's resources for personal gain. These take the form of kickbacks or the selection of services based on informal relationship. 5. There are also practices tagged as "Penguin" practices still applied. This term describes, for instance, practices based on life-long informal ties and networks used by companies to ensure state contracts or conflict of interest in the form of cronyism/nepotism. 6. Another potentially useful instrument may be "hook" practices, i.e. the use of a kompromat or other informal materials to exert pressure on competitors or officials. (Ledeneva & Shekshnia, 2011: 14-17).

There are also a few qualitative studies showing empirical evidence of private businesses coping with corruption. Kouznetsov and Dass have conducted a field research study of areas of corruption among Russian distributors of foreign goods (Kouznetsov & Dass, 2010). Their findings reveal that smaller distributors are more affected by corruption demands than large firms (Kouznetsov & Dass, 2010: 24). Another aspect was the focus of Karhunen and Kosonen. They looked at how a company rooted in a low-corruption home country can adopt domestic compliance standards in a corrupted host country. The authors found out that large companies can avoid corruption acts due to their financial and relational resources. SMEs, however, have to adapt to the corrupt environment (Karhunen & Kosonen, 2013: 88) and have also designed strategic responses to corruption as idealists, selective and

ceremonial conformists or pragmatists (Karhunen & Kosonen, 2013: 99). Having conducted a study of 36 foreign companies in Russia, Fey and Shekshnia (2011) worked out 8 recommendations for doing business in Russia. A piece of their advice focuses on corruption. The authors define some forms of bribes, such as contribution to a special project, bribes to judiciary and regulatory bodies, extortion from power ministries, or commissions to high state officials. Companies designed four strategies for dealing with corruption acts: outsourcing, pre-emption, abstention from corruption and postponement of entering a corrupt market.

## 3. Methodology

This paper presents the findings of a field study conducted by the author between March and September 2015 in Russia and Austria. The selection of a qualitative approach was made because of the sensitive character of the research problem. This method helped me to better investigate the "how" questions of my project.

Based on an extensive literature review I designed a questionnaire and, consequently, a classification system. The interviews were semi-structured and conducted on the basis of guidelines. The questionnaire provided me with valuable support for keeping to the structure. Expert interviews were selected as the data collection method. One to one interviews conducted in a confidential manner provided access to information that would not otherwise have been accessible.

The author contacted about 200 Austrian companies registered in Moscow and 10 scientists, journalists and NGO activists. Interview partners were chosen by targeted selection applying the position technique, while scientific respondents by applying the reputation technique. The selection criteria for both groups were their expertise and their position. All academic experts are renowned researchers; all but one business partner were of the executive level. This implies deep knowledge of and experience in all aspects of business operations in the host country. My aim was to get a broad field of expertise of various types of enterprises so as to be able to compare my findings with those of local experts from the academia.

The geographical focus on Moscow was dictated by the fact that the country strategies of foreign enterprises focus mostly on a few regions or regional centres. They settle their subsidiaries particularly in better developed regions of Russia, such as Moscow or Saint Petersburg. A rate of 40-50 % of FDIs is concentrated in these locations (Deuber, 2014: 13).

Representatives of Austrian businesses were invited to contribute to a project on the political-institutional framework for business activities. Russian scientists, journalists and representatives of NGOs were directly asked to share their relevant knowledge in their field of expertise. Both groups were approached with an offer of guaranteed strict confidentiality and were asked the same categories of questionnaire questions. Interviews were conducted orally and due to the confidentiality promised I have codified the names of the firms involved. Business interview partners represented a wide range of sectors and companies from SMEs to large enterprises. Eventually, I was granted access to enterprises active in high-tech, winter sports equipment, consulting, construction, mining, logistics, beverages, office equipment, agriculture and pharmaceuticals. This broad range enabled me to get a clearer picture of practical problems.

The response of Russian scientists, journalists and representatives of NGOs was high. The possible causes of the low rate of responses among Austrian companies may be explained by two factors: inopportune timing and thematic connotations. The project was launched at the time of the onset of the EU sanctions against Russia following the Crimea annexation. This may have increased the sensitivity perceived on possible answers to the research topic. Some companies may have interpreted the broadly formulated topic in the context of political corruption and were not willing to share their experiences in the light of official commitment to compliance standards. Another factor may have been the unclear political developments in regard to the sanctions eventually imposed by the EU and the Russian government's reactions.

The interviews were mostly conducted in person allowing for a broader conversation on topics of interest and helped the author gain deeper insights into the structure and workings of informal institutions in Russia. The language used was the choice of the interviewee from English, German or Russian. The 20 face-to-face interviews were supplemented by 3 telephone and Skype interviews. Each of them lasted one hour on average. Of the interviews 19 are recorded, while 4 respondents preferred not to be recorded; however, I obtained permission to take extensive notes.

After data collection, the interviews were transcribed in full in the respective original language. The classification system guided me through the analysis of the interview material. In this respect, I followed Mayring (2015). Furthermore, I summarised qualitative content analysis while applying the deductive approach. The reason to opt for this procedure was the plethora of literature already existing providing a wide range of variables and concepts which could be used as points of departure.

# 4. Political Risk and its Consequences for Western Business

#### 4.1 Political corruption

Corruption exists in many forms and it is committed at various levels in Russia. Due to the hidden nature of such transactions, it is very difficult to identify the structure and the final beneficiary of this practice. According to some Russian interviewees, the phenomenon is centrally conducted and the strings lie in the hands of President Putin (Interview 4, Interview 17). There is no consensus among them as to whether corruption has a pyramid form or as to the corruption money collected at each tier of the pyramid.

Some companies observe that since 2011-2012 the fight against corruption has made progress in Moscow, at least at the level of middle-sized companies (Interview 13, Interview 15, Interview 21). As a representative of a Winter Sports Supplier noted, "it is still a longer process, but the protests at Bolotnaya Square have contributed to it and the state is reading the signals".

Western firms encounter corruption mostly in state administration, where public officials demand things they should not (Interview 2, Interview 5, Interview 19, Interview 30). According to a Russian researcher, corruption structures operate in all state institutions (Interview 9). Each body has a certain sphere of specialisation. A representative of a Business Association stated that corruption is visible in some state regulatory bodies (at the senior official level) such as the finance police, the tax inspectorate, the customs service, the fire department, the labour inspectorate or the sanitary inspectorate (Interview 21).

As for the scale, the number of corruption transactions may be dropping but the range of the money trail grows, stated Russian researchers and the representative of an Austrian business (Interview 8, Interview 16, Interview 17). Corruption money also constitutes a kind of compensation: As salaries in public administration are quite high, it is a higher risk for an official to become involved in corrupt practices, explained the Russian interview partner. Companies are using this practice mainly for accelerating or delaying decisions or processes (Interview 2, Interview 5, Interview 21).

Official combat against corruption makes basic bribery more difficult. The old practice of an envelope containing money is not possible any more. This was noted by both Russian experts and representatives of Austrian businesses (Interview 9, Interview 13, Interview 15, Interview 17). The rate of corruption payments depends on the size of the contract and the type of service. The average rate of kickback is estimated at 30-40 percent of a transaction, confirmed a Russian researcher and an investigative journalist (Interview 9, Interview 11).

The incidence of such acts depends on the size of the project and the line of business. Bribing or kickbacks are still present in infrastructure projects, especially in state ordered projects (Interview 15). It is also used in branches which work with projects in general. Each one needs a permit from the state or from a licensed company (Interview 19).

A foreign company may encounter such a demand when it is awarded a state contract or a contract with companies close to the government. The chance of solicitation of bribes is bigger, when the company involved is a large one and when applying for a federal state project (Interview 6, Interview 15). Companies experienced corruption practices when interested in state procurement (Interview 4, Interview 9) and when dealing with public tenders (Interview 15, Interview 19, Interview 21, Interview 23, Interview 24). Corruption schemes are also applied in the areas of certification and licensing (Interview 5, Interview 13, Interview 19, Interview 21). The situation with public tenders has improved in Russia within the last ten years but winning at public tender remains quite a difficult task for a Western entrepreneur (Interview 21). The prevailing opinion is that compliance standards cannot be met at state tenders (Interview 21, Interview 13, Interview 22, Interview 23): there is no entry barrier, but a foreign company stands a much better chance when it is aware of the informal rules at play (Interview 19).

The situation has also improved in critical sectors like customs (Interview 21). There are still situations where customs services are delayed unnecessarily in order for the situation to be resolved by informal means (Interview 11, Interview 19). Operating companies have learnt how to obtain the willingness of Russian officials. The most frequently chosen practice is outsourcing this service to a broker who has the right contacts and knows the rules (Interview 2, Interview 5, Interview 13, Interview 19, Interview 20).

One of the informal practices is a bonus for the person who has mediated in the business deal (Interview 11, Interview 15). If a company wants to get a contract, it has to deliver a certain percentage for mediation services. However, this practice seems to be applied less and less (Interview 15).

# 4.2 Political favouritism

The general view among Russian experts was that there is a close linkage between political and business elites in the Russian model of economic development (Interview 4). As a result, one cannot always distinguish between the role of a politician or official and the role of a businessman. Informal patronage groups in Russia are dynamically developing and their composition changes over time (Interview 4, Interview 17). The core structure remains unchanged and entails politicians, representatives of big businesses (oligarchs), officials of law enforcement agencies (*siloviki*) and, sometimes, criminals (Interview 4). The underlying foundations of patronage relations are trust and loyalty (Interview 4). The binding element remains the President who plays the arbiter between various groups of interest (Interview 4, Interview 11).

Informal networks seem to be a minor problem for business activities of Western companies. There are a few companies that conduct projects with state companies, firms with government links or partners who are politically well-connected (Interview 10, Interview 18, Interview 19, Interview 20, Interview 23). Experience seems to suggest that one needs such networks to make business run more smoothly (Interview 19). Large companies need additionally political protection (Interview 4, Interview 17). As Russian researchers reported, if a domestic or foreign company wants to be competitive in its sector, it needs to be protected by political connections. The company pays a certain percentage for political protection, but has a guarantee that its work will not be disturbed (Interview 17). It is important to have connections in the right informal circle. A wrong decision will have negative consequences on one's business (Interview 4). According to an experienced entrepreneur, political contacts start at the United Russia party (Interview 13).

The Russian economy is based on a small number of large companies in which the state holds majority stakes. In big state projects, a foreign enterprise will be able to win such a project only with a strong Russian partner with political clout. This means that one needs to be in partnership with a key player (Interview 10, Interview 18). When a foreign investor wants to cooperate with companies close to the state, it is important to know a business partner who is well connected with the federal state apparatus (Interview 20).

There are very few cases, when an Austrian or German company has been threatened with an unfriendly takeover labelled as a raider attack (Interview 15, Interview 21). A representative of a business Association explained a Russian modus operandi: if a confronted company ignores a raider threat, the other party has so far refrained from further action. An attempt may happen in a sector where Russian oligarchs are operating (Interview 23). Enterprises offering good standards in production or trade belong to the high risk group (Interview 21). If an enterprise reaches a certain level of turnover or sales, it can also became the target of a raider attack (Interview 17, Interview 21).

Various options are applied, explained a Russian and an Austrian expert. A company may receive an offer to sell shares to a local, politically well-connected businessperson. In some instances, a Russian businessman becomes dissatisfied with a new Western competitor and falsifies the ownership documents of the foreign rival's company. Then an inspector may be sent, who starts to collect some informal information about the company of interest from insiders (Interview 4). Alternatively, a Western businessman may be forced to sign a document passing ownership over to the raider (Interview 15). Sometimes the raider does not stop short of using law enforcement agencies to fulfil his demands. The third option is to provide some criminal claim to the court. The judge is bribed to issue an illegal sentence which makes the foreign company pay a certain fine. This money is used to pay the law enforcement agencies to take the company over by physical force or use some sort of goods raiders (Interview 4).

#### 5. Coping Strategies of Western businesses

Western companies have developed some strategic options which help them minimise potential losses due to the institutional environment. The companies try to adapt to the local customs perceived in order to be successful.

Foreign businesspeople understand that running a business could be risky in Russia, but potentially brings large profits. Companies reduce some risks by seeking contacts with informal networks. A company can establish a rapport with high-ranking government representatives or look for support through business partners with political clout. It helps to outsource the corruption problem and obtain contracts. A company can also cooperate with a local consulting company, which has regular connections with the regional authorities and knows their demands and requirements.

Another strategic option in order to minimise the risk of raider attacks is to keep local exposure at a low level. This can be achieved through various measures. Instead of establishing a fully operating local subsidiary, the parent company only sets up a local representative branch which then outsources its activities to local firms or leaves core processes with the parent company in the home country. Another approach in this strategy would refer to property. In order to prevent exposure to expropriation, one has to avoid acquiring any property within the country. Instead, the firm rents its entire business infrastructure, from the building to the telephone line.

The second complex covers corruption. A first strategy is to avoid bribery completely. Some firms have given up doing business with public authorities and try to cooperate with independent "private" partners. They decided not to take part in public tenders or state procurement. Another option is to find a niche where foreign products are sought-after and local competition is low or to offer a high-tech product which cannot be provided by domestic companies or which is dependent on knowhow.

Another kind of corruption avoidance is strict maintenance of control over the sensitive areas of the company itself, like finances or accounting, by outsourcing them to external specialised companies. These companies report directly to their headquarters in Europe and thus give overall control to their regional manager.

Corruption, however, can also be outsourced. They can also shift responsibility for some business operations, such as sales and distribution, to the Russian partner or cooperate with independent local companies in these areas. In sectors where state permits, licenses or certificates are obligatory, a company can use a mediator to solve the problem.

# 6. Conclusions

Research on the effects of corruption and political favouritism in transition countries in the field of political science or of the emerging economies in business studies is still limited. Available data on corruption mostly concern quantitative information. The impact of informal networks on business is poorly analysed. This paper addressed a previously limited area of research in international business: how informal institutions embedded in the political system exercise their impact on the business activities of Western enterprises.

This article is intended to contribute to the development of knowledge concerning political risk in Russia. Empirical findings show the methods and schemes of corruption in a wide range of sectors and in companies of various sizes. The article also provides some insight into methods of operations related to the existence of political favourit-

ism in Moscow. Ledeneva and Shekshnia conducted a quantitative study on a similar topic, but treated Russia in general. They found out that the old practices of "black cash", such as cash bribes or defalcation, have decreased and have been replaced by more subtle practices, such as supporting pet regional projects or selling companies' assets under their real value to the selected companies. My research confirmed that nowadays bribery is more sophisticated and that supporting "important" regional projects or selling companies' shares under their real value to state officials seems to play no role for Western companies in Moscow. Additionally, it shows that corruption acts are not always committed in person, as there are specialised intermediaries.

As the authors mentioned above gained rather generalized results, the intent of this research project was to generate more concrete findings focused on one specific region. The results could serve as a starting point for further research. It would be particularly interesting to explore differences in exposure to political risk in particular sectors or for companies of various sizes. Several issues may emerge from these findings and, consequently, require additional work. Further exploration could provide important insights into the accessibility of local networks to Western businesses, corruption schemes in particular sectors or depending on a company's size or the use of legal intermediaries for problem solving purposes.

Empirical data were collected in a selected location in Russia (Moscow), which can justifiably be described as a large metropolis with a high concentration of foreign investors. This business environment encountered by foreign companies cannot be generalised for the whole country. Political risks in other regions would certainly make an interesting contribution. Furthermore, practitioners interested in entering the Russian market could also benefit from these findings. They could be warned of the role of informal institutions and develop coping strategies more easily or refrain from entering the market.

## References

- Agarwal J. and Feils D., 2007, "Political risk and the internationalization of firms: an empirical study of Canadian-based export and FDI firms", *Canadian Journal of Administrative Sciences*, 24, 165–181.
- Alon, I. and Herbert, T. T., 2009, "A stranger in a strange land: Micro political risk and the multinational firm", *Business Horizons*, 52(2), 127-137.
- Alon, I. and Martin, M. A., 1998, "A normative model of macro political risk assessment", *Multinational Business Review*, 6(2), 10–19.
- Bratton, M. and Van de Walle, N., 1997, *Democratic Experiments in Africa. Regime Transitions in Comparative Perspective.* Cambridge: Cambridge University Press.
- Cheloukhine, S, Haberfeld, M. R., 2011, Russian Organized Corruption Networks and their International Trajectories. New York: Springer.
- Deuber, G., 2014, "Russlands regionale Wohlstandsdisparitäten", Russlandsanalysen, 275, 9-16, retrieved from http://www.laender-analysen.de/russland/archiv.php (Mai 9, 2015).
- Erdmann, G. and Engel, U., 2007, "Neopatrimonialism Revisited: Critical Review and Elaboration of an Elusive Concept", *Journal of Commonwealth and Comparative Studies*, 45 (1), 95-119.

- Hanson, P., 2011, Networks, Cronies and Business Plans: Business-State Relations in Russia, in V. Kononenko and A. Moshes (eds), *Russia as a Network State. What Works in Russia When State Institutions Do Not?*, Palgrave Macmillan: Basingstoke, 113-138.
- Helmke G. and Levitsky S., 2004, "Informal Institutions and Comparative Politics: A Research Agenda", *Perspectives on Politics*, 2 (4), 725-740.
- Jensen, N. M., 2003, "Democratic Governance and Multinational Corporations: Political Regimes and Inflows of Foreign Direct Investment", *International Organization* 57(3): 587-616.
- Ledeneva, A., 2013, Can Russia Modernize? Sistema, Power Networks and Informal Governance, Cambridge University Press.
- Ledeneva, A., 2013a, "Russian's Practical Norms and Informal Governance: The Origin of Endemic Corruption", *Social Research*, 80(4), 135-1162.
- Ledeneva, A, Shekshnia, S., 2011, *Doing business in Russia: informal practices and anti-corruption strategies*, Institut français des relations internationales, Russie.Nei.Visions, n° 58.
- Liuhto, K., 2010, "Political risk for foreign firms in Russia", Romanian Journal of Economic Forecasting, 3, 141-157.
- Meyer K. E. end Peng M. W., 2016, "Theoretical foundations of emerging economy business research", *Journal of International Business Studies*, 47(1), 3-22.
- Mayring P., 2015, *Qualitative Inhaltsanalyse. Grundlage und Techniken*, Belz Verlage Weinheim und Basel.
- Nye J., 1967, "Corruption and Political Development: A Cost-Benefit Analysis," American Political Science Review, Vol. 61, No. 2, pp. 417-427 (June, 1967).
- Robinson, N., 2011, "Russian Patrimonial Capitalism and the International Financial Crisis", *Journal of Communist Studies and Transition Politics*, 27, 3-4.
- Robock, S. H., 1971, "Political risk: identification and assessment", *Columbia Journal of World Business*, 6, No. 4, 6-20.
- Rozov, N., 2013, "The Strategy of the new principal", Russian Politics and Law, 51(4), 85-101.
- Simon, J. D., 1982, "Political risk assessment: past trends and future prospects", Columbia Journal of World Business, 17(3), 62–70.
- Simon, J. D., 1984, "A theoretical perspective on political risk", *Journal of International Business Studies*, 15(3), 123–143.
- Torre, J., and Neckar, D. H., 1988, "Forecasting political risks for international operations, International Journal of Forecasting, Vol. 4, No: 2, 221-241.

# Interviews

Interview 1 (25.03.2014) with an executive of a large-sized engineering company (3 years of working record in Russia), established in 1904 in Russia

Interview 2 (25.03.2014) with an executive of a small-sized company from the transportation sector, established in 2005 in Russia

Interview 3 (25.03.2014) with an executive of medium-sized company supplier of pharmaceutical raw materials, established in 1992 in Russia

Interview 4 (26.03.2014) with a senior researcher, Economic University

Interview 5 (26.03.2014) with an executive of large-sized company for beverage products (4 years of working record in Russia), established in 2007 in Russia

Interview 6 (27.03.2014) with an executive of a large-sized winter sport supplier (5 years of working record in Russia), established in 2009 in Russia

Interview 7 (27.03.2014) with an executive a large-sized technology supplier in automotive industry (7 years of working record in Russia), established in 2007 in Russia

Interview 8 (28.03.2014) with a senior researcher, Economic University

Interview 9 (28.03.2014) with a senior researcher, local Research Institute

Interview 10 (31.03.2014) with an executive of a large-sized company for steel products (1 year of working record in Russia), established in 1982 in Russia

Interview 11 (01.04.2016) with an investigative journalist

Interview 12 (01.04.2014) with an executive of a large-sized enterprise supplier of refractory products (4 years of working record in Russia), established in 2006 in Russia

Interview 13 (01.04.2016) with an executive of a small-sized winter sports supplier (26 years of working record in Russia), established in 2012 in Russia

Interview 14 (02.04.2016) with an executive of a large-sized technology supplier for agriculture (2 years of working record in Russia), established in 2002 in Russia

Interview 15 (02.04.2014) with an executive of a medium-sized consulting company (11 years of working record in Russia), established in 2003 in Russia

Interview 16 (02.04.2014) with an executive of a company for office equipment (25 years of working record in Russia), established in 1991 in Russia

Interview 17 (03.04.2014) with a senior researcher, Economic University

Interview 18 (03.04.2014) with an executive of large-sized technology supplier for automotive industry (2 years of working record in Russia), established in 2009 in Russia

Interview 19 (10.04.2014) with an executive of medium-sized a company suppling technology to the biotechnological and pharmaceutical industry (3 years of working record in Russia), established in 1993 in Russia

Interview 20 (22.04.2014) with an executive of a large-sized company supplier for the cement and minerals industries (4 years of working record in Russia), established in 2001 in Russia

Interview 21 (14.05.2014) with the head of an economic association (6 years of working record in Russia)

Interview 22 (03.07.2014) with an executive of a large-sized construction company supplier of steel (7 years of working record in Russia), established in 1993 in Russia

Interview 23 (24.09.2014) with an representative of a large-sized construction company (2 years of working record in Russia), established in 1991 in Russia


# THE DYNAMICS OF CHINA'S EXPORT GROWTH: AN INTERTEMPORAL ANALYSIS<sup>1</sup>

# FRANCIS TUAN<sup>a</sup>, AGAPI SOMWARU<sup>b\*</sup>, SUN LING WANG<sup>c</sup>, EFTHIMIA TSAKIRIDOU<sup>d</sup>

\*Renmin University of China, Beijing, China and East Asia Economic Consulting, USA
 \*Economic Consulting, USA and East Asia Economic Consulting, USA
 \*Economic Research Service, USDA, USA
 \*Aristotle University of Thessaloniki, Greece

#### Abstract

China experienced dramatic growth throughout its entire economy during the 1978 to 2008 period. This significant economic expansion may be attributed to China's open policy after its 1978 overall economic reform. Our inter-temporal analysis of China's export growth between 1978 and 2008, based on the statistical stochastic decomposition approach, indicates that, in the first ten years after China's trade openness, most of its export growth occurred in the extensive margin of trade, but later most of China's export growth appeared in existing varieties or the intensive margin of trade. We find that the distribution of the extensive margin is more dispersed than of the intensive margin, as we use the country-product approach. After applying formal tests, the results show that the intensive margin plays a significant role in the growth of China's exports.

**JEL Classification:** F10, F14, F43, C5, C6, O40 **Keywords:** China, trade, intertemporal analysis, intensive margin, extensive margin.

\**Corresponding Author:* Agapi Somwaru, 607 Hawkesbury Lane, Silver Spring, Maryland 20904, USA. e-mail: agapi.somwaru@gmail.com

<sup>1.</sup> The views expressed herein are those of the authors and not necessarily those of the U.S. Department of Agriculture or the Economic Research Service.

### Introduction

China has now emerged as one of the world's largest trading nations for both agricultural and non-agricultural commodities. Moreover, China's trade is unique in many respects. The country is noted for its outstanding activities in the "processing trade" sector that involves importing inputs, which are then assembled into final products in China, and then re-exported (Naughton 1997). These activities play an important role in China's changing trade composition and patterns. China is one of the world's largest importers of raw materials and intermediate inputs. The country's exceptional processing trade ability impacts its trading partners, whether exporting or importing countries (Naughton 2004). China has been perceived as a competing threat by other labour-abundant developing countries that trade with developed countries. In recent years, however, the country's direct trade with developing countries, such as India, and countries in Southeast Asia and Africa has intensified dramatically (Batra 2007; Somwaru *et al.* 2007). In just 5 years, China's exports to India increased from \$2 billion to over \$50.49 billion in 2011 (WDI, 2013).

China competes world-wide not only on the basis of lower labour costs and abundance of natural resources, but also in terms of location that takes advantage of China's proximity to capital-rich East Asian economies and rapidly growing, developing markets in the Asia-Pacific region. In theory, geographic proximity often explains why neighbouring countries trade disproportionately, as Krugman (1991) suggests, in that neighbourhood trade is so strong as to create natural trading blocs. However, such explanations provide little insight into the roots of the rapid growth of China's exports.

Much of the recent theory assumes that developing countries imitate the production of goods invented in developed countries, ceteris paribus (Grossman and Helpman 1989; Hausmanne et al. 2005; Rodrik 2006). However, the speed at which countries can transform their productive structure and upgrade their exports depends on the basis of the knowledge of products and manufacturing technologies that has been acquired. China has entered a growth phase in its industrial development for producing and exporting a variety of goods. Many papers highlight a strong positive correlation between the number of export varieties a country produces and its living standard. For example, Hummels and Klenow (2005) find that larger and richer countries export more varieties of new goods or what they call the extensive margin of trade. Schott (2008) and Rodrik (2006) argue that China's exports are in high-quality sectors, which is similar to what happens in highly-developed countries. Amiti and Freund (2010), however, find that, despite the dramatic shift in China's exports to the United States over the 1992-2005 period, China's manufacturing exports remained unchanged after accounting for the processing trade. They find that China's export growth trade occurs in existing varieties or what they call the intensive margin of trade. Amiti and Freund find that the intensive margin of trade plays an important

role in China's exports. Our intertemporal analysis of China's export growth for the period 1978 to 2008 indicates that, in the first ten years after China's trade openness, most of its export growth occurred in the extensive margin of trade. Later, most of China's export growth has been in existing varieties (intensive margin trade). Our test results confirm that the intensive margin is a crucial factor in the growth of China's exports for the period 1978 to 2008.

The rest of the paper is organized as follows. Section 2 presents a snap-shot of China's economy and trade from an aggregate perspective. Section 3 discusses the methodology used to develop export growth and to decompose export flow patterns. Section 4 then follows and lays out all data employed in this study. Section 5 presents the empirical analysis, starting with the dynamics of export product differentiation, and continuing with the results of decomposing China's exports. We use bootstrapping to obtain the distributional stochastic characteristics of the trade margins and their variability over the study period. Section 6 empirically captures China's export growth and presents formal tests to identity the intertemporal relationship between China's export growth and extensive/intensive margin of trade following the country's significant economic expansion after its 1978 overall economic reforms. Finally, Section 7 concludes the study.

# 2. Structural shifts in China's economy and trade - an overview from an aggregate perspective

China has achieved the most persistent economic growth among developing countries. The annual growth rate in the country's real gross domestic product (GDP) averaged about 10 percent per annum from 1978 to 2011 (WDI, 2013). The country's GDP *per capita* or its purchasing power parity adjusted in constant 2005 U.S. dollars was \$523.95 in 1980 and \$7,417.89 in 2011. The importance of the agricultural sector's value added output, with respect to the entire economy, continued to decrease over the past decades, while the importance of the service sector increased and the manufacturing sector's shares remained largely the same (WDI, 2013). The agricultural share of the GDP decreased from 28.2% in 1978 to 10.0% in 2011, while the share of the service sector increased from 23.9% in 1978 to 43.4% in 2011 (WDI, 2013).

As expected, the share of China's workforce employed in agriculture decreased, while employment share in services increased over the period examined. Many economists maintain the hypothesis that a fundamental feature of growth and development is the decline in the proportion of the workforce employed in agriculture. China has been experiencing a similar trend since the 1980s.

In terms of the broader economy, China's agricultural exports have been playing a smaller role in the country's total exports. Since the middle 1980s, the share of agriculture in total merchandise exports gradually declined, from 5.71% in 1984 to 0.54% in 2011 (WDI, 2013). Manufacturing exports continued to grow and accounted

for 47.65% of total merchandise exports in 1984, a share that rose to an astonishing 93.30% in 2011. Exports of total merchandise to high income countries reached 88.25% in 1991 and then declined to 72.20% in 2011 (WDI, 2013). These high income countries include the United States, Canada, 18 European Member States (Austria, Belgium, Denmark, France, Greece, Iceland, Ireland, Italy, Luxembourg, the Netherlands, Norway, Portugal, Sweden, Switzerland, the United Kingdom, and Germany), Japan, Australia, and New Zealand. During the same period exports to developing countries increased from 7.55% to 20.02%. Among developing economies, the East Asia and Pacific regions accounted for most of China's merchandise exports, followed by Latin American and Caribbean countries. Manufacturing accounted for the largest and fastest growing share of total imports, while agriculture's share declined to about 4.65% in 2011 (WDI, 2013). From 1984 to 2011, imports of ores and metals along with fuel import shares into China dramatically increased by 10 and 16 percentage points, respectively (WDI, 2013).

## 3. Methodology for decomposing export growth

An examination of the time path of trade flows for any economy swiftly reveals that trade of both exports and imports tend to fluctuate along a long-term growth path. For this reason, to obtain the long-term growth of China's exports and to identify deviations from the growth path, we use the technique known as the Hodrick-Prescott (HP) (1997) method or Trend and Cycle Decomposition (TCD). We applied the HP/TCD method and derived the growth rates of China's exports (Diao *et al.* 2001; Somwaru *et al.* 2007).

The HP method, developed by Hodrick and Prescott (1997), removes a smooth trend,  $g_t$  component, from observed given data y, by solving the following expression:

$$\min_{(g_t)_{t=-1}^T} \left\{ \sum_{t=1}^T (y_t - g_t)^2 + \lambda \sum_{t=1}^T [(g_t - g_{t-1}) - (g_{t-1} - g_{t-2})]^2 \right\}$$
(1)

The residual  $c_t = y_t - g_t$  (the deviation from the trend) is commonly referred to as the business cycle component and it is the deviation from  $g_t$ , for t=1,..., T. The measure of the smoothness of the  $\{g_t\}$  path is the sum of the squares of its second difference. The notion is that, over a long time period, the cycles,  $c_t$ , where  $c_t = y_t - g_t$ , average near zero.

The  $\lambda$  parameter is a positive number that penalizes variability in the growth component of the series. The larger the value of  $\lambda$ , the 'smoother' the underlying growth trend  $g_t$  is. For a sufficiently large  $\lambda$ , at the optimum, all  $g_{t+1} - g_t$  must be arbitrarily near some constant  $\beta_t$  and, thus, the  $g_t$  approaches  $g_0 + \beta_t$ . This implies that at the limit, as  $\lambda$  approaches infinity, the solution is the least squares fit of a linear time trend mode, and for  $\lambda = 0$ , smoothed data are exactly the same as sample data.

The selection of *the smoothing parameter*  $\lambda$  is based on a probability model. If the cyclical components and the second differences of the growth components are identically and independently distributed normal variables with mean zero and variances  $\sigma_1^2$  and  $\sigma_2^2$  (which they are not), then the conditional expectation of the  $g_t$ , given the observations, would be the solution of the above equation when  $\sqrt{\lambda} = \sigma_1 / \sigma_1$ .

Different values of  $\lambda$  provide different information, e.g., a large  $\lambda$  value<sup>2</sup> approximates the annual average rate of growth given by an ordinary least squares fit to the log of the data. In other words, the  $\lambda$  parameter should be adjusted according to the fourth power of a change in the frequency of observations (King and Rebelo 2000; Baxter and King 1999; and Diao *et al.* 2001). In this paper, for the purpose of our annual data series, we adopt the procedure for the smoothing parameter given in Ravn and Uhlig (2002).

There is a wealth of studies in the literature on the importance of export growth and intensive and extensive margins. For example, Evenett and Venables (2002), and Hummels and Klenow (2005) find that the extensive margin played a significant role for growth in exports. When examining cross-country differences, Hummels and Klenow, using 1995 data, find that the extensive margin accounts for 60 percent of the greater exports of larger economies. On the other hand, Felbermayr and Kohler (2006) find that the intensive margin was a more important factor for trade growth between 1970 and the mid-1990s. Helpman *et al.* (2008), Eaton *et al.* (2008), and Amiti and Freund (2010), among others, find that the intensive margin of trade is more important for export growth.

Evenett and Venables (2002) define the extensive/intensive margin at the *country-product* level, while Amiti and Freund (2010) at the *product* level and, in particular, concerning exporting industries. However, Helpman *et al.* (2008) and Felbermayr and Kohler (2006) define the extensive/intensive margin at the *country* level. Unlike other studies in this paper, we adopt a *country-product* approach definition of the extensive/intensive margin. In other words, we account for all products that China exports to each of its trade partners during the period under study. Moreover, we do not compare the export growth in an initial year with the export growth in some year ahead (the so-called comparative approach), but we apply a dynamic approach in which export growth is attained for each year of the period under study (Besedes and Prusa 2011).

Below we provide a quick overview of the method adopted for obtaining the margins of trade. We employ the method originated by Feenstra (1994) and adopted by Amiti and Freund (2010) to distinguish between the intensive and extensive margins of China's exports. Unlike Amiti and Freund (2010), we define the extensive and intensive margin at the country-product level and not just at the product level. We employ an intertemporal method to decompose export growth flows into the intensive and extensive margins of trade for China for each product and country. The original

idea of Feenstra's work is to include new product varieties into an index. Denoting I as the set of varieties available in both periods,  $I \subseteq (I_t \cap I_{t-1})$ , the net variety growth index is defined as the fraction of expenditure in period t-I on the goods  $i \subset I$  relative to the entire set  $i \subset I_{t-1}$  as a ratio of the fraction of expenditure in period t on the goods  $i \in I$  relative to the entire set  $i \in I_t$ , minus one. Let  $V_{it}$  be the value of trade at time t in product  $i (V_{it} = p_t q_i)$ , then the Feenstra index of *net variety growth* (Amiti and Freund 2010) is defined as follows:

$$\frac{\sum_{i \in I} V_{t-1i} / \sum_{i \in I_{t-1}} V_{t-1i}}{\sum_{i \in I} V_{ti} / \sum_{i \in It} V_{ti}} - 1$$
(2)

Feenstra's (1994) seminal work on measuring export prices incorporating new goods leads to a natural index of variety growth (equation 2) that has been widely used in relevant literature. The index will be equal to zero if there is no growth in varieties relative to the base period and positive if the number of varieties has grown. If export growth classifications are split (or reclassified) then new classifications are merged and the index will tend to overstate the extensive margin.

Our analysis focuses on whether the growth in China's product exports to its partners falls into existing or new varieties. Using this decomposition concept, the value of trade,  $V_{ii}$ , for the *i*<sup>th</sup> product at time *t*, can be decomposed into the value of existing varieties  $V_{ii} \cdot D^e$  disappearing varieties  $V_{ii} \cdot D^d$  and new varieties,  $V_{ii} \cdot D^n$  where  $D^e$ ,  $D^d$ , and  $D^n$  are dummy variables indicating whether the product exists in both period *t* and 0, only in period 0, or only in period *t*, respectively. Thus,  $D^e = 1$  indicates an existing variety,  $D^d = 1$  a disappearing variety, and  $D^n = 1$  a new variety. The following decomposition equation is then used to identify the presence of new or existing varieties:

$$\frac{\sum_{i=1}^{I} V_{it} - \sum_{i=1}^{I} V_{it-1}}{\sum_{i=1}^{I} V_{it-1}} = \frac{\left(\sum_{i=1}^{I} V_{it} D^{e} - \sum_{i=1}^{I} V_{it-1} D^{e}\right) - \sum_{i=1}^{I} V_{it-1} D^{d} + \sum_{i=1}^{I} V_{it} D^{n}}{\sum_{i=1}^{I} V_{it-1}}$$
(3)

In the equation above, total growth in trade relative to the base period is decomposed into three parts: (i) the growth in products that were exported in both periods, the *intensive margin*; (ii) the reduction in export growth due to products no longer exported, disappearing goods; and (iii) the increase in export growth due to the export of new products. The share of export growth due to the *extensive margin* is defined as the new-goods share less the disappearing-goods. In other words, equation (3) separates export growth in trade into growth in existing varieties or the intensive margin, disappearing goods and growth in new varieties or the extensive margin. Note that, by construction, the intensive and extensive margins of exports sum up to one. There is a direct relationship between the Feenstra index of net variety growth and the decomposition index in equation (3). Feenstra's net variety growth index combines new exports and disappearing exports into one. In this respect, the net index is more robust than the decomposition index because the decomposition index attributes reclassified varieties to the extensive margin.

We apply an intertemporal approach as we develop a series of measures of extensive, intensive and export growth measures, as follows:

$$Int_{i} = \frac{\left(\sum_{i=1}^{l} V_{ii} D^{e} - \sum_{i=1}^{l} V_{ii-1} D^{e}\right)}{\sum_{i=1}^{l} V_{ii-1}} , for t = 1979-78, 1980-79, 1981-80, ..., 2008-07 (4)$$

$$Ext_{t} = \frac{\sum_{i=1}^{l} V_{it} D^{n}}{\sum_{i=1}^{l} V_{it-1}} , for t = 1979-78, 1980-79, 1981-80, \dots, 2008-07$$
(5)

$$Net_{t} = \frac{\sum_{i \in I} V_{t-1i} / \sum_{i \in I_{t-1}} V_{t-1i}}{\sum_{i \in I} V_{ti} / \sum_{i \in I_{t}} V_{ti}} - 1, for t = 1979-78, 1980-79, 1981-80, ..., 2008-07 (6)$$

Where, *Int* denotes intensive margin, *Ext* denotes extensive margin and *Net* denotes the Feenstra index for the  $i^{\dagger}$  product for each one of China's country partners. The share of export growth attributed to each margin is calculated using equation 3. Note that variety growth is 1/Feenstra index. This measure has the nice feature that if classifications are simply split, and their share of total trade remains unchanged, the index remains unchanged. However, if classifications tend to be split into their share of total trade changes, then the index is more likely to overstate the extensive margin.

Unlike other studies, we consider China's exports to all its trade partners. Thus, our analysis of trade growth is based on *country-product* relationships. We account for all exporting products that had never previously been sold abroad, as well as for all products already exported to a new destination country, as they can both change China's extensive margin.

Besedes and Prusa (2011) argue that, while a firm may have a clear idea of its home market conditions, it may not know the level of demand abroad or have all information about ongoing costs associated with exporting. Their empirical approach is motivated by an extension to the Melitz (2003) model to account for information uncertainties associated with foreign markets. In our paper we employ *bootstrapping* to quantify uncertainties associated with market risks. The bootstrap is a computing intensive statistical resampling technique. Its advantage is that it is less restricted by parametric assumptions than more traditional approaches to market uncertainties. We employ the "bootstrap" technique (Efron and Tibshirani 1994; Varian 1996) for assessing the variability associated with market risks and provide confidence intervals for the exporting margin of trade and export growth.

44 F. TUAN, A. SOMWARU, et al., South-Eastern Europe Journal of Economics 1 (2016) 37-57

Let 
$$\theta = \left[\frac{\sum_{i=1}^{l} V_{ii} D^{e} - \sum_{i=1}^{l} V_{il-1} D^{e}}{\sum_{i=1}^{l} V_{il-1}}\right]$$
 represent the intensive margin. Drawing 1,000 boot-

strap samplings of existing varieties, then we can obtain the *expected value* denoted as  $E(\hat{\theta})$  of the estimator ( $\hat{\theta}$ ) of the intensive margin along with the variance, denoted as  $var(\hat{\theta}) = E[(\hat{\theta} - E(\hat{\theta}))]$ . Similarly, we draw an additional three 1,000-bootstrap samplings for the extensive, net trade margin, and the export growth of all China's merchandise to obtain estimates of their mean, variability, and confidence intervals.

### 4. Data

China's trade data used in this study are at the 4-digit Standardized International Trade Classification (SITC) level, Revision 4 (United Nations 2006 and UNCTAD/WTO 2010). The data source is the Commodity Trade Statistics Database (UN-COMTRADE) maintained by the Statistic Division of the United Nations (UN) (United Nations 2010). China started to share public trade statistics with international organisations in 1984. However, the validity of some trade flows reported remains questionable (Gehlhar 1996). Thus, we draw upon data reported by China's *trade partners* and compiled by the UN, starting from 1978 (Gehlhar 1996). The SITC 4-digit data of China's bilateral trade flows (United Nations 2006) for each year of the period under study were processed using the Statistical Analysis System (SAS). We adopted the same approach for processing bilateral data as in Gehlhar (Gehlhar 1996 and https://www.gtap.agecon.purdue.edu/databases/trade\_data.asp, 2012; Somwaru *et al.* 2008).

In each one of the study years we processed the bilateral trade for each country/ partner of China (United Nations 2010, Country Classification). By maintaining the country-product association, we were able to obtain the extensive and intensive margins at the country level for each product. It should be acknowledged that defining the extensive margin at the product level might artificially impose an upper bound on the extensive margin, as there might be no additions on the extensive margin. For this reason, we defined the margins at the country-product level. This allows the extensive margin to have the largest possible role.

Moreover, we developed aggregate country groups and these were used to develop growth margins to understand the impact of country aggregation. Such groups include: the EU-25, the High Income country group, the Developing country group, the Oceania country group, the High Income South East Asia, the South and South East Asia, the Central and South America, the High Income Oil Exporting group, the Africa country group, and the Transitional Economies country group (see Appendix, Table 1 for country list in country groups; the Africa country group includes all China's African trade partner countries). Some country groups, such as the African country group or the Transitional Economies, are subsets of larger groups.

A primary drawback encountered when using the Harmonized System (HS) trade flow classification is that there have been major reclassifications of trade data in 1996 and 2002 at the HS 6-digit level; thus, a product might be classified as a new variety simply because there has been a new product code or because previous codes were split (United Nations 2010, Economic and Social Classifications, and UNCTAD/WTO 2010). Amiti and Freund (2008, 2010), using China's exports to the world from 1997 to 2005 in HS 8-digit categories, show that the extensive margin accounted for only 26 percent of the total export growth. They find that reclassifications push the extensive margin up. They state that existing product codes are not likely to be a random sample since entirely new products will, by definition, require a new code; therefore, this can be treated as a lower bound of the extensive margin.

Besedes and Prusa (2011) use the 10-digit data of the Harmonized System (HS) to verify whether results of measuring the margins of trade are specific to the 1972–1988 period or extend to the 1989–2001 period, as well. They conclude that when measuring growth in the extensive margin, it is more insightful to consider changes over a *longer-run horizon*, since the value of exports in new product codes is generally small when they are first introduced. Furthermore, Besedes and Prusa (2011), using 4-digit *long horizon* trade data (SITC), argue that their choice arises from concerns about quality and inconsistency due to reclassification of product codes in consequent years. Following Besedes and Prusa (2011), this study uses relationships defined at the 4-digit SITC level.

Our extended data work is aimed at better understanding changes and trends in China's export growth over time across products and countries. For this reason, we draw data from all China's bilateral trade partners/countries in the datasets. Our analysis focuses on the growth rates of exports, including both agricultural and nonagricultural products, over the study period. For each year of the study we account for all *cross sectional* data of exports from China to its trade partners.

## 5. Empirical analysis

## 5.1 The dynamics of export product differentiation

Our technique for analysing China's export growth pattern is based on the Trend and Cycles Decomposition (TCD) approach. This approach, by capturing the dynamics of growth in trade and trade policy regimes, allows us to obtain factually based evidence of China's annual trade growth. We find that one of the distinguishing features of China's trade growth is the persistent diversification of partners over time. Table 1 captures China's export growth pattern using trade data at 4-digit SITC (Somwaru *et al.* 2007).

To capture the dynamic features of China's exports, we use its annualized growth rates. The resulting series of growth rates indicate relatively large annual variability due to a range of reasons. Many of the causes for these fluctuations in year to year data are not essential for capturing a "true" trajectory in China's trade growth. Instead, these deviations tend to obscure the underlying longer-term trend in export

#### 46 F. TUAN, A. SOMWARU, et al., South-Eastern Europe Journal of Economics 1 (2016) 37-57

growth rates. The longer-term trends in China's export growth would better reveal China's prevailing export patterns. Thus, by employing the TCD methodology, we were able to remove or 'filter' these fluctuations from the primary data (Diao *et al.* 2001, Somwaru *et al.* 2007).

Item	1980s	1990s	2000s
Total imports	17.36	13.39	22.40
Total exports	16.67	16.75	20.25
Imports from Developed countries	14.57	13.56	17.39
Exports to Developed countries	14.30	19.73	21.87
Imports from Developing countries	20.27	13.31	25.96
Exports to Developing countries	18.05	15.03	18.97
Imports from India	11.34	33.61	38.12
Exports to India	47.23	23.88	36.96
Imports from South America	7.86	13.44	33.42
Exports to South America	7.87	28.16	29.21
Imports from Southeast Asia	5.76	24.42	28.72
Exports to Southeast Asia	14.31	18.94	23.06
Imports from Africa	3.03	25.67	37.68
Exports to A frica	0.94	20.07	26.45

 Table 1. China's estimates of total merchandise trade annual growth\* by trading partners

\* *Note:* Growth rates estimated using Trend and Cycle Decomposition method annualized by time period.

Source: Authors' calculations.

In the 1980s, China's average annual trade growth rates with the developing country group outpaced the average annual trade growth rates with the developed country group. It was in the 2000s that China's average annual growth import rates from the developing country group were almost the same as its average annual growth rates of exports to developed countries. The ongoing global relocation of labour-intensive manufacturing has accommodated China's export growth, while the transition to a more market-based economy has helped diversify China' product mix. This, however, did not necessarily mean that China's trade growth has limited developing country export opportunities from direct competition in the case of similar products. In fact, the growth of China's imports from developing countries generally out-

paced import growth from developed countries (Table 1). In the most recent decade, China's import growth has been greater than its export growth, an acceleration of China's trade with developing countries that has often been overlooked.

To clearly capture the dynamic features of China's trade and competitiveness, we calculated the annual growth rate of China's detailed trade. Unlike other studies that concentrated on China's trade with the United States (Hammer 2006, Naughton 2004), we include all of China's major high-income trade partners (see Appendix Table 1 for the list of all countries and country groups). While we process China's bilateral trade with all of its trade partners, we only report the growth rates with selective countries/country groups due to space limitations.

Although the growth rates of total exports with India were unstable in the early 1980s, during the last 15 years, China's trade growth with India has steadily increased (Table 1). This growth coincides with India's liberalising industrial sectors and India's higher sustained growth rates in income and consumer purchasing power.

Numerous studies have analysed the importance of trade in intermediate goods and the influence of geographic proximity on production for countries sharing borders (Naughton 1997, Gupta 1997, Ng and Yeats 1999). China's trade performance indicates that geographic proximity is a factor enhancing the value-added processing chain observed in the country (Figure 1). In the 1980s trade with the economies of the Arab world and Middle East and North Africa countries dominated China's trade flows. In the 1990s and 2000s trade with neighbouring East Asia and Pacific countries dominated China's trade (Figure 1). China's rise in international processing activities reflects the strategies of Asian firms to relocate their industries to China so as to take advantage of China's comparative advantage in production processing that arises primarily from competitively lower labour costs. China's trade policy has favoured assembly and processing operations through tariff exemption on intermediate goods, and set off the expansion of China's trade in intermediate goods in foreign-invested enterprises (FIEs) and economic and technological development zones (ETDZs) (Berger and Martin 2013, Tan and Khor 2006; Somwaru et al. 2007). These selective trade policies have accelerated China's international processing activities and became the engine for rapid diversification of its manufacturing exports, well beyond geographic proximity regions (Berger and Martin 2013). Indeed, the most noticeable annual growth of China's total merchandise exports and imports are with African countries in the 1990s and 2000s (Table 1 and Figure 1). Intermediate products, while amounting to almost two-thirds of China's total imports, display China's comparative advantage in production "by stage." These findings tend to weaken the Krugman-Bhagwati (Krugman 1991, Bhagwati 1992) debate on whether neighbourhood determines the direction of trade or geographic proximity is more irrelevant, a fact also supported by the annual growth rates of trade between China and its neighbouring South and Southeast Asian countries (Table 1).



Figure 1. China's total merchandise export shares by country group, 1978-2011

Source: Authors' calculations.

#### 5.2 Intensive versus Extensive Margin of Trade

By applying equations (4), (5) and (6), we obtain the intensive and extensive margins of trade of China's total merchandise along with net variety growth for each year of the period under study. The decomposing measures confirm that China's exports grow primarily along the intensive margin trade. Based on China's export data from 1978 to 2008, we find that only 32.1 percent in the period from 1978 to 1989 are existing product varieties (Table 2) while in the 2000 to 2008 period, exports in existing product varieties account for 87.3 percent (Table 2). For the entire period (1978-2008), China's exports in the intensive margin trade account for 79.8 percent of total exports and present the largest variability (measured by the standard deviation). The 90-percent confidence interval of the mean measured by the lower (66.5) and upper bounds (99.3) indicates the probability that the confidence interval contains the true population mean at 90 percent confidence level (Table 2).

Results show that there was a significant reorientation of China's export growth during the period under study. In this sense, in the first ten years after China's trade openness, most of its export growth occurred in the extensive margin, at 67.9 percent

versus 32.1 percent for the intensive margin. After 1989, the pattern switches with the intensive margin accounting for 74.5 percent of export growth, and the extensive margin for 25.5 percent. Between 1978 and 1989 China started exporting many products. The growth in the number of products peaks during that period. Afterwards, much of the export growth occurred in the intensive margin. In sum, during the period under study we observed reallocation/reorientation of export growth.

				Confidence	ntorral of the mean *	
All type of m	echandise		- Standard	Lower Upper		
Partner-Wo	orld	Mean	deviation	bound	bound	
	110	%	de vintion	bound	ootand	
Intensiva						
mensive	1978-1989	32.14	2.62	28.91	36.21	
	1990-1999	74.51	4.34	70.89	82.57	
	2000-2008	87.29	8.13	76.18	97.18	
	1978-2008	79.83	12.32	66.46	99.03	
Extensive						
	1978-1989	67.86	9.75	55.54	78.91	
	1990-1999	25.49	5.36	19.65	32.57	
	2000-2008	12.71	8.35	4.18	28.18	
	1978-2008	20.17	11.01	8.03	31.97	
Net variety growth						
(Feenstra index)	1978-1989	17.67	7.56	6.21	28.91	
	1990-1999	14.25	4.43	8.57	20.69	
	2000-2008	25.74	7.56	14.18	36.18	
	1978-2008	22.06	12.25	8.03	41.97	
Total export growth						
	1978-1989	224.01	32.01	174.41	259.54	
	1990-1999	320.01	45.58	262.69	387.65	
	2000-2008	337.45	32.56	296.18	384.18	
	1978-2008	8230.47	315.96	7735.97	8631.46	

Table 2.	Varietv	growth in	China's exports.	1978-2008
THULL TO THE	,		china b chiporto,	1770 2000

\* 90% Confidence level.

Source: Authors' calculations.

Besedes and Prusa (2011), argue that they use a 4-digit level trade dataset (SITC) "due to concerns about quality and consistency of more disaggregated data as well as for earlier years' data". Our analysis, using 4-digit level data, is in agreement with Amiti and Freud (2010) in that export growth to the U.S. from China took place along its intensive margin from 1997 onwards. They demonstrate this by using a Törnqvist chain-weighted price index to measure the term-of-trade effect, namely, that this growth in the intensive margin supports the traditional theory of welfare gains for importing countries through lower import prices. In this sense, as China increases

its supply of existing varieties in world markets, this is likely to exert downward pressure on world prices of these goods.

The export growth from 1978 to 2008 for the extensive margin amounts to 20.17 percent (mean), with variability being almost the same as that of the intensive margin (11.01, see Table 2). The 90-percent confidence interval of the mean measured by the lower (8.03) and upper bounds (31.97) indicates the probability that the confidence interval contains the true population mean.

It should be noted that by employing the bootstrapping technique we are able to assess the *variability*, and derive estimates of the intensive and extensive margins, the export growth of China's total merchandise statistics, such as mean, standard deviation, and confidence intervals. Otherwise, the intensive and extensive margins derived and the intertemporal export growth estimates would completely lack statistical properties or measures of uncertainty. Since the bootstrapping procedure is distribution-independent, it provides an indirect method of assessing the stochastic properties underlying the intensive and extensive trade margins and export growth. Furthermore, the 90-percent confidence interval of the mean measured by the lower and upper bounds indicates the probability that the confidence interval contains the true population mean.

The variability (measured by the standard deviation) of the extensive margin for the entire study period is almost the same as that of the intensive margin, while the mean and median of extensive and intensive margin distributions are different. This implies that their distributions are not limited by the number of product codes (4-digit) or by how the data of each distribution are spread apart. It should be noted that the distribution of the extensive margin is more dispersed than the intensive margin in our analysis because we use the country-product approach.

## 6. Export-variety growth

In the late 1970s and 1980s, China's export growth surged from its base that largely consisted of consumer goods. Thereafter, during the 1990s, China's export growth of capital goods took the lead. This indicates that China's trade growth was broad-based and not fuelled by a few products (see Somwaru *et al.* 2007). Figure 2, depicting the export shares of selected commodities, clearly show that China has entered a growth phase in its industrial development for producing broad-based exporting goods. China's exports of apparel, textiles, and footwear have heavily shifted towards chain and parts thereof of iron or steel.

China's adoption of open foreign direct investment (FDI) strategies might have attracted multinational enterprises, which are more likely to have expanded their trade flows to an extensive margin growth pattern through growth in the number of trade varieties in the late 1970s. China's spur of export growth in more recent years, however, lies in value added goods along the intensive margin (Amiti and Freund 2010). China's rise in international processing activities reflects the strategies of Asian firms for relocating their industries to China so as to exploit China's comparative advantage. Moreover, China's trade policy has favoured assembly and processing operations through tariff exemptions (Somwaru *et al.* 2007). This selective trade policy might have accelerated China's international processing activities with China's high export growth on existing products. China's trade growth with developing countries has taken the lead as the country gains in intensive margin goods of trade. The increased supply of existing varieties of China's exports in recent years is the engine for its trade with emerging and developing countries.

Following Kreuger (1999), we perform formal statistical tests in an attempt to gain insight into the correlation of export growth and the intensive and extensive margins of trade in an *ex-post* mode. The advantage of nonparametric methods over econometric methods is that nonparametric methods do not require any specification of a functional form. In addition, nonparametric methods account for more complicated non-linear relationships between different policies and export growth. Moreover, nonparametric methods accommodate non-linear relationships between different policies and export growth. Moreover, nonparametric tests - the Kruskal-Wallis and the Van der Waerden Scores (Normal) test - are performed to investigate the effects of China's policies. The Kruskal-Wallis test statistic is given by:



Figure 2. China's export shares of selected merchandise, 1978-2008

Source: Authors' calculations.

F. TUAN, A. SOMWARU, et al., South-Eastern Europe Journal of Economics 1 (2016) 37-57

$$H = \frac{12}{N(N+1)} \sum_{i=1}^{k} \frac{T_i^2}{n_i} - 3(N+1)$$
(7)

Where, N is the sample size,  $T_i$  is the sum of ranks for the  $i^{th}$  group, and  $n_i$  is the number of observations in the  $i^{th}$  group. Test statistic H approximately follows a chisquared distribution with k-l degrees of freedom, where k is the number of groups or populations. The Van der Waerden (Normal) Scores are the quantiles of a standard normal distribution and are computed as follows:

$$a(R_j) = \Phi^{-1}\left(\frac{R_j}{n+1}\right)$$
(8)

Where,  $\Phi$  is the cumulative distribution function of a standard normal distribution. Our aim is to test China's independence of export growth and the margin of trade. The Kruskal-Wallis statistic for the intensive margin and export growth is 4.67 and 2.33, respectively, for the extensive margin 8.00, while for the critical value 6.49. The Kruskal-Wallis  $X^2$  statistic rejects the null hypothesis at 0.05% significance level for the intensive margin and export growth, but fails to reject it for the extensive margin. This implies that the first moments of the distribution (mean and variance) of export growth and the intensive margin are the same, but the null hypothesis is rejected for the extensive margin (Table 3). In other words, the intensive margin plays a significant role for the growth of China's exports. The Van der Waerden  $X^2$  statistic also rejects the null at the 0.05% significance level for the intensive margin (-0.09) and exports (-0.79) but not for the extensive margin (6.88), since the value of the asymptotic statistic is 6.31 (Table 3).

In sum, China experienced dramatic growth throughout its entire economy during the 1978-2008 period. Export growth along the intensive trade margin seems to have benefited from this great economic expansion.

Item	Kruskal-Wallis statistic (Wilcoxon mean score)	Van der Waerden statistic (mean score)
Francisco	8.00	0.88
Intensive	4.67	-0.09
Exports (growth)	2.33	-0.79
Asymptotic statistics $(X^2)$	6.49	6.31
Critical value @ 0.05 significance	0.039	0.043

Table 3. Results of the paired - t test on export growth and trade margins

Source: Authors' calculations.

52

## 7. Conclusions

Despite extensive discussions about China's benefiting from embracing globalization, better understanding of the dynamics of China's trade patterns requires a comprehensive profile of China's trade growth using bilateral time-series data. Our methods allowed us to perform such an analysis and results indicate that China's rapid export growth is largely driven by expanding trade in existing goods, or the intensive margin of trade, especially in more recent years. We find that other developing countries are not only playing a complementary role in China's trade growth through trading with China, but have also enabled China's export growth to acquire a faster pace. China's trade growth patterns with major high income countries clearly indicate that the partners' adjacency or neighbourhood hypothesis alone is unlikely to explain the country's unprecedented export growth.

China's outstanding performance in export growth can be traced back to the late 1970s and early 1980s with changes in its policies and increasing involvement in the international segmentation of production processes through FIEs and ETDZs. China's great flexibility via FDI and 'joint ventures' spurred by accumulated assets might have provided the foundation for China to redeploy its capabilities from sector to sector and, consequently, to its export growth expansion. This study finds that there was significant reorientation of China's export growth during the 1978-2008 period. Our intertemporal analysis of China's export growth for the period 1978 to 2008, based on the statistical stochastic decomposition approach, indicates that in the first ten years after China's trade openness, most of its export growth occurred in the extensive margin of trade. However, after 1989, the pattern switched with the intensive margin accounting for 74.5 percent of the growth of exports, and the extensive margin for 25.5 percent. Export growth in the 1990s and 2000s is mainly accounted for by high export growth of existing products (the intensive margin) rather than by new varieties (the extensive margin). One caveat is that the extensive margin might be facing with an upper bound given the definite number of exporting 'codes.' In this study, we define each observation at the country-product level so that we have better estimates of the extensive and intensive margins. Additionally, this study can be treated as the starting point of further research into identifying China's trade growth and patterns.

#### References

- Amiti, M. and and C. Freund. 2008. "The Anatomy of China's Export Growth." Policy Research Working Paper 4628, The World Bank, Development Research Group, Trade Team, Washington, DC.
- Amiti, M. and and C. Freund. 2010. "An Anatomy of China's Export Growth." in Robert C. Feenstra and Shang-Jin Wei, (Eds.) China's Growing Role in World Trade, The University of Chicago Press.
- Batra, Amita. 2007. "Structure of Comparative Advantage of China and India: Global and Regional Dynamics." *China & World Economy*, Volume 15, Issue 6, November–December 2007, Pages: 69–86.

- Baxter, M. and Robert King. 1999. "Measuring Business Cycles: Approximate Band-Pass Filters for Economic Time Series," The Review of Economics and Statistics, 81:4 (1999), 573–593.
- Berger, B. and R.F. Martin. 2013. "The Chinese Export Boom: An Examination of the Detailed Trade Data." *China & World Economy*. Volume 21, Issue 1, January-February 2013, Pages: 64–90.
- Besedes, Tibor and Thomas J. Prusa. 2011. "The Role of Extensive and Intensive Margins and Export Growth." *Journal of Development Economics*, 96(2), 71-379.
- Bhagwati, J. 1992. "Regionalism vs. Multilateralism." The World Economy, 15, 5.
- Diao, X., T. Roe, and A. Somwaru. 2001. "What is the Cause of Growth in Regional Trade: Trade Liberalisation or RTA? the Case of Agriculture." *World Economy*, 24(1): 51-79.
- Eaton, J., M. Eslava, M. Kugler, and J. Tybout. 2008. "The Margins of Entry into Export Markets: Evidence from Colombia." in Elhanan Helpman, Dalia Marin, and Thiery Verdier, 15 eds., The Organization of Firms in a Global Economy, Cambridge, MA: Harvard University Press, 231-72[12].
- Efron, B., and R. Tibshirani (1994), R. J. An Introduction to the Bootstrap. Boca Raton, FL: CRC Press.
- Evenett, S. and A. J. Venables. 2002. "Export Growth in Developing Countries: Market Entry and Bilateral Trade Flows." University of Bern working paper, mimeo.
- Feenstra, R. 1994. "New Product Varieties and the Measurement of International Prices", *American Economic Review*, LXXXIV, 157-77.
- Felbermayr, G. and W. Kohler, 2006. "Exploring the Intensive and Extensive Margins of World Trade." *Review of World Economics*, 42(4): 642-674.
- Gehlhar, Mark, 1996. "Reconciling Bilateral Trade Data for Use in GTAP." *Global Trade Analysis Project (GTAP) Technical Paper No.10*, Purdue University

and https://www.gtap.agecon.purdue.edu/databases/trade\_data.asp, 2012

- Grossman, G. and E. Helpman, 1989. "Product development and international trade." The *Journal* of *Political Economy* 97(6): 1261-1283.
- Gupta, D.D. 1997. "China Engaged: Integration with the Global Economy." The World Bank Publication.
- Hammer, A. 2006. "The Dynamic Structure of U.S.-China Trade, 1995-2005." U.S. International Trade Commission, Working paper No 2006-07-A.
- Hausmann, R. J. Hwang and D. Rodrik. 2005. "What You Export Matters." NBER Working paper 11905. National Bureau of Economic Research, Cambridge, MA.
- Helpman, E., M. Melitz and Y. Rubinstein. 2008. "Estimating Trade Flows: Trading Partners and Trading Volumes." *Quarterly Journal of Economics*, 123(2).
- Hodrick, R.J., and E.C. Prescott. 1997. "Postwar U.S. Business Cycles: An Empirical Investigation," Journal of Money, Credit and Banking, 29, 1997. Hollander, M. and D. A. Wolfe. Nonparametric Statistical Methods. New York: John Wiley & Sons, pages 115–120.
- Hummels, D. and P. Klenow. 2005. "The Variety and Quality of a Nation's Exports." *American Economic Review*, 95, 704-723.
- King, R. and S. Rebelo. 2000. "Resuscitating Real Business Cycles," National Bureau of Economic Research (NBER), Working Paper 7534. http://www.nber.org/papers/w7534, 1050 Massachusetts Avenue, Cambridge, MA 02138.
- Krueger, A.O. 1999. "Trade Creation and Trade Diversion under NAFTA." NBER Working Paper 7429. pp 32. http://www.nber.org/papers/w7429
- Krugman, P. 1991. "Is Bilateral Bad?." in E. Helpman and A. Razin, (eds), *International Trade and Trade Policy*, Cambridge, MA, MIT Press.
- Melitz, M. 2003. "The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity." *Econometrica*, 71(6):1695-1725.

- Naughton, B. 1997. The China Circle: Economics and Technology in the PRC, Taiwan, and Hong Kong, Brookings Institute Press.
- Naughton, B. 2004. "China's Economic Growth and Technology Development: International Linkages and Implications for the U.S." Prepared Statement for a hearing before the U.S-China Economic and Security Review Commission, One Hundred Eighth Congress, Second Session.
- Ng F. and A. Yeats. 1999. "Production Sharing in East Asia: Who Does What, for Whom and Why?" World Bank, Policy Research Working Paper, WPS 2197, 1999.
- Rodrik, Dani. 2006. "What is so Special About China's Exports?" *China & World Economy*, Volume 14, Issue 5, September/October 2006, Pages: 1–19.
- Ravn, M. and H. Uhlig. 2002. On Adjusting The Hodrick-Prescott Filter For The Frequency of Observations, *The Review of Economics and Statistics*, 84(2): 371–380.
- Statistical Analysis System (SAS). SAS Institute Inc., 100 SAS Campus Drive, Cary, NC 27513-2414, USA (www.sas.com).
- Schott, Peter. 2008. "The Relative Sophistication of Chinese Export." *Economic Policy*, 53, January, 5–49.
- Somwaru, A., F. Tuan, M. Gehlhar, S. Langley and X. Diao. 2007. "Forces Behind China's Surging Trade: Competitiveness or Policy Driven?" Selected paper, International Agricultural Trade Research Consortium, Summer Symposium, Beijing, China, July 8-9.
- Somwaru, A., F. Tuan, M. Gehlhar, X. Diao and J. Hansen. 2008. "Developing Country Trade: Implications of China's Changing Trade and Competitiveness in Intensive and Extensive Margin Goods." Selected paper, American Agricultural Economics Association, Annual Meeting, July 27-29, Orlando, Florida.
- Tan, K. S. and H. E. Khor. 2006. "China's Changing Economic Structure and Implications for Regional Patterns of Trade, Production and Integration." *China & World Economy*, Volume 14, Issue 6, November–December 2006, Pages: 1–19.
- UNCTAD/WTO. 2010. International Trade Center, Market Analysis and Research, Difference between the Standard International Trade Classification (SITC) and the Harmonized System (HS) http://legacy.intracen.org/mas/sitchs.htm
- United Nations. (2010). Commodity Trade Statistics Database (UNCOMTRADE), Statistic Division of United Nations, http://comtrade.un.org/db/
- United Nations. 2006. Department of Economic and Social Affairs, Statistics Division, Statistical Papers, Series M No. 34/Rev. 4, Standard International Trade Classification, Revision http://unstats.un.org/unsd/publication/SeriesM/SeriesM 34rev4E.pdf
- United Nations. 2010. Statistics Division, International Family of Economic and Social Classifications, http://unstats.un.org/unsd/cr/registry/regso.asp?Ci=50&Lg=1
- United Nations, (2010). Conference on Trade and Development, UNCTADSTAT, Country Classification, http://unctadstat.unctad.org/EN/Classifications/UnctadStat.DimCountries.Indi-vidualEconomiesList.Classification.pdf
- Varian, H. 1996. Computational Economics and Finance: Modeling and Analysis with Mathematica. Springer-Verlag, New York, Inc.
- World Development Indicators (WDI), International Bank for Reconstruction and Development (IBRD), World Databank, Database, 2013 http://databank.worldbank.org/data/views/variableSelection/selectvariables.aspx?source=world-development-indicators#c\_c

# Appendix Table 1

South and South East Asian Country Group	Central & South America Country Group	High-Income Asian Country Group
Bangladesh	Argentina	Japan
Burundi	Barbados	South Korea
Cambodia	Bermuda	Taiwan
Sri Lanka	Bolivia	Singapore
Indonesia	Brazil	Malavsia
Nepal	Costa Rica	
Philippines	Cuba	<b>Ocean Country Group</b>
India	Dominica	Australia
Viet Nam	Dominican Republic	New Zealand
Thailand	Ecuador	
Laos	El Salvador	<b>Oil-Exporting</b>
Pakistan	Guatemala	<b>Country Group</b>
	Guyana	
	Haiti	Bahrain
	Honduras	Canada
	Jamaica	Iran
	Mexico	Iraq
	Nicaragua	Kuwait
	Panama	Oman
	Paraguay	Nigeria
	Peru	Qatar
	Antigua and Barbuda	Saudi Arabia
	Bahamas	Venezuela
	Belize	Yemen
	Chile	
	Colombia	
	Suriname	
	Grenada	
	Uruguay	

Transition Economies Country Group	European Union Country Group
CEE Albania	EU12-
Bulgaria	Austria
Croatia	Belgium
Czech Republic	Finland
FYR Macedonia	France
Hungary	Germany
Poland	Greece
Romania	Ireland
Slovak Republic	Italy
Slovenia	Luxembourg
Estonia	Netherlands
Latvia	Portugal
Lithuania	Spain
Armenia	EU15
Azerbaijan	Austria
Belarus	Finland
Georgia	Sweden
Kazakhstan	EU25
Kyrgyz Republic	Cyprus
Moldova	Czech Republic
Russia	Estonia
Tajikistan	Hungary
Turkmenistan	Latvia
Ukraine	Lithuania
Uzbekistan	Malta
Cambodia	Poland
Laos	Slovakia
Vietnam	Slovenia





# A REVIEW OF ONLINE CROWDSOURCING PLATFORMS

**EVANGELOS MOURELATOS\*** MANOLIS TZAGARAKIS EFTHALIA DIMARA University of Patras, Greece

#### Abstract

Over the years a great number of different websites have emerged that offer crowdsourcing services, which aim at taking advantage of the vast number of anonymous workers globally available to complete tasks. In this paper we review and analyze existing websites providing crowdsourcing services in an attempt to establish a framework that will allow systematic discussion, comparison and assessment of existing crowdsourcing platforms.

JEL Classification: A19, C14, J49, L15 Keywords: Online Labour, Websites, Crowdsourcing, Review, Evaluation Criteria.

\*Corresponding Author: Evangelos Mourelatos, University of Patras, Department of Economics, 26504 Rio, GREECE e-mail: vagmour@upatras.gr

## 1. Introduction

The volume of knowledge sharing online via the World Wide Web (WWW) is exponentially increasing. In today's WWW environment, users exchange knowledge and opinions by using discussion fora, social networks, as well as a variety of collaboration support systems. The ubiquity of such WWW and users' large-scale interaction make it possible to characterise these environments as exhibiting "collective intelligence" (*Malone, 2009*), defined as "a universally distributed intelligence, constantly enhanced, coordinated in real time, and resulting in the effective mobilization of skills" (*Levy, 1997*).

While in the abovementioned environments collective intelligence emerges rather implicitly, there are attempts to explicitly harness and exploit such collective intelligence in today's WWW settings. One such example is websites supporting "crowdsourcing", which rely on the motto "Everyone Knows Something" (Adamic et al. 2008). Crowdsourcing is the practice of obtaining services, ideas or content needed and solutions to problems in general by soliciting contributions from a large group of people, and, in particular, from an online community, rather than from traditional employees or suppliers (Zhao, Zhu 2012). This process is often used to subdivide tedious work or to fund-raise start-up companies and charities, and may also occur offline. Crowdsourcing combines the efforts of numerous self-defined volunteers or part-time workers, where each contributor of their own initiative adds a small portion to the greater result. Practically, during crowdsourcing, a user (called the "requester") requests an amount of work - which can be associated with some form of payment from an open, undisclosed set of contributors (called "workers"). Workers can browse through existing tasks and complete them for a monetary payment set by the requester who can ask that workers fulfil certain qualifications before engaging in the task at hand. They may also set up a test in order to verify workers' qualifications. Requesters can also accept or reject the 'product' sent by a worker, which reflects on the worker's reputation. Crowdsourcing has proven to be well-suited for accomplishing certain kinds of tasks. (Mason & Duncan 2009)

Over the years a great number of different websites have emerged (e.g. Amazon Mechanical Turk, Crowdflower, Microworkers) that offer crowdsourcing services which focus on specific tasks, which range from general purpose simple chores to research and development assignments. As the number of crowdsourcing websites is today rapidly increasing, research efforts concentrate on examining and analysing this new way of providing labour, while, at the same time, addressing problems that may arise. No attempt has been made so far to group and classify existing crowdsourcing websites in order to discuss and reflect on them in a systematic manner (*Howe 2008*). The overall purpose of our research, recently conducted at the Department of Economics of the University of Patras, is to investigate various economic aspects of this new form of work currently becoming all the more important. As a first step, our

E. MOURELATOS, M. TZAGARAKIS, *et al.*, South-Eastern Europe Journal of Economics 1 (2016) 59-74

aim is to review and analyse existing websites providing crowdsourcing services in an attempt to establish a framework that will allow systematic discussion, comparison and assessment of existing crowdsourcing platforms. The framework also aims at helping potential requesters in their efforts to choose a suitable platform. Our approach reviews a number of popular crowdsourcing websites and assesses them, drawing qualitative conclusions concerning their aims and purposes. Our approach focuses, in particular, on the scope, the services provided and the quality assurance techniques of existing websites offering crowdsourcing services. When using these results, we aim to outline a framework to serve as a first footing in facilitating discussion on the provision of this new kind of labour *(Bradham 2009)*.

#### 2. Motivation and Related Work

In recent years crowdsourcing has attracted the interest of researchers from various fields, who aspire to survey, analyze, comprehend or, even, improve this new form of labour. Research has focused on how platforms offering crowdsourcing services recruit and retain users as workers to enhance their smooth operation (*Doan et al. 2011*). Other research investigates models of workers supplying labour to paid crowdsourcing projects in an attempt to estimate worker's reservation wage (*Horton et al. 2010*). Discussions also revolve around workers' incentives when they participate in crowdsourcing platforms (*Kaufmann & Schulze 2011*), as well as around the relationship workers have with their own performance (*Winter & Duncan 2009*). Moreover, many surveys have been conducted to investigate the demographics of both requesters and workers who interact in the context of crowdsourcing (*Ipeirotis 2010*). The methodology employed by the research reported above includes conducting experiments and surveys by using the services of crowdsourcing platforms such as Amazon Mechanical Turk and Clickworkers (*Paolacci et al. 2010*).

While existing research focuses mainly on examining issues related to workers and requesters, the websites offering crowdsourcing services and their impact on all participants have not received due attention. Mapping services provided and the ways of WWW crowdsourcing may prove beneficial in establishing a framework for systematic discussion, comparison and assessment of such methods. Moreover, it may also facilitate interpreting some of the results reported in relevant literature.

## 3. Methodology

We selected a number of popular websites offering crowdsourcing services for review. In selecting these websites, we were guided by the following criteria:

a) Language: All crowdsourcing websites reviewed had to present their services in English. This facilitated the work of assessing services provided and comprehending their use.

- E. MOURELATOS, M. TZAGARAKIS, et al., South-Eastern Europe Journal of Economics 1 (2016) 59-74
  - b) Presentation of services provided: Websites had to provide the information required in order to facilitate their review.
  - c) Information needed for completing the review had to be offered. Many websites do not disclose all information required and such websites were excluded from our analysis.

Based on the criteria above, 98 websites were selected for analysis. Top websites were selected based on their Alexa ratings on 15/01/2014 (Alexa, 2014). The full list of websites reviewed is presented in the Appendix.

## 3.1 Website criteria

The websites selected were assessed against a number of criteria, which aim to capture various aspects of the services offered. These criteria cover technical as well as operational features of the websites reviewed. Below we present the criteria in greater detail:

- i. Type of service provided. Services provided by websites were grouped into the following ten categories (Shenk, Guittard 2011)
  - a. Microworks/Simple tasks, which are considered the smallest unit of work in a virtual assembly line, e.g. categorisation, tagging, Web research, transcription, etc.
  - b. Crowdfunding, which is the collection of finance from backers (the crowd) to fund an initiative (project). Crowdfunding has its origins in the concept of crowdsourcing, which is the broader concept of an individual reaching a goal by receiving and leveraging small contributions from many parties. Crowdfunding is the application of this concept to collect funds through small contributions by many parties in order to finance a particular project or venture.
  - c. Mobile crowdsourcing services, which are applications for mobile phones based on the "crowd".
  - d. Content Generation services, in which content is generated by the crowd. This method is becoming increasingly popular because it offers an alternative to content creation and content curation.
  - e. Data Entry services, which are projects using many different modi operandi, e.g. Excel, Word, electronic data processing, typing, coding and clerical assignments.
  - f. High knowledge intensity services, which are specialised services in particular fields, e.g. health, law, insurance, consultancies, data management, market research and cloud applications.
  - g. Program developing services, which focus on having software implemented by the crowd.
  - h. Web and graphic design services, which use the crowd contribution in the creation of artistic projects.

62

E. MOURELATOS, M. TZAGARAKIS, *et al.*, South-Eastern Europe Journal of Economics 1 (2016) 59-74

- i. Translation services, which aim at translating content from a source language into a target language.
- j. Product reviews and testing, in which such tasks are conducted by the crowd
- ii. *Open Call.* This reflects whether the task a requester submits solicits contribution from anyone. (*Brabham 2008*)
- iii. Quality & Reliability. This variable is used to report which techniques the website employs to ensure the quality of results provided by workers. It also includes the techniques the platform provides in order to ensure workers' reliability. (Wang et al. 2013)
- iv. *Provision of APIs (Application Programming Interfaces)*. This investigates whether the platform provides an API or not. In general, APIs allow services provided through the website to be integrated in third-party applications and facilitate interaction with the website to programs rather than only humans.(*Linares-Vásquez et al. 2014*)
- v. Country. Indicates the country of origin the platform is operating in. (Ross et al. 2010)
- vi. *Monetary reward*. Indicates whether the website allows monetary rewards for tasks completed by workers. (*Acar & Ende 2011*)
- vii. *Free Trial*. Indicates whether the platform provides a demo use of its services or not. (*Chiou et al. 2010*)
- viii. *Online help*. This variable examines the means a crowdsourcing platform offers in order to help users (both requesters and workers) better understand how the platform works and to engage new users. (*Zhou et al. 2009*)
- ix. *Case studies*. This variable helps examine whether the platform reports success stories as a proof of the usefulness and trustworthiness of its services. (*Thies et al. 2011*)
- x. *Job Openings*. This variable examines whether the website is currently offering job openings or not. (*Taylor 2015*)
- xi. *Social Sites*. This variable reflects the social network profile each platform uses as a tool of marketing. (*Thackeray et al. 2008*)
- xii. *Task visibility*. This variable indicates whether it is possible for visitors/guests to the platform to see all tasks offered by requestors or not. (*Eickhoff & Vries 2013*)

## 4. Results

We have reviewed and analysed the 98 crowdsourcing platforms selected on the basis of the factors outlined in the previous section. Each website was visited and both worker and requestor accounts were opened in order to assess the necessary variables. Furthermore, relevant information was also gathered from the presentation of services provided on the websites and from relevant literature research. We reviewed each of the 98 crowdsourcing platforms selected and have derived some descriptive statistics in order to get an initial high-level view of the current state of the provision of such services. In the next section, we present some findings of this analysis related to the following aspects:

• Distribution of crowdsourcing services in a range of countries

• Distribution of services provided by platforms reviewed

• Quality assurance and reliability techniques currently deployed by crowdsourcing platforms.

# 4.1 Country

Table 1 and Figure 1 show the percentage of websites reviewed operating in different geographic regions and specific countries.

		Frequency	Percentage	
	USA	66	67.3	
	Germany	6	6.1	
	England	5	5.1	
	Canada	2	2.0	
Valid	India	1	1.0	
	Switzerland	4	4.1	
	Australia	2	2.0	
	Hong Kong	2	2.0	
	Other	10	10.2	
	TOTAL	98	100.0	

**Table 1.** Percentage of websites reviewed that operate in each country

## Figure 1. Percentage of websites reviewed in each region



E. MOURELATOS, M. TZAGARAKIS, *et al.*, South-Eastern Europe Journal of Economics 1 (2016) 59-74

The majority of platforms (6.3%) currently operate in the USA, while the number of websites operating in Europe is lagging behind (15%). Most European sites in operation are located in Germany (6.1%).

#### 4.2 Services provided

This sub-chapter presents the kind of services provided by the websites reviewed (*Yang et al. 2008*). Table 2 summarises findings, showing the percentage of websites for each service category. Below is a simple frequency table indicating how services are distributed depending on the number of platforms providing them.

		Services Provided								
	mwk	crf	mcw	cntg	dte	hts	pdvp	dsns	trs	rtp
Frequency	27	31	4	11	15	16	9	14	5	10
Percentage (%)	27.6	31.6	4.1	11.2	15.3	16.3	9.2	14.3	5.1	10.2

Table 2. Percentage of each kind of services provided by websites reviewed

*Note:* (mwk=microworks/simple tasks, crf=crowdfunding, mcw=mobile crowdsourcing, cntg=content generation, dte=data entry, hts=high knowledge intensity, pdvp=program developing, dsns= web and graphic design, trs=translation, rtp=product review and testing).

Most top crowdsourcing websites (31.6%) focus on providing crowdfunding services while only a small percentage offers mobile crowdsourcing services (4.1%).

Examining the type of services provided by websites in relation to their geographic location reveals some interesting results. Table 3 presents this information.

		Percent Table (%)								
	mwk	crf	mcw	cntg	dte	hts	pdvp	dsns	trs	rtp
USA	30.3	28.8	4.5	12.1	18.2	18.2	7.6	12.1	4.5	12.1
Europe	26.7	26.7	6.7	20	13.3	26.7	0	6.7	6.7	13.3
Rest of the world	17.6	47.1	0	0	5.9	0	23.5	29.4	5.9	0

Table 3. Percentage of each kind of services provided in each region

*Note:* (mwk=microworks/simple tasks, crf=crowdfunding, mcw=mobile crowdsourcing, cntg=content generation, dte=data entry, hts=high knowledge intensity, pdvp=program developing, dsns= web and graphic design, trs=translation, rtp=product review and testing).

The results in Table 3 indicate that different regions focus on different services. While in the USA and Europe the prevailing service is supporting microwork (USA 30.3% and Europe 26.7%) followed by crowdfunding services (USA 28.8% and Europe 26.7%), the Rest of the world websites appear to focus their attention on supporting

65

crowdfunding (47.1%) followed by web & graphic design (29.4%) services. Furthermore, some services are supported only by websites operating in certain geographic areas, as is the case with content generation services (cntg). Content generation services are provided only by sites operating in the USA (12.1%) and Europe (20%), while websites in the rest of the world do not seem to support such activities at all.

In general, crowdsourcing websites do not focus on providing exclusively only one kind of service. Many platforms provide more than one service (30.61% of the total number of platforms reviewed).

# 4.3 Quality and Reliability

A critical factor of any crowdsourcing platform is the mechanisms in place for ensuring the quality of work provided by workers, particularly when they are operating in an open participation model. We surveyed the sites selected with respect to the mechanisms they implement to ensure the quality of work undertaken.

To this end we enumerated and categorised all existing quality assurance mechanisms, as presented in the table below (Table 4).

qrl	Not providing any information with respect to quality assurance
qr2	Support for reviews and ratings of jobs carried out by workers
qr3	Workers' Profiles indicating their skills and ratings
qr4	Skill tests and/or practice tests for workers so that the most suitable worker for a task may be selected
qr5	Mechanisms to detect spamming and plagiarism

Table 4. Categorisation of all existing quality assurance mechanisms

We assessed each website selected against these five categories. Surveys showed that half of the platforms examined (49 platforms) do not provide any mechanism to assess the quality of workers' jobs. Moreover, when some form of quality assurance is provided, only 4.1% of websites reviewed offer more than one quality and reliability mechanism in order to assess the quality of a worker's results on a task.

 Table 5. Percentage of all existing quality assurance mechanisms

	Quality & Reliability						
	qrl	qr2	qr3	qr4	qr5		
Frequency	49	20	16	9	17		
Percentage (%)	50	20.4	16.3	9.2	17.3		

E. MOURELATOS, M. TZAGARAKIS, *et al.*, South-Eastern Europe Journal of Economics 1 (2016) 59-74

In addition, we also examined the provision of quality assurance mechanisms across geographic regions. While websites operating in the USA offer all five categories of quality control mechanisms, they seem to favour spam and plagiarism detection techniques (Table 6, qr5), as opposed to European websites, which favour workers' profiles. Furthermore, some quality assurance mechanisms can be found only in specific geographic areas, such as Skill tests (qr4), which seem to be popular in the USA and Europe, but do not appear in crowdsourcing platforms operating in other areas.

Percentage Table (%)							
qr2 qr3 qr4 qr5							
USA	37.1	25.7	20	40			
Europe	37.5	62.5	25	25			
Rest of the world	66.7	33.3	0	16.7			

Table 6. Percentage of all existing quality assurance mechanisms for each region

Finally, we examined the provision of quality assurance mechanisms with respect to the services provided by the websites selected. These results are presented in Table 7 and indicate that platforms providing "program development" (pdvp) or "web & graphic design" (dsns) services prefer quality assurance mechanisms in the form of reviews and ratings of the job done (71.4% and 71.4%, respectively). Furthermore, some quality assurance mechanisms are not used in certain service categories. For example, platforms that support "program development" or "web & graphic design" services do not offer Skill tests to assess workers, while platforms that provide translation services do not provide review and rating mechanisms to assess the work of workers.

Table 7. Percentage of al	l existing quality as	surance mechanis	sms with respect	to kind
of services				

	Percentage Table (%)									
	mwk	crf	mcw	cntg	dte	hts	pdvp	dsns	trs	rtp
qr2	42.1	16.7	50	55.6	50	33.3	71.4	71.4	0	33.3
qr3	26.3	33.3	25	22.2	25	66.7	14.3	42.9	33.3	16.7
qr4	31.6	0	50	33.3	41.7	11.1	0	0	33.3	33.3
qr5	47.4	50	50	55.6	58.3	0	28.6	14.3	33.3	50

# 4.4 Associations of services provided

The review of existing crowdsourcing websites reveals that more than 30% of sites surveyed offer more than one type of service. When investigating how services provided are associated in the context of platforms, the patterns of co-appearances in the set of types of services supported can provide useful insights into how platforms specialise their activities and gain further understanding to support categorisation efforts.

To this end, we analysed the dataset collected using the Apriori algorithm. The Apriori algorithm (Agrawal & Srikant, 1994) is a well-known and widely used data mining algorithm for discovering association rules in large datasets. Association rules are implications of the form  $X \to Y$  where X (called the antecedent) and Y (called the consequent) are disjoint sets of attributes indicating that whenever a set of attributes X are present in an observation, then the same observation is highly likely to will also contain Y. An association rule does not imply causality; it simply indicates a highly probable co-occurrence of attribute X and Y sets. Strong rules are based on two metrics, namely the support and the confidence of a rule. Support of the rule X $\rightarrow$  Y is defined as the percentage of observations that contain the attribute set X U Y, i.e. both attribute sets, while the confidence of a rule is defined as the percentage of observations containing the attribute set in the set of observations containing X and can be perceived as the conditional probability P(Y|X). Desirable thresholds for support and confidence are entered as an input to the Apriori algorithm. Furthermore, among strong association rules, the lift (or interest) of a rule (Brin et al, 1997) is a metric to assess a rule's interestingness i.e. how interesting a rule is; this is calculated as the ratio of the rule's confidence to the expected confidence, considering the antecedent and the consequent of the rule independent. The lift of a rule indicates how many times more likely it is that the attributes in the rule would occur together than these attributes occurring separately in the dataset or together at random. A lift value greater than 1.0 implies that the relationship between the antecedent and the consequent in the rule is more significant than would be expected if the two attribute sets were independent. In general, the larger the lift ratio, the more significant the association of the two attribute sets. The Apriori algorithm was applied to websites surveyed examining only variables capturing the type of service provided. The algorithm was executed with a support value of 0.2 (support=0.2) and a confidence of 1. Table 8 shows the association rules discovered that meet the support and confidence threshold with a lift greater than 1, after redundant rules had been pruned.

E. MOURELATOS, M. TZAGARAKIS, *et al.*, South-Eastern Europe Journal of Economics 1 (2016) 59-74

No	Association rule	Support	Confidence	Lift
1	{mwk=Yes} => {crf=No}	0.27	1	1.46
2	${crf=Yes} \Longrightarrow {mwk=No}$	0.31	1	1.38
3	${crf=Yes} \Longrightarrow {cntg=No}$	0.31	1	1.12
4	{mwk=Yes} => {pdvp=No}	0.27	1	1.10
5	{mwk=Yes} => {hts=No}	0.27	1	1.19
6	${crf=Yes} \Longrightarrow {hts=No}$	0.31	1	1.19
7	{mwk=Yes} => {dsns=No}	0.27	1	1.16
8	{mwk=Yes} => {trs=No}	0.27	1	1.05
9	${crf=Yes} \Longrightarrow {trs=No}$	0.31	1	1.05
10	${crf=Yes} \Longrightarrow {rtp=No}$	0.31	1	1.11
11	${crf=Yes} \Rightarrow {dte=No}$	0.31	1	1.18

0.31

0.23

0.57

0.25

0.21

1

1

1

1

1

1.04

1.04

1.18

3.16

**Table 8.** Strong and interesting association rules mined from the websites surveyed.

 Support, confidence and lift for each rule are presented

The association rules discovered indicate that, for example, websites that support Microwork ({mwk=yes}) with high probability do not support crowdfunding ({crf=No}) (rule 1). A general pattern that emerges from the association rules in Table 8 is that websites that support microwork and crowdfunding are strongly associated (high lift ratio) with not supporting any other type of service (rules 15 and 16). This indicates that platforms supporting microwork and crowdfunding are highly specialised and focus their efforts only on such activities. Hence, such platforms form a distinct class that differs considerably in focus and range from services provided on other sites.

## 4.5 Issues Encountered

12

13

14

15

16

 $\{crf=Yes\} \implies \{mcw=No\}$ 

 $\{mwk=Yes.cntg=No\} \implies \{mcw=No\}$ 

{mwk=No,cntg=No,pdvp=No} => {dte=No}

{mwk=No,mcw=No,cntg=No,hts=No,pdvp=No,dsns=No,trs=No,rtp=No} => {crf=Yes}

{crf=No,mcw=No,cntg=No,hts=No,pdvp=No, dsns=No,trs=No,rtp=No} => {mwk=Yes}

Some important information of crowdsourcing platforms reviewed online was not readily available, which impacted the extent of this research.

In particular, existing crowdsourcing platforms did not provide quantitative data related to, for example, the total number of workers registered, the average number and volume of tasks completed per worker per day, the completion rate of tasks, etc. In order to gain insights into such aspects, a survey was conducted in the form of a questionnaire, which was sent to all websites reviewed. The questionnaire asked providers to answer questions related to various aspects of their website, such as their workforce, tasks completed and revenue earned. Fewer than 10% of websites surveyed responded to this request and completed the questionnaire. This made it impossible to include such data in the review.

## 5. Conclusions and Future Work

It is certain that the power of the crowd will drive the future of businesses, because paid crowd work offers remarkable opportunities for improving productivity (*Vukovic 2009*). Moreover, with such labour force arrangements, the global economy can complete complex tasks on demand and at a large scale with no geographic boundaries (*Schneider 2012*).

Research in this field is currently undertaken so as to investigate, understand and facilitate this new form of supplying and demanding labour. In this context, we have presented an initial attempt to review contemporary crowdsourcing websites considering how they offer their services. We reviewed 98 top websites, as ranked by Alexa, which offer crowdsourcing services and have presented descriptive statistics related to their country of operation, the kind of services provided and the mechanisms used to ensure the quality of work performed.

Review results indicate that the majority of websites are located in the US, while Europe and the rest of the world are lagging behind. Most websites provide microwork and crowdfunding services, while other kinds of services are rather under-represented. Moreover, sites offering without supporting any other type of service. Finally, while only half of the sites surveyed offer some form of quality control of work undertaken, only a very small number offers two or more such mechanisms.

The survey presented here is a first attempt at laying down a framework for discussing and analysing the kind of services that are becoming increasingly wide-spread. Future work includes laying down a conceptual framework for systematic evaluation and assessment of such service providers, as well as for designing some form of taxonomy of such kinds of services. Furthermore, we will also examine issues related to workers' incentives/motivation and issues related to quality control and efficiency *(Kittur et al. 2013).* To this end, we are currently conducting experiments in various environments (controlled laboratory settings, as well as social networking sites, such as Facebook) in order to correlate workers' incentives/motivation and quality of work *(Wang et al. 2011).* 

## References

- Acar O., Ende J., "Motivation, Reward Size and Contribution in Idea Crowdsourcing", DIME-DRUID ACADEMY Winter Conference 2011, Comwell Rebild Bakker, Aalborg, Denmark, January 20 - 22, 2011.
- Adamic L., Zhang J., Bakshy, E. & Ackerman M. "Knowledge sharing and yahoo answers: everyone knows something". WWW '08 Proceedings of the 17th international conference on World Wide Web, Beijing, China, 2008, pp 665-674
- Agrawal, R. & Srikant, R. (1994). Fast algorithms for mining association rules in large databases. Proceedings of the International Conference on Very Large Databases, Santiago, Chile, pp. 478-499. Los Altos, CA: Morgan Kaufmann.
- Alexa, 2014, http://www.alexa.com/topsites.

E. MOURELATOS, M. TZAGARAKIS, et al., South-Eastern Europe Journal of Economics 1 (2016) 59-74

Brabham D.C. "Crowdsourcing as a Model for problem Solving, An Introduction and Cases". The International Journal of Research into New Media Technologies, vol. 14(1), pp.75-90, 2008.

- Bradham D.C "Crowdsourcing the Public Participation Process for Planning Projects", Planning Theory (8: pp.242-262), August 2009.
- Brin, S., Motwani, R., Ullman, J. D. and Tsur. S Dynamic itemset counting and implication rules for market basket data. In Proceedings of the ACM SIGMOD Conference, pp 255-264, 1997.
- Chiou W., Lin C. & Perng C., "A strategic framework for website evaluation based on a review of the literature from 1995–2006", Information & Management, Volume 47, Issues 5-6, August 2010, pp 282-290.
- Doan A., Ramakrishnan R. & Halevy A. "Crowdsourcing Systems on the World Wide Web", Communications of the ACM(Volume 54 pp 86-96), New York, Usa, April 2011.
- Eickhoff C., Vries A., "Increasing cheat robustness of crowdsourcing tasks", Information Retrieval Journal, Vol. 16, Issue 2, pp 121-137, April 2013.
- Horton J. John & Chilton B. Lydia "The Labor Economics of Paid Crowdsourcing", 11th ACM Conference on Electronic Commerce, San Jose, USA, 2010, pp 209-218
- Howe J. "Crowdsourcing: Why the Power of Crowd is Driving the Future of Business" Crown Business, New York, 2008.

Ipeirotis P. "Demographics of Mechanical Turk", NYU Working Paper No.CEDER-10-01.

- Kaufmann N., Schulze T. & Veit D. "More than fun and money. Worker Motivation in Crowdsourcing - A Study on Mechanical Turk", 6th Mediterranean Conference on Information Systems (MCIS), Limassol, Cyprus, 2011, paper 340
- Kittur A., Nickerson J., Bernstein M., Gerber E., Shaw A., Zimmerman J., Lease M., Horton J. "The future of crowd work", CSCW '13 Proceedings of the 2013 conference on Computer supported cooperative work, New York, USA, 2013, pp 1301-1318
- Lévy, P., Collective Intelligence, Perseus Book, USA, 1997.
- Linares-Vásquez M. Bavota G., Di Penta M., Oliveto R., Poshyvanyk D. "How do API changes trigger stack overflow discussions? a study on the Android SDK" ICPC 2014 Proceedings of the 22nd International Conference on Program Comprehension, pp 83-94, ACM New York, NY, USA ©2014.
- Malone, T. W., Laubacher, R. & Dellarocas, C. N., Harnessing, "Crowds: Mapping the Genome of Collective Intelligence", MIT Sloan Research Paper No. 4732-09, 2009.
- Mason W. & Duncan W. "Financial Incentives and the Performance of Crowds", HCOMP '09 Proceedings of the ACM SIGKDD Workshop on Human Computation, New York, USA, 2009, pp 77-85
- Paolacci G., Chandler J., & Ipeirotis P.G. "Running experiments on Amazon Mechanical Turk", Judgment and Decision Making, 5(5), pp 411-419, 2010.
- Ross J., Irani L., Silberman M., Zaldivar A., Tomlinson B., "Who are the Crowdworkers?: Shifting Demographics in Mechanical Turk", CHI '10 Extended Abstracts on Human Factors in Computing Systems, April 10-15, 2010, Atlanta, Georgia, USA, pp 2863-2872
- Schneider, D., Moraes, K., Moreira de Souza, J., Esteves, M.G.P. "CSCWD: Five characters in search of crowds", Computer Supported Cooperative Work in Design (CSCWD), IEEE 16th International Conference on, Lisbon, Portugal, 2012, pp 634-641
- Shenk E., Guittard C. "Towards a characterization of crowdsourcing practices" Journal of innovation economics & Management (pp 93), 2011.
- Taylor J., "Crowdsourcing IT Work: A Three-Fold Perspective from the Workers, Buyers, and Platform Providers", Proceedings of the 2015 ACM SIGMIS Conference on Computers and People Research, New York, NY, USA ©2015, pp 1-2.
- Thackeray R, Neiger BL, Hanson CL, McKenzie J., "Enhancing promotional strategies within social marketing programs: use of Web 2.0 social media". Health Promot Pract 2008; 9: pp 338-43.

71

### 72 E. MOURELATOS, M. TZAGARAKIS, et al., South-Eastern Europe Journal of Economics 1 (2016) 59-74

- Thies W., Ratan A. & Davis J., "Paid Crowdsourcing as a Vehicle for Global Development", ACM CHI 2011 Workshop on Crowdsourcing and Human Computation.
- Vukovic M. "Crowdsourcing for Enterprises", Services I, World Conference on, Los Angeles, USA, 2009, pp 686-692
- Wang J., Ipeirotis P., & Provost F. "Managing Crowdsourcing Workers", Winter Conference on Business Intelligence, Utah, USA, 2011.
- Wang, Jing and Ipeirotis, Panagiotis G. and Provost, Foster, "A Framework for Quality Assurance in Crowdsourcing" (June 2013). NYU Working Paper No. 2451/31833.
- Winter M. & Duncan J.W "Financial incentives and the performance of crowds", ACM SIGKDD Workshop on Human Computation, New York, USA, 2009. pp 77-85
- Yang, J; Adamic, L; Ackerman, M, "Crowdsourcing and Knowledge Sharing: Strategic User Behavior on Taskcn", Proceedings of the 9th ACM Conference on Electronic Commerce, Chicago, USA (2008). pp 246-255
- Zhao Y., Zhu Q. "Evaluation on crowdsourcing research: Current status and future direction", Information Systems Frontiers, 2012. pp 2-3
- Zhou T., Lu Y. & Wang B., "The Relative Importance of Website Design Quality and Service Quality in Determining Consumers' Online Repurchase Behavior", Information Systems Management, Volume 26, Issue 4, 2009, pp 327-337.
# Appendix

1.	https://www.mturk.com/mturk/			
2.	http://www.clickworker.com/			
3.	https://microworkers.com			
4.	http://www.mykindacrowd.com/			
5.	http://www.kickstarter.com/			
6.	http://www.indiegogo.com/			
7.	https://www.crowdfunder.com/			
8.	https://www.crowdtilt.com/			
9.	http://www.rockethub.com/			
10.	http://www.zequs.com/			
11.	http://invested.in/			
12.	https://www.oocto.com/			
13.	https://www.somolend.com/			
14.	https://angel.co/			
15.	http://crowdflower.com/			
16.	https://www.crowdmed.com/			
17.	http://babyloan.org/fr/			
18.	http://www.cloudcrowd.com/			
19.	https://www.odesk.com/			
20.	http://www.crowdsource.com			
21.	http://www.shorttask.com/			
22.	http://www.microtask.com/			
23.	http://samasource.org/			
24.	https://www.mobileworks.com/			
25.	https://www.crowdguru.de/			
26.	http://www.optask.com/			
27.	https://www.smartsheet.com/			
28.	http://cloudfactory.com/			
29.	http://www.liveops.com/			
30.	http://www.textbroker.com/			
31.	http://www.content.de/			
32.	http://taskarmy.com/			
33.	http://www.asksunday.com/			
34.	http://www.10eqs.com/			
35.	https://www.starmind.com/			
36.	http://g2link.com/			
37.	http://www.trada.com/			
38.	http://www.taskus.com/			
39.	http://rapidworkers.com/			
40.	http://www.infetneworld.com/			
41.	http://www.agentaliytiling.com/			
42.	http://www.mykhidaciowd.com/			
43.	https://www.crowdononic.com/			
45	https://www.taskrahhit.com/			
46	http://www.trucrowd.com/			
47	http://99designs.com/			
48	http://akickincrowd.com/			

73

E. MOURELATOS, M. TZAGARAKIS, et al., South-Eastern Europe Journal of Economics 1 (2016) 59-74 74

-	
49.	http://www.openideo.com/
50.	http://ioby.org/
51.	http://startsomegood.com/
52.	http://crowdlab.anythingabout.de/index.php
53.	http://www.crowdcube.com/
54.	http://www.iron.io/
55.	http://gengo.com/
56.	https://www.tolingo.com/en
57.	http://www.lingotek.com/
58.	http://www.onehourtranslation.com/
59.	http://gigwalk.com/
60.	https://www.elance.com/
61.	http://www.script-lance.com/
62.	http://www.freelancer.com (vworkers.com)
63.	https://www.atizo.com/
64.	http://www.ideasplatform.in/
65.	http://www.causetofund.com/
66.	http://www.projektwerk.com/de/
67.	http://projectgeld.nl/
68.	http://crowdlever.com/
69.	http://www.crowdfundme.it/
70.	http://projects.csail.mit.edu/soylent/
71.	http://www.crowdcube.com/
72.	https://www.crowdbnk.com/
73.	http://www.crowdbaron.com/
74.	http://www.microtoilers.com/
75.	http://minijobz.com/
76.	http://www.crowdspring.com/
70	https://crowdin.net/
/8.	http://www.c-crowd.com/en/nome-englisch/
/9.	http://www.babyloan.org/en/
<u>80.</u> 91	http://www.crowdint.com/
01.	http://www.jobboy.com/
02.	http://www.clowdaboutnow.in
84	http://www.dvto.com/
85	https://www.denasignt.com/
86	http://www.herevolent.net/index.html
87	http://www.benevoient.net/index.ntin
88	https://www.guirky.com/
89	http://www.apphackr.com/
90	https://www.somolend.com/
91.	http://www.designcrowd.com/
92.	http://domystuff.com/
93.	http://www.utest.com/
94.	http://www.epinions.com/
95.	http://expertbids.com/
96.	http://whinot.com/
97.	https://www.crowdworx.com/
98.	http://www.fiverr.com/





## TURKISH NON-CORE BANK LIABILITIES

ENGİN YILMAZ<sup>a\*</sup>, BORA SÜSLÜ<sup>b</sup>, <sup>a</sup>Carlos III University of Madrid, Spain <sup>b</sup>Muğla Sıtkı Koçman University, Turkey

#### Abstract

Developed countries implemented a loose economic policy after the global crisis, which encouraged huge capital inflows into the emerging markets. After the global crisis, the Turkish banking system took advantage of such foreign capital inflows and experienced significant credit growth. This paper focuses on noncore liabilities, which are the sources of the credit growth in the Turkish banking system. It seems that the Turkish banking system has depended more on noncore liabilities since the beginning of 2011. Most of the non-core liabilities of the Turkish banking system are largely foreign exchange denominated and the average term structure of foreign exchange liabilities is relatively medium-term. Foreign exchange non-core liabilities of the Turkish banking system have been more sensitive to international liquidity shocks. Non-core liability growth and its medium-term foreign exchange structure are a warning signal for the Turkish banking system.

JEL Classification: E51, G21, G32 Keywords: Balance Sheet, Banks, Core Liabilities, Non-Core Liabilities, Turkey

Acknowledgements: The authors gratefully acknowledge the anonymous referees for their insightful comments and constructive suggestions on earlier drafts.

<sup>\*</sup>Corresponding Author: Engin Yılmaz, Visiting Researcher, Department of Economics, Carlos III University of Madrid, Spain. e-mail: eyilmaz@eco.ucm3.es

#### Introduction

Turkey is one of the important emerging countries and a candidate country for the European Union. Turkey has a young population compared to European Union countries, which is a great advantage, but it also has very serious structural economic problems. One of those is that Turkey's growth is dependent on international capital inflows (Basci et al. 2013; Karaçimen, 2014; Orhangazi, 2014; Akyüz, 2015; Kara, 2015; Özgür and Orhangazi, 2015) due to the inadequate domestic savings level. For instance, the saving rate in Turkey in 2013 was 13 percent, which is the lowest rate not only in the Turkish economic history but also among all emerging countries in 2013. The International Monetary Fund (IMF, 2014a) indicated that the average saving rate of emerging countries was 33 percent of their GDP, a ratio is almost three times higher than that of Turkey. Turkey experienced an unprecedented flow of foreign resources during/after the global crisis. These inflows create a vicious cycle (more foreign resource, more credit, more consumption and fewer national saving) in the economic structure resulting in volatility in the country's growth path (Süslü and Balmumcu, 2015). The gap between credits and deposits (Credits > Deposits) has been increasing in the Turkish banking system and this credit expansion arises from the growth of the banks' non-core liabilities. Turkey's credit growth coincides with changes in the composition of Turkish banking liabilities (from core to non-core liabilities). It is noted that foreign exchange (FX) liabilities are key components of the non-core liabilities of the Turkish banking system. There is significant increase in the foreign exchange short position of Turkish banks' balance sheets. All of these situations are connected to each other and have caused a dangerous vicious cycle in the Turkish banking system. Fluctuations in the economic policy of developed countries or national/international liquidity shocks may cause capital outflows and decrease the borrowing capacity of the Turkish banking system. Consequently, sudden capital outflows are a major risk for the financial stability of Turkey. The IMF Staff Report (IMF, 2014b) indicated that capital outflows remain the main risk in the Turkish economy. Banks are vital in maintaining the financial stability in Turkey because the financial system is based on banking activities and intermediation services.

This paper aims at indicating the change of the Turkish banks' balance sheets during/after the global crisis, the shift of Turkish banks' liabilities in favour of non-core liabilities and the term structure and composition of non-core liabilities. This paper intends to expose that the composition and term structure of non-core liabilities pose a threat to the Turkish banking system.

This paper is organised as follows: The literature of non-core liabilities is presented in the first section. The liabilities of the Turkish banking system are examined in the second section. Turkish non-core liabilities are evaluated in the third section. The final section presents the results.

#### 1. Literature

In economic literature, dealing with the financial cycle and financial vulnerability issues is generally focused on banking asset sheets. Credit figures, the growth of credit, the credit-to-asset ratio and the growth of the credit-to-asset ratio in the banking system have played a very important role in the financial cycle and the financial vulnerability analysis. This traditional perspective is reflected in banking regulations, which are related to preventing financial vulnerability. This focus has been given a specific form with requirements on the minimum capital for banks as a proportion of the risk-weighted assets of the bank (Hahm et al., 2012a). Excessive increase in the level of credit variables (credit figures, the growth of credit, the credit-to-asset ratio and the growth of the credit-to-asset ratio) has been perceived as an indicator of the financial cycle from this traditional perspective. Shin and Shin (2010) have indicated that this traditional perspective is not enough for understanding the financial cycle and preventing financial vulnerability. Hahm et al. (2012b) have concluded that noncore liabilities may be usefully monitored as a complementary measure to the creditto-GDP ratio. The global crisis revealed risks to the financial stability arising from the banks' reliance on certain types of non-core funding (IMF, 2013). Many researchers indicate that non-core funding is a major reason for banks' vulnerability (Demirgüç-Kunt and Huizinga, 2010; Huang and Ratnovski, 2010; Goldsmith-Pinkham and Yorulmazer, 2010; Bologna, 2011; Chen et al., 2012; Hamn et al., 2012a; Vazquez and Federico, 2012; Errico et al., 2014). Funding structures matter for financial stability because a healthy funding structure lowers the probability of a bank falling into distress (IMF, 2013).

Banking regulations should also focus on the bank liabilities' composition in addition to the bank assets' composition. This idea has been accepted after the global crisis. Generally speaking, the liability side of banks' balance sheets was analysed with regard to the term structure (short or long term) of liabilities. The new idea focuses on the core and non-core structure of bank liabilities. Recent literature suggests that the size, source, and composition of non-core liabilities provide useful insights into a financial system's health and the potential for spill-overs to the real economy (Chen et al., 2012 and Harutyunyan et al., 2015). Shin and Shin (2010) firstly argue that liabilities due to an ultimate domestic creditor (mainly households) are classified as core liabilities and liabilities due to an intermediary and a foreign creditor are classified as non-core liabilities. This distinction is based on actual claim-holders and it is usually very difficult to classify liabilities in this respect. Shin and Shin (2010) secondly refer the other distinction based on financial instruments (deposits, securities other than shares, loans, government loans, call loans, call money, financial derivatives, and other foreign debts). Finally, authors classify bank liabilities through their harmonisation (by instruments and by claimholders).

		Core Liability	Intermediate	Non-Core Liability	
				Repos	
Highly	Liquid	Demand Deposits (Households)	Demand Deposits	Call Loans	
			(Non-financial Corporate)	Short Term Foreign	
				Exchange Bank Debt	
Intermediate		Time Deposits (Households)	Time Deposits (Non-financial Corporate)	Time Deposits (Banks & Securities Firms)	
Illiquid		Trust Accounts   Covered Bonds (Households)	Trust Accounts (Non-financial Corporate)	Long Term Bank Debt Securities (Banks & Securities Firms) Mortgage Securities	

Table 1. Definition of Core and Non-Core Liabilities

Source: Shin and Shin (2010).

Shin and Shin (2010) and Kim *et al.* (2013) have defined non-core liabilities as the sum of (i) bank FX liabilities, (ii) bank debt securities, (iii) promissory notes, (iv) repos, and (v) certificates of deposit (Table 1). Hamn *et al.* (2012a) have adopted two new alternative measures for non-core liabilities. One of them includes banks' liability to the foreign sector and banks' liability to the non-banking financial sector. Other includes banks' liability to the foreign sector and the difference between M3 and M2. Hamann *et al.* (2014) have defined another two alternative non-core liabilities, which are very similar to the aforementioned definitions. One of them includes repos, bonds, foreign exchange debt, other debt and CDT<sup>1</sup>. Another definition excludes the liability position in derivatives. Harutyunyan *et al.* (2015) have defined core liabilities as bank deposits from non-financial corporations and households, while non-core liabilities as all remaining funding sources<sup>2</sup>.

Hahm *et al.* (2012b) have found that non-core liabilities are a powerful predictive indicator of financial crises and Hahm *et al.* (2012a) have clarified the relationship between credit booms and changes in the composition of bank liabilities. The 2007-2009 global banking crises showed that banks depended disproportionately on short term non-core funding (Gasperini and Rixtel, 2013). Hamann *et al.* (2014) have

<sup>1.</sup> CDT is payable to the credit of the Central Government and CDTs owned by large investors are classified under non-core liabilities.

<sup>2.</sup> Harutyunyan *et al.* (2015) have used non-core liabilities of banks as well as of non-bank financial institutions.

indicated that there is an interesting connection between bank credit/asset growth, composition of bank liabilities and bank leverage in Colombia. Harutyunyan *et al.* (2015) have calculated core and non-core liabilities for 26 countries from 2002 to 2013. Harutyunyan *et al.* (2015) have indicated that both the growth rate of the non-core liabilities of Turkey and the standard deviation of annual growth rates of the non-core liabilities of Turkey are more than those of others countries in this period.<sup>3</sup> In spite of the significance of non-core liabilities in Turkey, there are relatively few studies on non-core liabilities. Köksal and Binici (2012) have indicated that non-core liabilities are important drivers of leverage in the Turkish banking system. Kılınç *et al.* (2013) have revealed a robust relationship between credits and non-core liabilities in Turkey. Özen *et al.* (2013) have showed that the accumulation of foreign exchange denominated non-core liabilities could potentially amplify the severity of external financial stress shocks in Turkey. Akdoğan and Yıldırım (2014) have concluded that the level of non-core liabilities in the Turkish banking system is currently at moderate levels.

## 2. Turkish Bank Liabilities

The banking sector is the key-element of the financial system in Turkey. It represents 87% of the domestic financial system. The major source of borrowing in the Turkish banking system is deposits. Other sources of borrowing are payables to banks, funds from repo transactions, securities issued, other liabilities, payables to the central bank and payables to the money market.

Deposits	59.36%
Payables to Banks	17.35%
Funds from Repo Transactions	7.45%
Securities Issued	4.68%
Other Liabilities	9.79%
Payables to the Central Bank	1.12%
Payables to the Money Market	0.25%

**Table 2.** The Liabilities of the Turkish Banking System (December 2015)

Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, December 2015.

<sup>3.</sup> Harutyunyan et al. (2015): their findings are indicated in Tables 2 and 3 and Figure 8.

The key elements of bank liabilities in the Turkish banking system are presented in Table 2. Deposits, payables to banks, funds from repo transactions and securities issued come to nearly 90% of bank liabilities in the Turkish banking system. It is a well-known fact that the size of every element in the liabilities depends on the specific country characteristics. In this paper, deposits are classified as core liabilities of the Turkish banking sector. Other components of bank funding (payables to the money market, payables to banks, funds from repo transactions and securities issued) are classified as non-core liabilities of the banking sector in Turkey<sup>4</sup>.

 Core Liabilities
 Non-Core Liabilities

 Payables to Banks (BB)
 Payables to Banks (BB)

 Deposits
 Funds from Repo Transactions (REP)

 Securities Issued (IMK)
 Payables to Money Market (PPB)

Table 3. Core and Non-Core Liabilities of the Turkish Banking System

Source: Authors' definitions.

This classification is an essential tool in analysing credit growth in Turkey during/ after the global crisis. Turkey received huge capital inflows through the banking system during/after the global crisis. Banks applied to less reliable and volatile sources of funding during the global crisis in order to meet their credit demands, which intensified the vulnerability of the banking sector to systemic risks (Akdoğan and Yıldırım, 2014).

Graph 1 presents deposits as a percentage of total liabilities declined in the beginning of 2011 due to decrease in local currency (TL) deposits and foreign exchange (FX) deposits in the Turkish banking system. Deposits might limit the capacity for growth, but they constitute a sounder and more stable source of funds, not so dependent on external factors of the bank (Oliver, 2013). Turkey is an emerging country which has not solved its structural and economic problems. As a conclusion, declining deposits in the banking system exacerbate the economic vulnerability of the country.

<sup>4.</sup> Deposits represented 60% and non- core liabilities represented 30% of the liabilities of the Turkish banking system in December 2015. In this analysis, 10% of total liabilities are classified as "other liabilities".

**Graph 1.** Deposits to Total Liabilities Ratio (%) in the Turkish Banking System (2003-2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))



Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, 2015.

**Graph 2.** Credit to Total Assets Ratio (%) in the Turkish Banking System (2003-2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))



Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, 2015.

An increasing trend in credit demand (except 2008-2009) is clearly apparent in Graph 2. Credit to total assets ratio began to increase at the onset of 2011. Credit growth reached 30 percent between early 2010 and late 2011. Such rapid credit growth is a main indicator of financial and economic crises (Mendoza and Terrones, 2008; Schularick and Taylor, 2012; Dell'Ariccia *et al.*, 2012). The Central Bank of the Republic of Turkey has introduced the interest rate corridor, the reserve requirement

and the reserve option mechanism so as to lessen credit growth and credit volatility (Başçı and Kara 2011; Aysan *et al.*, 2014). Special emphasis has been placed on credit variables as an indicator of financial stability (Kara *et al.*, 2013). Emerging economies have increased their credit levels during the past decade, but Turkey has experienced a very different condition compared with other emerging countries. Escribano and Han (2015) have calculated the average credit growth rate for 32 emerging countries in the last decade<sup>5</sup> and this study showed that Turkey has had the highest annual credit growth in the last decade. We emphasise the sources of credit growth rather than the credit growth rate. All historical data of banking balance sheets in Turkey indicate that the gap between credits and deposits (Credits > Deposits) arises from the enormous growth of non-core liabilities. The equilibrium between credits and deposits in the 2003-2015 period can be seen in Graph 3. This equilibrium was in favour of deposits (Deposits > Credits) before 2011, but it changed in favour of credits > Deposits) since the beginning of 2011.

**Graph 3.** Credits and Deposits (Million Liras) in the Turkish Banking System (2003-2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))



Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, 2015.

The credits-deposits gap started in the first quarter of 2012 and increased dramatically until December 2015. When credit grows faster than the pool of available deposits, then banks turn to other sources of funding to support their credit growth (Hamn *et al.*, 2012a). There is a close correlation between credit growth and growth of non-core liabilities in the Turkish banking system. Non-core liabilities are the main drivers of this gap.

<sup>5.</sup> Annual growth rates in 2003-2012, real credit variables according to Escribano and Han (2015).

**Graph 4.** Deposits and Non-Core Liabilities to Total Liabilities Ratio (%) in the Turkish Banking System (2003-2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))



Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, 2015.

It is clearly seen that deposits, as a percentage of total liabilities, have decreased and non-core liabilities, as a percentage of total liabilities, have increased (Graph 4). Deposits as a percentage of total liabilities decreased from 73% in 2003 to 60% in 2015. Non-core liabilities as a percentage of total liabilities increased from 17% in 2003 to 30% in 2015. Decreasing deposits as a percentage of total liabilities have been compensated by increasing non-core liabilities as a percentage of total liabilities.

Key elements of non-core liabilities in the Turkish banking system are presented in Graph 5. PPB represents payables to the money market, BB represents payables to banks, REP represents funds from repo and IMK represents securities issued. It is preferable to use only three of them (BB, REP and IMK) because the size of other source (PPB) is merely symbolic. The drastic drop in deposits is compensated by an enormous rise in payables to banks, funds from repo and securities issued. Most of the Turkish banking system liabilities are largely foreign exchange-dominated and this situation is particularly important for Turkey. In this perspective, the non-core liabilities can be classified as foreign exchange (FX) denominated non-core liabilities and local currency (TL) denominated non-core liabilities. **Graph 5.** Payables to the Money Market (PPB), Payables to Banks (BB), Funds from Repo (REP) and Securities Issued (IMK) to Total Liabilities (TLI) Ratio (%) in the Turkish Banking System (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))



Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, 2015.

**Graph 6.** FX | TL Payables to Banks (BB), FX | TL Funds from Repo (REP) and FX | TL Securities Issued (IMK) to Total Liabilities (TLI) Ratio (%) in the Turkish Banking System (December 2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))



Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, 2015.

The TL-BB represents the local currency denominated 'payables to banks', FX-BB represents foreign currency denominated 'payables to banks', TL-REP represents local currency denominated 'funds from repo', FX-REP represents foreign currency denominated 'funds from repo', TL-IMK represents local currency denominated 'securities issued', FX-IMK represents foreign currency denominated 'securities

issued' and TLI represents total liabilities in Graph 6. FX denominated non-core liabilities (FX-BB and FX-IMK) are greater than TL denominated non-core liabilities (TL-BB and TL-IMK). TL denominated 'funds from repo' is slightly greater than FX denominated 'funds from repo'. It is apparent that FX denominated non-core liabilities are two times greater than TL denominated non-core liabilities.

### 3. Evaluating Turkish Non-Core Liabilities

There are no international standards in the academic literature for evaluating the level of non-core liabilities, although there are several approaches for analysing and comparing the level of non-core liabilities. A first approach uses the non-core liabilities to Gross Domestic Product<sup>6</sup> (GDP) ratio. In Graph 7, it is clearly shown that there has been a progressive upward trend in non-core liabilities to GDP ratio in Turkey since the global crisis. This trend accelerates after the global crisis and coincides with the growing gap between credits and deposits.

**Graph 7.** The Non-Core Liabilities to GDP Ratio (%) in Turkey (January 2003-September 2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))



*Source:* Banking Regulation and Supervision Agency Statistics (BRSA) and Turkish Statistical Institute Statistics (TURKSTAT), September 2015.

The drawback in this ratio is that GDP is not a real time variable and it is frequently revised. Furthermore, non-core liabilities are published monthly; however, GDP is published quarterly in Turkey. A second approach uses monetary aggregates, which are financial data and are published more frequently (weekly and monthly). Shin

<sup>6.</sup> Expenditure on the Gross Domestic Product (at Current Prices) (2015 third quarter) is taken from the national account of the Turkish Statistical Institute (TURKSTAT).

and Shin (2010) used the non-core liabilities to M2 money supply ratio in analysing the financial system of Korea and concluded that this ratio reached maximum level during the global crisis and then decreased rapidly. This ratio is used to indicate the pro-cyclical nature of non-core liabilities in the Korean economy. It is seen that, in the Korean economy, this ratio reached a peak level in the economic crisis (1997 and 2009) and then sharply decreased. M2 money supply in Turkey comprises the currency in circulation, demand deposits and time deposits in the financial system.

**Graph 8.** The Non-Core Liabilities to M2 Ratio (%) in the Turkish Banking System (2006-2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))



*Source:* Banking Regulation and Supervision Agency Statistics (BRSA) and Turkish Central Bank Statistics (CBRT), 2015.

Graph 8 illustrates that the non-core liabilities to M2 money supply ratio increased from 23.6 % in 2006 to 54.8 % in 2015. Shin and Shin (2010) and Hahm *et al.* (2012a) have indicated that this ratio was nearly 50% in Korea at the onset of the global crisis and then decreased sharply. This ratio is another important indicator to determine the size of non-core liabilities in Turkey, but the drawback in this ratio is that there is no common M2 definition among the countries. Third approach uses the non-core to core ratio to determine the size of non-core liabilities.

The non-core liabilities to core liabilities ratio increased from 24% in January 2003 to 52% in December 2015 (Graph 9). This ratio slowly increased between 2003 and 2010 and it was only 28% in December 2010. However, it began to increase dramatically in January 2011 and it reached 52% in December 2015.

**Graph 9.** The Non-Core Liabilities to Core Liabilities Ratio (%) in the Turkish Banking System (2003 -2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))



Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, 2015.

**Table 4.** Elements of Non-Core Liabilities in the Turkish Banking System (%) (December 2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))

BB / Non-Core Liabilities	REP / Non-Core Liabilities	IMK / Non-Core Liabilities	
58.85%	25.27%	15.89%	

Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, 2015.

The elements of non-core liabilities in the Turkish banking system are shown for December 2015 (Table 4). Most of non-core liabilities are composed of payables to banks. The expansion of non-core liabilities originated from the enormous growth of payables to banks. Non-core liabilities should be analysed from the point of FX and TL composition.

**Table 5.** FX and TL Composition of Non-Core Liabilities in the Turkish Banking System (%) (December 2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))

	BB	REP	IMK
FX	90.1%	30.5%	70.4%
TL	9.9%	69.5%	29.6%

Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, 2015.

FX and TL composition of non-core liabilities in the Turkish banking system are shown for December 2015 (Table 5). FX denominated "payables to banks" and "securities issued" are greater than TL denominated "payables to banks" and "securities issued", whereas TL denominated "funds from repo" is greater than FX "funds from repo". It is clearly seen that, in December 2015, FX denominated non-core liabilities were greater than TL denominated non-core liabilities in the Turkish banking system. There is no doubt that the distinction between short-term and long-term non-core liabilities is significant for the detailed analysis of non-core liabilities. The term structure of non-core liabilities is presented in Table 6.

**Table 6.** The Term Structure of Non-Core Liabilities in the Turkish Banking System (Million Liras) (December 2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))

	FX BB	FX REP	FX IMK	TL BB	TL REP	TL IMK
Short	133.456	48.023	6.255	11.011	109.425	19.888
Long	176.657		70.870	4.948		8.875

*Source:* Banking Regulation and Supervision Agency Statistics (BRSA) and Central Bank of the Republic of Turkey Statistics (CBRT) and The Banks Association of Turkey Statistics (BAT), Turkey, 2015.<sup>7</sup>

Short-term non-core liabilities were greater than long-term non-core liabilities in December 2015. The term structures of FX and TL denominated non-core liabilities are significantly different. The short term TL denominated non-core liabilities are greater than the long term TL denominated non-core liabilities. The long term FX denominated non-core liabilities are slightly greater than the short term FX denominated non-core liabilities. It is clear that the average term structure of foreign exchange liabilities is relatively medium-term. This situation is very important because the Turkish banking system has widened the net FX short position since the beginning of 2011.

<sup>7.</sup> FX - BB values are provided by CBRT (Outstanding Long Term Loans Received From Abroad by the Private Sector), TL - BB values are provided by BRSA (Sectoral Loan Distribution - Loans Extended to Banks), FX - REP | TL - REP values are provided by BRSA (Funds from Repo Transactions) and FX – IMK|TL - IMK values are provided by BAT (Bills and Bonds).

**Graph 10.** FX Assets and FX Liabilities in the Turkish Banking System (2003-2015) (Million Liras) (December 2015) (Adjusted FX Values by CPI Based Real Effective Exchange Rate (2003=100))



Source: Banking Regulation and Supervision Agency Statistics (BRSA), Turkey, 2015.

The gap  $(FX \text{ Liabilities} > FX \text{ Assets})^8$  has increased sharply since the beginning of 2011 (Graph 10). The net FX short position of the banks in the balance sheet reached -170.2 billion TL in December 2015. This gap (FX Liabilities > FX Assets) is covered by the same amount of the net long TL position (TL Assets >TL Liabilities) of the banks. The credit growth of the Turkish banking system over the past four years has been largely funded by FX non-core liabilities. The long term FX non-core liabilities are slightly more than the short term FX non-core liabilities in the Turkish banking system and, therefore, FX non-core liabilities are medium-term. The first problem is that the Turkish banking system has a net FX short position (-170.2 billion TL) and the secondly problem is that the long term FX non-core liabilities are slightly more than short term FX non-core liabilities in the Turkish banking system. The Central Bank of the Republic of Turkey (CBRT) raised the remuneration rate for the required reserves maintained in Turkish liras for the purpose of supporting core liabilities and took additional measures to extend the maturity of banks' non-core foreign currency liabilities (CBRT, 2015a). The Central Bank of the Republic of Turkey (CBRT) has claimed that the share of non-core liabilities with maturities up to one year gradually dropped and this drop was mainly substituted by liabilities with maturities of 1-3 years (CBRT, 2015a). This declaration seems to be unclear and doubtful when analysing

FX Liabilities equal Foreign Exchange General Position - Total Foreign Exchange Liabilities (million TL) and FX Assets equal Foreign Exchange General Position - Total Foreign Exchange Assets (million TL) in the Banking Regulation and Supervision Agency Statistics (BRSA).

banks' FX borrowing policy. Of the \$15.5 billion of syndicated loans that Turkey's banks have borrowed since the start of 2015, about 90 percent have a maturity that is longer than a year by between one and 10 days (Bloomberg, 2015). Turkish banks take advantage of paying fewer reserve requirements through this borrowing behaviour and Turkish banks record these loans in 1-3 years' maturity.

Any sudden change in international liquidity conditions leads to an FX funding problem which is a very dangerous situation for the Turkish banking system. The Turkish banking system should focus on creating core liabilities for financing their assets.

This paper finds out that

- The Turkish banking system depends more on non-core liabilities
- FX non-core liabilities are more than TL non-core liabilities
- Short term non-core liabilities are more than long term non-core liabilities
- Long term FX non-core liabilities are slightly more than short term FX non-core liabilities.

## 4. Conclusion

Developed countries used a loose economic policy after the global crisis and it encouraged huge capital inflows into emerging markets. The Turkish banking system took advantage of foreign capital inflows during/after the global crisis. Turkey has experienced significant credit growth and credit volatility. It has been shown that the main source of the credit growth in Turkey between 2010 and 2015 has been the high level of non-core liabilities in the Turkish banks' balance sheets. This paper is the first study in which term structure and composition of non-core liabilities are examined in Turkey. Non-core liabilities in the Turkish banking system have significantly increased after the global crisis and non-core liabilities have been the main source of credit growth. It is apparent that FX denominated non-core liabilities are greater than TL denominated non-core liabilities, short term non-core liabilities are more than long term non-core liabilities and long term FX non-core liabilities are slightly more than short term FX non-core liabilities in the Turkish banking system. Non-core liabilities have been growing dramatically in the Turkish banking system. Most of the non-core liabilities are medium-term foreign exchange denominated liabilities. This system is very susceptible to national or international liquidity problems.

### References

- Akdoğan, K. and Yıldırım, B. 2014. "Non-Core Liabilities as an Indicator of Systemic Risk and a Liquidity Stress Test Application on Turkish Banking System", *CBRT Working Paper Series*, 14/12.
- Akyüz, Y. 2015. "Internationalization of Finance and Changing Vulnerabilities in Emerging and Developing Economies", *UNCTAD Discussion Paper*, No. 217, 2014.

- Aysan, A., Fendoğlu, S. and Kılınç, M. 2014. "Managing Short-Term Capital Flows in New Central Banking: Unconventional Monetary Policy Framework in Turkey", *CBRT Working Paper Series*, February 2014.
- Başçı, E., Akçelik, Y., Aysan, A. and Oduncu, A. 2013. "The Turkish Approach to Capital Flow Volatility", *CBRT Working Papers*, No:13/6, 2013.
- BAT, 2015. The Banks Association of Turkey, https://www.tbb.org.tr/en
- Bloomberg, 2015. "The Two-Day Trick Masking Turkey's Short-Term Debt Habit", available at http://www.bloomberg.com/news/articles/2015-10-04/the-two-day-trick-masking-turkey-s-short-term-debt-habit, (accessed 18 March 2016)
- Bologna, P. 2011. "Is There a Role for Funding in Explaining Recent U.S. Banks' Failures?", *IMF Working Paper*, No. 11/180.
- BRSA, 2015. Banking Regulation and Supervision Agency Statistics, www.bddk.org.tr
- CBRT, 2015. Central Bank of the Republic of Turkey Statistics, http://evds.tcmb.gov.tr
- CBRT, 2015a. "Financial Stability Report", Volume 21, November 2015.
- Chen, S., Philip, L., Andrea, M., Chris, S. and Hyun, S. 2012. "Exploring the Dynamics of Global Liquidity", *IMF Working Paper*, 12/246.
- Dell'Ariccia, G., Igan, D., Laeven, L. and H, Tong. 2012. "Policies for Macro Financial Stability: How to Deal with Credit Booms", *IMF Staff Discussion Note*, SDN/12/06.
- Demirgüç-Kunt, A. and Huizinga, H. 2010. "Bank Activity and Funding Strategies: The Impact on Risk and Returns", *Journal of Financial Economics*, Vol. 98, No. 3, pp. 626–50.
- Errico, L., Artak H., Elena L., Richard W., Yevgeniya K., Goran, A. and Hanan, A. 2014. "Mapping the Shadow Banking System Through a Global Flow of Funds Analysis", *IMF Working Paper*, No. 14/10.
- Escribano, G.M. and Han, F. 2015. "Credit Expansion in Emerging Markets", *IMF Working Papers*, WP/15/212.
- IMF, 2013. "Global Financial Stability Report: Transition Challenges to Stability", World Economic and Financial Surveys, 2013.
- IMF, 2014a. "World Economic Outlook: Recovery Strengthens, Remains Uneven", *IMF Reports*, April 2014.
- IMF, 2014b. "2014 Article IV Consultation—Staff Report; Press Release; and Statement By The Executive Director For Turkey", *IMF Country Report No. 14/329*, 2014.
- Hahm, H. J., Shin, H. S. and K, Shin. 2012a. "Non-Core Bank Liabilities and Financial Vulnerability", *NBER Working Paper Series*, 18428.
- Hahm, H. J., Shin, H. S., Mishkin, S. F. and K. Shin. 2012b. "Macroprudential Policies in Open Emerging Economies", *NBER Working Paper Series*, 17780.
- Hamann, F., Hernandez, R., Silva, L. and Tenjo, G.F. 2014. "Leverage Pro-Cyclicality and Bank Balance Sheet in Colombia", *Ensayos sobre Politica Economica*, No:32, P:50-76.
- Harutyunyan, A., Massara, A., Ugazio, G., Amidzic, G. and Walton, R. 2015. "Shedding Light on Shadow Banking", *IMF Working Paper*, No. 15/1.
- Huang, R. and Ratnovski L. 2010. "The Dark Side of Bank Wholesale Funding", *Journal of Financial Intermediation*, Vol. 20, No. 2, pp. 248–63.
- Gasperini G. and Rixtel, A. 2013. "Financial Crises and Bank Funding: Recent Experience in Euro Area", *BIS Working Papers*, No:406, March 2013.
- Goldsmith-Pinkham, P. and Yorulmazer, T. 2010. "Liquidity, Bank Runs, and Bailouts: Spillover Effects During the Northern Rock Episode," *Journal of Financial Services Research*, Vol. 37, No. 2–3, pp. 83–98.
- Schularick, M. and Taylor, A.M. 2011. "Financial Crises, Credit Booms, and External Imbalances: 140 Years of Lessons", *IMF Economic Review*, 59 (2, 2011): 340-378.

- Kara, H., Tiryaki, S. T. and Yüksel, C. 2013. "In Search of a Reasonable Credit Growth for Turkey", *CBRT Research Notes in Economics*, No: 13/03.
- Kara, H. 2015. "Turkey's Experience with Macroprudential Policy", *Macroprudential Policy: Effectiveness and Implementation Challenges CBRT-IMF-BIS Joint Conference*, October 26-27, 2015.
- Karaçimen, E. 2014. "Financialization in Turkey: The Case of Consumer Debt", *Journal of Balkan* and Near Eastern Studies, Vol. 16, No.2, 1611-180, 2014.
- Kılınç, Z., Karasoy, G.H. and Yucel, E. 2013. "Non-Core Liabilities and Credit Growth", CBRT Working Papers, 13/24, 2013.
- Kim, J.H., Shin, H.S. and Yun, J. 2013. "Monetary Aggregates and the Central Bank's Financial Stability Mandate", *International Journal of Central Banking*, January 2013.
- Köksal, B. and Binici, M. 2012. "Is the Leverage of Turkish Banks Procyclical ?", *Central Bank Review*, Vol. 12 (July 2012), pp.11-24.
- Mendoza, E. G. and Terrones M. E. 2008. "An Anatomy of Credit Booms: Evidence from Macro Aggregates and Micro Data", NBER Working Paper, 14049.
- Oliver, A. M. 2013. "Financial Integration and Structural Changes in Spanish Banks during the Pre-Crisis Period", *Revista de Estabilidad Financiera*, 24, 109-123.
- Orhangazi, Ö. 2014. "Capital Flows and Credit Expansions in Turkey", *Review of Radical Political Economics*, 46:509, 2014.
- Özen, E., Sahin, C. and Unalmis, I. 2013. "External Financial Stress and External Financing Vulnerability in Turkey: Some Policy Implications for Financial Stability", *Central Bank Review*, 13, 65-74.
- Özgür, G. and Orhangazi, Ö. 2015. "Capital Flows, Finance-Led Growth and Fragility in the Age of Global Liquidity and Quantitative Easing: The Case of Turkey", *PERI Working Papers Series*, Number:397, September 2015.
- Schularick, M. and Taylor, A. M. 2012. "Credit Booms Gone Bust: Monetary Policy, Leverage Cycles, and Financial Crises, 1870-2008", *American Economic Review*, 102(2).
- Shin, H.S. and Shin, K. 2010. "Procyclicality and Monetary Aggregates", NBER Working Papers, 16836, 2014.
- Süslü, B. and Balmumcu, Ö. 2015. "Saving Gap for Turkey's Economy and Monetary Policy", International Journal of Management Economics and Business, 11(25). (in Turkish)
- TURKSTAT, 2015. Turkish Statistical Institute, http://www.turkstat.gov.tr/Start.do
- Vazquez, F. and Federico, P. 2012. "Bank Funding Structures and Risk: Evidence from the Global Financial Crisis", *IMF Working Paper*, No. 12/29, 2012.



# **Book Review**

## *Economic and Policy Foundations for Growth in South East Europe: Remarking the Balkan Economy*

edited by Adam Benett, G. Russel Kincaid, Peter Sanfey, Max Watson published by Palgrave Macmillan, United Kingdom, 2015, pp. 139 ISBN: 978–1–137–48835–0 epub, ISBN: 978–1–137–48834–3 pdf) ISBN: 978–1–137–48833–6 Hardback

reviewed by George Makris\*

The book by the writers mentioned above provides an invaluable analysis of the transition south-eastern European countries recently experienced from a planned economy system to market economy and the democratic functioning of society. The book is also a useful tool aiming at analyzing the techniques through which a more effective implementation of reforms and practices will lead to the economic growth of these countries. The need for a more carefully designed economic policy has become obvious, especially after the global financial crisis reached the region in 2008/09 and following the recent turmoil within the Eurozone. This is necessary if we want to avoid the distortions and impasses caused by the transition model formed in the course of twenty five years. These problems were not readily visible during the euphoric second decade of the period in question. More specifically, the need to strengthen the existing economic policy of the countries of the region is supported, through targeted and coordinated assistance from relevant expertise of international institutions. In addition, EU's development of more efficient practices to implement reforms is supported, particularly in regard to its architecture and the architecture of the Eurozone, the aim being to accelerate European integration.

The first chapter of the book describes the economic model of the region, as this was formed during the transition period, which identifies three stages: the difficult implementation of the first decade's reforms; the rapid growth that ensued until the global financial crisis and the Eurozone turmoil. This chapter focuses on the transmission of these recent crises to the economies of the region through their integration into the global and, mainly, the European financial system and, in particular, through the financing of domestic consumption and the impact this had on shaping the production structure of these economies.

<sup>\*</sup> Professor, School of Economic and Regional Studies, Department of Balkan, Slavic and Oriental Studies, University of Macedonia, Thessaloniki, Greece

The second chapter focuses on structural reforms carried out until the beginning of the crisis. It should be noted that a lack of enthusiasm on the part of the public in the countries concerned prevented the continuation of such reforms. This, in some cases, is masked by the encouraging developments in production and employment. According to the authors, an ongoing implementation of these reforms is deemed necessary, through using a different approach from that of past experience. This concerns both EU member states and candidate countries. Therefore, the role of the EU, as well as that of the International Monetary Fund, is vital, without underestimating the role of the EBRD, the EEIB and the World Bank within a project conditionality context.

The fiscal policy being applied and the financial institutions available - an issue whose importance was strongly felt after the transmission of the crisis into the region - are the subject of the third chapter. The need for fiscal reforms has become evident, although there are differences between countries as to the intensity of the problems caused to the budgetary balance. However, according to the authors, the best guarantee to tackle both current difficulties and future contingencies is to strengthen the financial architecture of the region, namely, the institutions and rules. In this case the recent EU experience can be particularly useful.

The fourth chapter focuses in detail on the problem of the financial sector, the role of which is considered largely responsible for the derailment of certain economies of the region during the second transition period. In that sense, the negative effects of credit expansion, which contributed towards overheating these economies, ignoring the requirements of a rational configuration of production structure, became apparent during the crises. In this chapter, after a detailed presentation of the financial sector in SEE, attention is drawn to the criteria of Optimal Currency Areas, which should be met by countries outside the Eurozone. In addition, experience on the effectiveness of macroprudential measures is gained, which, in combination with traditional macroeconomic instruments, has tried to help limit the impact of the crisis on certain countries of the region. Nevertheless, according to the authors, implementation of such macroprudential policies cannot possibly have long-term potential for success with chances of growth, unless applied in coordination by national and supranational authorities.

The fifth and final chapter of the book is a continuation of the previous one and focuses on the solutions needed so as to address the financial sector challenges to be faced in SEE. The region's particularities are portrayed in detail by the authors. What they stress is that risks are arising from the intense europisation of this sector, the extensive participation of banks in public debt or the high volatility of the economic activity in the region. These hazards, which also entail a high financial cost for non-financial sectors, can be managed by ensuring financial stability, effectively using a grid of macroprudential tools over and through the cycle. The authors consider this to be effective at a cross-border level, within the EU and with the EU, and, indeed, in the context of the new architecture of the EU financial sphere, which is analysed in detail. This book can be useful in two ways: on the one hand, it helps us understand the current state of the economies of SEE countries, as shaped over a course of twenty five years, highlighting errors and weaknesses of its dynamics. On the other hand, it recommends concrete solutions to overcome obstacles impeding these economies from long-term growth.

There is one issue that could possibly lead us to further reflection, though it is somewhat remote from the very specific goal of the book: the, albeit theoretical, connection between achieving the objectives of the nominal economic variables in the region -which is what the proposed reforms and measures seek to do- and reaching an actual convergence of the countries' economies with the European average. Indeed, this is a challenging issue that will have to be addressed in the future. This final comment, however, does not detract anything at all from the fact that this book can be of great interest to academics, practitioners and policymakers.

# **Book Review**



# Should Britain Leave the EU? An Economic Analysis of a Troubled Relationship

edited by Patrick Minford, Sakshi Gupta, Vo Phuong Mai Le, Vidya Mahambare and Yongdeng Xu published by Edward Elgar, Cheltenham, UK • Northampton, MA, USA, with the Institute of Economic Affairs, 2015, pp. 200, ISBN: 978 1 78536 032 9 (hardback)

reviewed by Christopher Tsoukis\*

This book is an effort to bring economics to bear upon the question of whether being a member of the European Union (EU), or exiting from it, is in the interests of the UK. The second edition of the book (2015) is timely, with an impending referendum on EU membership that is now set for 23 June 2016; however, it was announced several months ago (and, thus, known, when the book was published, to be due at an uncertain time in the near future). Arranged in seven Chapters, the book brings together, in a fairly impressive way, much of what economics can offer in answering the question of its title – including factual information, literature reviews, technical analyses and simulations.

Chapter 1 is a pre-amble, summarising the relationship of the UK with the EU and the conclusions arising from subsequent Chapters. Chapter 2 is a descriptive account of EU regulation in all spheres, namely, labour and product markets, financial services, social protection. A key conclusion running through the book is that regulations of all kinds have eased in recent years, while the UK remains more liberal on all fronts, with lighter regulation than the EU average; that in service markets, the EU is more interested in harmonising regulatory practices than creating greater competition; and, finally, that in the case of financial services, in particular, EU regulation threatens to weaken, often purposefully, the UK's comparative advantage. In the book's estimates (see Table 2.1), the costs of imposing protective measures like those in the EU on the UK labour market are moderate to large (mostly in the order of a few percentage points of lost output or increased unemployment).

Chapter 3 is about the benefits and costs (more likely) of the UK joining the Eurozone. This is now more of a historical question, as the prospect is not likely to arise any time soon, but the Chapter is a tour de force of relevant literature including references to

<sup>\*</sup>Keele University, Keele Management School, Facutly or Humanities and Social Sciences, Staffordshire, United Kingdom.

optimal currency areas. The main point is that a potential Euro entry would deprive the UK of the stabilising force of an independent monetary policy, but would also recalibrate the volatility of the UK currency; it would stabilise the latter versus the Eurozone (self-evidently) but increased volatility vis-à-vis the rest of the world (including the US) would ultimately be detrimental.

Chapter 4, on the cost of EU trade policies for the UK, is in many ways the core of the book. Currently, the EU is a 'common market', where there is free trade between member states and common trade policies vis-à-vis the rest of the world. The key-argument of the book is that the UK would benefit from exiting the EU and trading with the rest of the world independently, with trade policies decided by the UK in line with WTO rules and likely communication with trade partners. The big question that emerges here is what trade arrangements will arise between the UK and the EU. The discussion in pages 19-20 treats the issue as a straightforward question, arguing that, as a rule, the "EU [will levy] its usual (MFN) tariff-equivalents on us when we leave." (p. 19) Then, it is conceded, highly integrated sectors, such as the high volume car industry, are likely to seek special arrangements in order to preserve market integration. However, fixing such arrangements will certainly require protracted and costly negotiations with the EU during a period of uncertainty. Anyway, disregarding this question, it is implied that trade protection is costly, but estimates are rather small, in the order of 3-4% of one-off lost output or welfare (Tables 4.3/4). Consequently, the UK would gain from exiting and liberalising trade, but relevant estimates are hardly show-stoppers (or rather, show-movers).

The same overall conclusions emerge from Chapters 5-7, which present a detailed, informative, and quite interesting -in their own right- descriptive accounts of agriculture, manufacturing and service sectors and the state of EU trade protection for them. These sectors are interesting for a variety of reasons: agriculture is the sector subject to the heaviest trade and overall policy intervention by the EU; protection in manufacturing is weakening as (any correlation here?) the sector is becoming less important, while services are increasingly more important but still fairly heavily protected. Whatever the estimates might be (from other sources surveyed) on the effects of trade liberalisation, they point to modest gains at best (see Section 5.5 on agriculture, Section 6.5 on manufacturing, Tables 7.6/7 on services). It could be argued that dynamic effects (gains?) may be greater than such, mainly static, estimates, but there is limited concrete evidence supporting this view. Estimates in Chapter 4 on trade liberalisation gains are derived from the authors' previous work using dynamic simulations. Those are rigorous, drawing on state-of-theart technical advances, based on carefully thought out assumptions and questions raised seeking answers (nature of shocks, etc). But, inevitably, brought to bear on big questions like that of the title, such simulations are in for the Lucas critique (as the author(s) themselves acknowledge): Simulations are OK when a limited change to the structure is being considered (a 'shock'); would, however, a model's structure hold unaltered when a potential game-changer (a massive and multi-faceted 'shock'), such as an exit from the EU, occurs? Would elasticities, factor shares and intensities, and other such parameters, routinely taken as 'deep', 'structural' and constant, not change? If so, simulations and other technical welfare analyses of a possible entry into the Eurozone, or trade liberalisation, etc. offer rather tenuous information.

A bigger point emerges here: One is led to concur with Backhouse (2010, The Puzzle of Modern Economics, CUP) that Economics is better geared to answering specific, but, from a broader point of view, more focused questions, rather than answering 'big issues' (e.g., it helped design efficient telecoms auctions, but failed to predict the 2007-9 and other crises). Of course, big questions do not admit of precise answers; what one should aim for (I think in line with Backhouse's ideas) is reasonable arguments indicating clear directions - whether the gains be 'x%' or 'y%' is irrelevant or even misleading. The book's arguments stem from the basic tenets of neoclassical economics, i.e., free markets and free, unimpeded trade maximise economic welfare. The authors see no 'market failures' (such as some goods being of the nature of public goods, or the presence of externalities, natural monopolies, asymmetric information, and the like) that would justify departure from these principles. Trade protection, it is argued, is often motivated by rent-seeking rather than the objective of maximising welfare. These are the only (albeit strong) guiding principles, coupled with a belief that the EU is inexorably moving towards an ever-closer union with an (unspoken) aversion to joint decision-making. There is no consideration of the 'unknown unknowns': The rise of other economic areas and a possible challenge that this might present to the 'first world'; climate change and the challenge of shifting to sustainable resources; demographics and debt crises; migration and other political and security challenges for Europe. Many of these challenges involve important economic dimensions that the discussion, as framed in the book, ignores. There is much emphasis these days, even within mainstream economics, on the fact that cooperative solutions are superior exactly in cases of shared problems like those above, which would, otherwise, be likely to receive selfish, but inferior, solutions. The overarching question is whether global or regional challenges can best be met using national or pooled powers? The likely answer, i.e. shared decision-making, is hugely different to subservience to the EU, as projected by popular discourse (though the book never suggests that). With its simple prior guiding principles and technical focus, the book misses out on all these considerations.

Besides, not everyone shares the view that socially-minded regulations, like EU social and labour market protection regulations, are bad – for many people, they are, indeed, good, and help protect the UK labour force from being exposed to an otherwise ruthless labour market (as evidenced by the duality of labour markets and arrangements like the 'zero hours contracts', for example). Moreover, on the whole, the EU is more interested in fighting inequalities (social, regional) and promoting industrial regeneration than the UK seems to be. What we are encountering in the UK these days is a reminder of regional inequalities and the threat posed by industrial decay, with the impending bankruptcy of

Tata Steel and the possible loss of tens of thousands of jobs; we hear that UK politicians are lobbying the EU to allow anti-dumping tariffs so as to protect the industry from the unduly low prices of Chinese imports. In other words, not only is the EU's concern on such matters justified and greater than the UK's, but at a moment of crisis, the EU is part of the solution and not of the problem. Similarly, on the issue of general inequality, the EU shares many people's great concern, which is greater than that of either general mainstream economics or of the book, the implicit throughout emphasis of which is mainly on 'efficiency' and very little on 'equity'.

The question of membership, of course, involves even broader considerations: Despite numerous criticisms it can and does receive, paramount among which must be the democratic deficit, the European Union is a zone of economic prosperity, political liberty and social solidarity. Do we want this to break up? Are we sure of the alternatives that will emerge? If the break-up process of the EU begins, for the first time in its history, what is the 'economic atom', the smallest unit that will be immutable to break-up, that will emerge? If it is in the interests of the UK to be an independent, free-market economy, might that not also be the case for Scotland or Wales versus the UK? Might we, therefore, see a large number of regions breaking away from current conventional states? What will their number be? Who can guarantee stability in this possibly chaotic world? These considerations may lead one to suggest that wholehearted engagement and work towards reform, rather than exit, should be the way forward.

It was said (by the late UK Labour Party politician Denis Healey) that an economist is one who, asked about a telephone number, offers an estimate. That may be so more due to the nature of reality than because of the nature of economists, but this book offers quite a sophisticated estimate. As precise and clear an estimate as the jumbled and fuzzy section of human affairs called 'the economy' can offer, the book draws on a big pool of knowledge of all relevant, recent and rigorous advances concerning the subject. For any professional economist, no less interested than the general public, it is a joy to read and sharpen one's arguments whatever one's views. Ultimately, though, one is left with the impression that the real driver of the book's unambiguous conclusion, i.e., that exit from the EU is in the UK's interests, implies two preconceived tenets: that free, unimpeded trade and a free-market economy are the best creators of economic welfare; as those ideals are best pursued outside the EU, an exit is the best option. Technical, factual and other analyses are only providing limited support to this prior. If someone is completely agnostic, the results of Chapters 2-7 would perhaps suggest an exit but would not be enough to create strong convictions. Put it another way, due to priors, there is an implicit null hypothesis, namely 'exit is best', that technical results fail to disprove. I strongly suspect that, had the null been 'stay in', the evidence would have been sufficient to disprove that, as well.

So, one could discern three parts in the title question: firstly, 'what input can mainstream economics offer to the question of the UK's EU membership?', secondly, 'what are the broader considerations involved in the question and, particularly, their economic dimensions?', and, finally, what is a possible final answer?' A review of this book cannot but be concerned with all three questions. As far as the first question is concerned, taking neoclassical, mainstream economics as the reference framework, the book gets nearly full marks; it suggests that there are small to moderate (at best and not fully certain) gains to be had by the UK moving out of the EU and pursuing 'free' trade with the rest of the world, with trade relation between the UK and the EU left to be decided. Yet, the book fails to consider the second question, the broader questions and challenges mentioned above and their economic dimensions. We argued that there are reasons to support that the EU is a force for good on these matters, that co-operative solutions and joint decision-making may be the answer to regional and global questions. By framing the discussion in terms of seeking precise and, therefore, narrow estimates of free trade gains outside the EU, and the like, the book misses out on such considerations. As a result, when it comes to the third question on the final verdict, readers are left with considerable food for thought before they can make up their own minds.

# Book Review



# Full Industry Equilibrium A Theory of the Industrial Long Run

by Arrigo Opocher and Ian Steedman Published by Cambridge University Press, Cambridge, United Kingdom, 2015, pp. 232. ISBN: 978-1-107-09779-7

reviewed by Lefteris Tsoulfidis\*

Opocher and Steedman, two influential economists working within the modern classical economic theory, attempt to combine two competing approaches to microeconomics, namely, the neoclassical long-run theory of the firm and the Sraffa-inspired classical version of economics. Both approaches share some common ground and their cross-fertilisation could, arguably, be profitably utilised to develop more sensible microeconomic principles leading to firmer theoretical conclusions of practical significance. Such a synthesis of two approaches, hitherto developing parallel to each other, has not been tried so far and, in this sense, the book is characterised by originality. It is important to stress at the outset that the potential for synthesis lies only in the production or supply (not consumer demand) side of the economy, where the authors lay out the implications of a rigorous long-period point of view. They show, in particular, that there is, generally, no presumption made to the effect that demand for a factor is inversely proportionate to this factor's price, as is commonly assumed in the neoclassical microeconomic analysis currently prevailing in economics classrooms.

The process through which the authors seek to achieve such a synthesis is the method of comparative statics analysis, according to which one starts off with a state of equilibrium and then hypothesises an exogenous change to a variable, such as, for example, real wage (or profit rate), taxation, terms of trade, price of a strategic input (e.g. the price of oil), productivity (total factor), etc. Then the object of study becomes the movement of all relative prices, which, following the aforementioned exogenous event, must change in such a way so as to be consistent with arriving at a situation of zero maximum profits earned by all industries. The authors describe this situation using the Wicksellian term of 'full industry equilibrium' (FIE). The question at issue concerns predicting the size and

\* Professor, Department of Economics, University of Macedonia, Thessaloniki, Greece.

the sign of such price changes by taking into account the inter-industry structure of the economy and the attainment of the final equilibrium status, where there are no net (excess economic) profits. It is noteworthy that FIE differs from neoclassical general equilibrium, since it does not assume consumer preferences or demand decisions associated with them. Opocher and Steedman are two authors with significant contributions to both neoclassical (from a critical perspective) and classical theories and, as such, they are well equipped to also make a meaningful contribution towards bridging these two approaches. To this end, they argue that both approaches would benefit from mutually recognising that they are built upon the key concept of FIE. More specifically, the authors argue that what both classical and neoclassical approaches share is the long-run method of analysis in production. The study is further simplified by stipulating firms' homogeneity and by leaving aside complications arising from economies of scale external or internal to the firm or industry. If we find any difference in the approach of the book from the usual Sraffian one, it is that the focus is on individual industries rather than the totality of industries comprising a national economy. The book focuses on "micro" productive choices and often assumes a zero interest rate. Moreover, in order to make the approach comparable to neoclassical microeconomics, the authors use the standard U-shaped average cost curves, although these are precisely the kind of curves that Sraffa, in his famous (1926) article, found to be inconsistent with the neoclassical requirements of a perfectly competitive firm, while subsequent empirical research at Cambridge in the 1930s showed that in manufacturing, at least, cost curves are mainly characterised by constant returns to scale. Opocher and Steedman seem to apply the methodological principle, i.e., that the worse the circumstances, the stronger the argument used becomes. It is important to note that the neoclassical microeconomic theory, unlike macroeconomics, has pretty much remained the same for decades. In effect, microeconomics, as it emerged in the 1930s, mainly as a result of Sraffa's (1926) article, and as formulated in the 1970s, has not essentially changed to date.

The book is divided into ten chapters, all connected to each other and all aspiring to attain the authors' goal, i.e., to contribute to the formulation of a new and broader, long-run microeconomic approach derived from the cross-fertilisation of two microeconomic approaches moving in parallel to each other. Chapter 1 is about preliminaries and explanations of the FIE, its properties and its connections with both the neoclassical microeconomics and the Sraffa-inspired economic approach.

Chapter 2 presents the neoclassical long-run theory of the firm and explains the underlying structure of relative prices. Some further complications are added to this model, such as the presence of produced (as opposed to primary) inputs and discontinuities in profit-maximising input use. Furthermore, some unexpected behaviour is identified in the movement of the capital-output ratio with changes of the rent-wage rate. The analysis is generalised from single production to more realistic and harder to theorise multiproduct or joint production industries.

The details of the latter more complex case of production are taken up in Chapter 3, where it is shown that, in most cases, joint production may be thought of as a generalisation of the single product case, when we simply assume the presence of a single input. In the simplest formulation, this case bears similarities to the many inputs and the production of single products one.

In Chapter 4 the firm in the industry is placed in competition with other similarly motivated firms over produced and non-produced inputs. The analysis is carried out under the oxymoron scheme of maximum net profits equal to zero. Hence, there is no distinction between profits and the rate of profits, that is, profits estimated on a basis such as invested capital. Apparently, this is another methodological concession made to the standard neoclassical theory, according to which the mere presence of profits is sufficient to attract an inflow of firms until profits become equal to zero. In similar fashion and in an attempt to facilitate synthesis, the authors make frequent use of twice differentiable cost functions and they also rule out possible complementarities between inputs. In other words, the cost functions used in the analysis are assumed to be "well behaved". All of the above is done in an attempt to facilitate analysis and make the synthesis of the two approaches easier for economists of both approaches to follow. Under these assumptions, two types of shocks are considered: (a) a change in the price of a primary input; (b) a change in the rate of interest. Competition (perfect) leads to changes in the structure of relative commodity prices and, as far as the prices of primary inputs are concerned, it is possible to derive qualitative restrictions within the framework of conventional partial equilibrium analysis. Furthermore, commodity prices react differently in the face of external shocks.

Chapter 5 relaxes the assumption of the well-behaved functions and focuses on primary input use per unit of gross output. The analysis is carried out at the industry level by taking into account inter-industry connections and a positive interest rate. The theoretical analysis suggests that employment coefficients and real wage can be positively correlated. This chapter is supposed to share common ground with the famous capital theory controversies and with the Sraffa-based critique in general, but these connections are not sufficiently explicit. The focus of the chapter is on FIE and the results of comparative statics exercises and not on possible connections to capital theory controversies.

Chapters 6, 7 and 8 extend and further elaborate on the first five chapters, as they examine key applied microeconomic questions, such as price effects of the introduction of a tax, changes in terms of trade, as well as effects of technological change. The answers are often surprisingly different from those derived in standard microeconomics; what is more important is that these answers are derived from a theorisation of the operation of total economy. In Chapter 6 the authors study what the effects on prices are when cost is, for some reason, reduced. The analysis is further concretised in Chapter 7, where price effects of taxation within an inter-industry setting are examined and certain regularities are singled out. The effect of sales taxation on primary inputs is magnified when the presence of produced inputs is accounted for. A uniform sales tax rate is expected to

have quite dissimilar effects on prices; however, we may find a neutral sales tax when it is imposed in proportion to the added value of industries. Not surprisingly, a higher sales tax imposed on an industry does not necessarily imply a higher relative price, because everything depends on the use of the taxed commodity by other industries and the various linkages between industries, not to mention the proceeds of taxation and the way in which these are spent by the government.

Chapter 8 is about the effects of (total factor) productivity on prices, which could be quite similar to those of taxation. However, there is an important difference in that, while taxation is directly observable (and, therefore, it can become a policy variable), productivity is not. The total factor productivity (TFP) growth rate in neoclassical economics is residually determined after subtracting the growth rates of normally employed factor inputs (weighted by their income shares) from output growth rate. The advantage of the FIE approach is that it breaks down the technical change measured by TFP into average primary input prices and distribution. Moreover, FIE price accounting enables identification of trend components of productivity in each individual industry. Lastly, the change in the average primary input price may be broken down to industry productivity change per se and to changes due to other industries. All of these are important in their own right and shed new light on discussions of TFP. Chapter 9 adds a retrospective and prospective view to the whole book and approach. Chapter 10 contains a summary and makes some concluding remarks about possible further research questions.

It is expected that the analysis and findings of the book will attract the interest of neoclassical and modern classical economists who will further use and expand FIE in order to tackle old and new microeconomic questions. The book will be useful to those taking intermediate or advanced courses in microeconomics or economic theory and to those interested in a rigorous formulation of economic problems and answers derived from such a formulation, which can shed further light on controversial issues in economic theory as well as use these answers and results for planning more effective economic policies. Finally, the book might also be useful to those interested in the history of economic thought and developments that have taken place, particularly those of the 1970s.

# call for papers



20<sup>th</sup>Anniversary 1996-2016

Association of Economic Universities of South and Eastern Europe and the Black Sea Region



Anadolu University Faculty of Economics and Administrative Sciences

# 12<sup>th</sup> INTERNATIONAL CONFERENCE of ASECU

organized by

Anadolu University Faculty of Economics and Administrative Sciences

# INCLUSIVE AND SUSTAINABLE DEVELOPMENT AND THE ROLE OF SOCIAL AND SOLIDARITY ECONOMY

September 29-30, 2016 in Eskişehir, Turkey

New Deadline for receipt of abstracts: March 14, 2016 Deadline for receipt of full papers: June 12, 2016

http://asecu2016.anadolu.edu.tr or www.asecu.gr

# call for papers

#### **GENERAL INFORMATION**

We welcome submissions addressing the ASECU 2016 International Conference theme:

## "INCLUSIVE AND SUSTAINABLE DEVELOPMENT AND THE ROLE OF SOCIAL AND SOLIDARITY ECONOMY"

12th International Conference of ASECU, is organized by Anadolu University, on the 29<sup>th</sup>-30<sup>th</sup>, September, 2016 in Eskişehir, Turkey, on the occasion of the 20<sup>th</sup> Anniversary of the Association.

The conference aims to provide a professional discussion platform for both academic and business professionals from Southern and Eastern Europe as well as the Black Sea Region, and from the neighbouring countries.

The conference will address inclusive and sustainable development and the role of social and solidarity economy (SSE). SSE, is a phenomenon that has gained growing economic, social, political visibility in recent times, and that in practice allows us to think about actions that envisage the aspects making up the integrated approach to development.

The SSE, defined as a concept including organizations and enterprises producing goods, services and knowledge while pursuing social and economic objectives, emerges as a response to the need for innovation within the current model of production and consumption. During recent years a great deal of legislation has been adopted concerning SSE.

This legislative activity and institutional recognition of the SSE on an international level is characterized by efforts to incorporate economically dynamic and socially innovative aspects into the joint construction of public policies that aim to promote greater cohesion and inclusion.

Hence, organizers of the conference especially expect participants to explore and examine the impact of the SSE on inclusive and sustainable development.

The conference official language is English.

#### **CONFERENCE SUB-TOPICS**

1. Socioeconomic and Environmental Sustainability: holistic, balanced development, integrating environmental, social, and economic aspects, considering different sectors and recognizing their inter-linkages. Special issue: utilizing the advantages of technology - entrepreneurship in the digitized ecosystem.

2. Towards an alternative socio-economic model and institutional framework, based on the further evolution of the post-modern conceptual paradigm the values and principles - (re)introducing mutualism and cooperation, equity, social welfare, social and economic democracy, pluralism and diversity, for the "creation of a better world". Necessary innovations in education, ethics and ethos, integrating social entrepreneurship and sustainability.

**3.** Innovative, progressive approaches for the introduction of participatory functionality, allowing for the promotion of social cohesion and contributing to an increase of a region's social capital.

**4.** Concerns about the "big picture". Socio-economic re-evolution with respect to wider issues of humanity: climate change; aging, population explosion and demographic issues; unemployment, inequality and poverty; misuse of children, women and special social groups, trafficking; environmental integrity and sustainability.
### call for papers

#### **IMPORTANT DATES**

March 14, 2016	: New Deadline for receipt of abstracts
June 12, 2016	: Deadline for receipt of full papers and application forms
July 3, 2016	: Notification for full-paper acceptance
July 10, 2016	: Deadline for registration and payment
	(there is no returning payment in case of cancellation)
August 1, 2016	: Announcement of the preliminary conference program
September 19, 2016	: Announcement of the detailed conference program
	(at http://asecu2016.anadolu.edu.tr/)

#### **CONFERENCE COMMITTEES**

#### **SCIENTIFIC Committee**

EŞKİNAT Rana Anadolu University, Turkey (Chair of the Scientific Committee) AÇIKALIN Sezgin Anadolu University, Turkey

AFŞAR Muharrem Anadolu University, Turkey

BERBEROĞLU Necat Anadolu University, Turkey

**BİÇERLİ Kemal** Anadolu University, Turkey

**BODYAGIN Oleg** Rostov State University of Economics, Russia

**CEROVIĆ Bozidar** University of Belgrade, Serbia

**ÇAKMAK Erol** TED University, Turkey

**DAĞDEMİR Elif** Anadolu University, Turkey

**DAĞDEMİR Özcan** Osman Gazi University, Turkey

**DAMYANOV Atanas** "D. Tsenov" University of Svishtov, Bulgaria

**DEMİR Gülten** Marmara University, Turkey

DÖNMEZ Recai Anadolu University, Turkey

**ERLAT Güzin** METU, Turkey **ERLAT Haluk** 

METU, Turkey FOUNTAS Stylianos

University of Macedonia, Greece

GEREK Sevgi Anadolu University, Turkey KARADZIĆ Vesna University of Montenegro, Montenegro KAYA Ayten, Ayşen Ege University, Turkey

KILIÇASLAN Yılmaz Anadolu University, Turkey

KUTLU Erol Anadolu University, Turkey

**MEMAJ Fatmir** University of Tirana, Albania

MIKEREVIC Dejan University of Banja Luka, Bosnia Herzegovina

NAKOV Leonid "Ss Cyril and Methodius" University, FYROM

ÖZER Mustafa Anadolu University, Turkey

PARASCHIV Dorel Bucharest University of Economic Studies, Romania

SUKLEV Bobek "Ss Cyril and Methodius" University, FYROM

**ŞIKLAR İlyas** Anadolu University, Turkey

TABAN Sami Osmangazi University, Turkey

**TONUS Özgür** Anadolu University, Turkey

**TSEKOURAS Yannis** University of Macedonia, Greece

WIERZBISNKI Bogdan University of Rzeszow, Poland

YILDIRIM Kemal Anadolu University, Turkey

**ZAROTIADIS Grigorios** Aristotle University of Thessaloniki, Greece

## call for papers

#### **ORGANIZATION** Committees

#### Executive Committee

EŞKİNAT Rana, Chair DÖNMEZ Recai, Dean of Faculty of Economics and Administrative Sciences BERBEROĞLU Necat, Director of Dept. of Economics AFŞAR Muharrem İSLATİNCE Hasan KILIÇASLAN Yılmaz TEZCAN Mediha TEPECİK Filiz YAZICI Ayla

#### Local Arrangements

AKTAŞ Tuba ÇATALBAŞ Nazım İSLATİNCE Hasan KOSTAKOĞLU S. Fatih TEPECİK Filiz TEZCAN Mediha YAZICI Ayla

Press and Communication

**DOĞAN** Burhan

**ERDOĞAN** Meltem

**KOYUNCU** Fatma

**DURAL** Betül

Conference Secretariat

AYTUN Uğur BARI Bilgin GELDİŞEN Ayşen KÜÇÜKSAKARYA Sevilay TEPECİK Filiz

#### **PROGRAMME** Committee

AÇMA Bülent ADAÇAY Funda Rana BARI Bilgin BERBEROĞLU Bahar ÇEKEROL Kâmil ERDINÇ Zeynep ERDOĞAN Levent ESEN Ethem İSLATİNCE Hasan KÜÇÜKSAKARYA Sevilay ÖZATA Erkan ÖZDEMİR B. Kağan ÖZSOY Ceyda SUBAŞI Meri TEPECIK Filiz TEZCAN Mediha TIRYAKİ Ahmet TOSUNOĞLU Tuğberk ULUKAN Bengül Gülümser YAZICI Ayla

#### GUIDELINES FOR ABSTRACT SUBMISSION (All submissions are refereed)

- Submissions must not have been published, submitted or presented at other conferences.
- Title of the paper written in English (type size 16, font Times New Roman, bold)
- The title page must include names, degree, affiliations, complete addresses (e-mail, telephone) for all authors
- Abstracts, written in English, should be no longer than 10-18 lines (format: Times New Roman 11 italic, Spacing: Before: 0, After: 0, Line Spacing: 1.5, Alignment: Justified, Indentation: Left: 0, Right: 0, Special: 0). JEL classification and up to 6 keywords should be used.

Detailed information concerning abstracts' submission can be found at <u>Abstract</u> <u>Submission</u> on the <u>asecu2016.anadolu.edu.tr</u>

#### **GUIDELINES FOR FULL-PAPER SUBMISSION**

- The full paper should be a maximum of ten (10) pages including references and exhibits. Requirements for the paper formatting: Page: A4; Font: Times New Roman, Size: 11 pt; multiple line spacing (1,15); Page Setup: Margins: Left: 3; Right: 3; Top: 4; Bottom: 3.
- In the first mandatory page (presentation page) you insert the title of the paper (Center, times new roman, 16 point, bold). Leave 1 blank line (16 point) between the titles. Authors' names (Center, times new roman, 12 point). Affiliations such as Institution/Department, City, Country (Center, times new roman, 10 point) and e-mail of the corresponding authors.
- Headings: enumerate Chapter Headings by Arabic numbers (1., 2., etc.). First level Chapter Headings use all caps (Times New Roman, 11). Subchapter headings are font 11, italic, bold and will follow the enumeration of the previous heading (1.1., 1.2., etc.).

More information about full-paper submission are given at <u>Full-paper Submission</u> at the <u>asecu2016.anadolu.edu.tr</u>.

All <u>the presented papers</u> will be included at the Proceedings of the Conference, which will be issued, with an ISBN nr in electronic form. The authors will have a certain time after the conference for submitting the revised version of their papers.

The Editorial Board of South Eastern Europe Journal of Economics (SEEJE), the official journal of ASECU, after having the agreement of the authors, plans to publish selected conference papers following a blind refereeing process. More information at <a href="http://www.asecu.gr/Seeje/">http://www.asecu.gr/Seeje/</a>

Paper presented at the Conference ASECU2016 will have also the opportunity to be published at the Anadolu University Journal of Social Sciences (indexed in Econ Lit). Further details for the journal <u>http://sbd.anadolu.edu.tr/home.html</u>

The papers for both journals will follow the usual procedure of evaluation and the journals will be under no commitment to publish any article. However, they do promise an expedited refereeing process.

## call for papers

#### **CONFERENCE FEE AND HOTEL ACCOMMODATION**

The conference fee amounts to 125 EURO per participant (for all participants).

Account name: 12<sup>th</sup> Inter.Conference of ASECU2016

Account owner: Anadolu University, Genclik ve Spor Kulubu

Address: Anadolu Universitesi 26470 Eskisehir / Turkiye (Turkey)

Bank: VakifBank; Branch: Vakifbank Anadolu University Branch

SWIFT Code (BIC): TVBATR2A

IBAN (for Euro): TR21 0001 5001 5804 8015 8281 38

IBAN (for Dollar): TR13 0001 5001 5804 8015 8281 85

IBAN (for Turkish Lira): TR25 0001 5001 5800 7303 9908 52

The payments should be clearly mentioned by the note: <u>ASECU 2016</u>, as well as <u>the</u> <u>Name and Surname of Participant</u>.

All prices are per person. Conference material as well as lunches, dinner, refreshments during the conference are included in the registration fee.

Registrations could be accepted also at the registration desk during the Conference. In this case the fee will amount 150 EURO.

Receipts will be prepared for the payments with the full name, university name and the country of the participant. In case that specific information is needed to be mentioned at the receipt, the participant is obliged to inform the Secretariat accordingly

Payment of the registration fee should be made by bank transfer. Payments by credit cards will not be accepted.

Each participant should cover travel costs to and from Eskisehir, and should directly book accommodation in Eskisehir (Accommodation is not included in the registration fee).

The following hotels are recommended by the Organizing Committee (list of hotels close to conference venue):

- Albatros Hotel: This three star hotel is also located next to the university campus, 15 min. walking distance to the conference hall at the university campus, 20 min. walking distance to the train station, and 10 min. to the intercity bus station by tramway. For further information: http://www.esalbatroshotel.com/
- **Dedepark Hotel:** This four star hotel is also located in the city center. It is within 20 minutes walking distance (or 5 min. by tramway) to the conference hall at the university campus, 15 minutes walking distance to the train station, and 10 minutes distance to the intercity bus station by tramway. For further information: <u>http://dedepark.com/</u>
- **Divan Express Eskischir:** This four star hotel is also located in the city center. It is within 20 minutes walking distance to the conference hall at the university campus (or 5 min. by tramway). The train station is within 15 minutes walking distance. It takes 20 minutes to the intercity bus station by tramway. <u>http://www.divan.com.tr/</u>
- **IBIS Hotel:** This three star hotel is also located in the city center. It is within 20 minutes walking distance to the conference hall at the university campus (or 5 min. by tramway). The train station is within 15 minutes walking distance. Further information: http://www.ibishotel.com/gb/hotel-6567-ibis-eskisehir/index.shtml



- **Namli Otel:** This three star hotel is located in the city center. It is within a 30 minutes walking distance to the conference hall at the university campus (by bus it takes 15 minutes). It is within a walking distance to the train station and it takes 20 minutes to get to the intercity bus station by tramway. For further information: http://www.namliotel.com.tr/tr/eskisehir.asp;
- **Sör Otel**: This three star hotel is located within a 10 minute walking distance to the conference hall at the university campus, 20 minute walking distance to the train station, and it takes 10 minutes to the intercity bus station by tramway. For further information see: <u>http://www.sorotel.net/</u>
- **The Merlot Hotel:** This four star hotel is also located in the city center. It is within 20 minutes walking distance to the conference hall at the university campus (or 5 min. by tramway). The train station is within 15 minutes walking distance. It takes 20 minutes to the intercity bus station by tramway. For further information: <u>http://www.merlot.com.tr/</u>

#### **CONTACT INFORMATION**

Organizing Committee of the 12<sup>th</sup> International Conference of ASECU

Anadolu University Faculty of Economics and Administrative Sciences Eskişehir, 26470, Turkey

Fax: +90 222 335 05 95 Tel: +90 222 335 05 80 Secretariat of the Conference: asecu2016@anadolu.edu.tr

Technical support: ftepecik@anadolu.edu.tr (TEPECİK Filiz)

## refereeacknowledgment

The *South Eastern Europe Journal of Economics* gratefully acknowledges the continued support of the distinguished members of the Editorial Board and the following roster of economists who have contributed their services as referees during the period Fall 2012 - Spring 2016.

Abeliotis Konstadinos, Harokopio University, Greece Adnett Nick, Staffordshire University, United Kingdom Arghyrou Michael, Cardiff Business School, United Kingdom Aristotelous Kyriacos, Otterbein College, USA Ayadi Felix, JP Morgan Chase Center for Financial Education, USA Bampinas Georgios, University of Macedonia, Greece Bayar Yilmaz, Usak University, Turkey Bejaković Predrag, Institute of Public Finance, Croatia Bernet Beat. University of St Gallen, Switzerland Castellano Rosella, University of Macerata, Italia Christopoulou Revekka, University of Macedonia, Greece Chung Richard, Griffith University, Australia Constantatos Christos, University of Macedonia, Greece Constantelou Anastasia, Aegean University, Greece Ćurak Marijana, University of Split, Croatia

Danopoulos Constantine, San Jose State University, USA Desli Evangelia, Aristotle University of Thessaloniki, Greece Elekdağ Selim, International Monetary Fund, USA Fauziah Wan Yusoff, University Tun Hussein Onn, Malaysia Fountas Stilianos, University of Macedonia, Greece Fouskas Konstantinos, University of Macedonia, Greece Gillet-Monjarret Claire, University of Montpellier, France Ginoglou Dimitrios, University of Macedonia, Greece Gogos Stylianos, Athens University of Economics and Business, Greece Haiss Peter. Vienna University of Economics and Business Administration, Austria Ioannides Stavros, Panteion University, Greece Kampas Athanasios, Agricultural University of Athens, Greece Kanellopoulos Costas, Bank of Greece, Grrece Karagianni Roxani, Centre for Planning and Economic Research, Greece Karras George, University of Illinois at Chicago, USA Kbat-Rudnicka Danuta, Crakow University of Economics, Poland Kılınç Mustafa, Central Bank of the Republic of Turkey, Turkey Kollias Christos, University of Thessaly, Greece Korotayev Andrey, Russian State University for the Humanities, Russia Kostopoulou Stella, Aristotle University of Thessaloniki, Greece

South-Eastern Europe Journal of Economics 1 (2016)

Koukouritakis Minoas, University of Crete, Greece Kottaridi Constantina, University of Piraeus, Greece Kuznar Andzelika, Warsaw School of Economics, Poland Lee M Sang, University of Nebraska, USA Lianos Theodore, Athens University of Economics and Business, Greece Liuhto Kari, University of Turku, Finland Makris George, University of Macedonia, Greece Margaritis Dimitris, University of Auckland, New Zealand Markovic Ivan, University of Nis, Serbia Mattas Kostadinos, Aristotle University of Thessaloniki, Greece Mesjasz Czeslaw, Cracow Economic University, Poland Mikropoulou Christina, University of Macedonia, Greece Miniaoui Hela. University of Wollongong in Dubai, United Arab Emirates Murphy Eithne, National University of Ireland, Ireland Mylonidis Nikolaos, University of Ioannina, Greece Nikolaou Dimitrios, Illinois State University, USA O'Donnell Owen, University of Macedonia, Greece Palivos Theodore, Athens University of Economics and Business, Greece Panagiotidis Theodore, University of Macedonia, Greece Partalidou Maria, Aristotle University of Thessaloniki, Greece

Papadogonas Theodore, Technological Educational Institute of Chalkis, Greece Patsiaouras Georgios, University of Leicester, United Kingdom Pervan Maja, University of Split, Croatia, Pseiridis Anastasia, Panteion University, Greece Polychronopoulos George, Technological Educational Institute of Athens, Greece Savva Christos, Cyprus University of Technology, Cyprus Schneider Friedrich, University Johannes Kepler, Austria Sekeres Vojislav, University of Belgrade, Serbia Sideris Dimitrios, Panteion University, Greece Siriopoulos Costas, University of Patras, Greece Spathis Charalambos, Aristotle University of Thessaloniki, Greece Triguero Cano Ángela, University of Castilla-La Mancha, Spain Trimi Silvana. University of Nebraska, USA Tsounis Nikolaos, Technological Educational Institute of West Macedonia, Greece Vlahopoulou Maro, University of Macedonia, Greece Williams Colin, University of Sheffield, United Kingdom Ziarko-Siwek Urszula, Poznan University of Economics and Business, Poland

# guidefor authors

The articles should be written as follows: (1) Papers must be in English.

(2) Papers for publication (two copies) should be sent to:

Mrs. Melina Petromelidou Editorial Secretary South-Eastern Europe Journal of Economics ASECU, University of Macedonia, 156, Egnatia Str., 546 36 Thessaloniki, Greece

The *Journal's* phone number is (+30) 2310891793, fax: (+30) 2310891748 e-mail: asecu@uom.edu.gr

Submission of a paper will be held to imply that it contains original unpublished work and is not being submitted for publication elsewhere. The Editor does not accept responsibility for damage or loss of papers submitted. Upon acceptance of an article, author(s) will be asked to transfer copyright of the article to the publisher. This transfer will ensure the widest possible dissemination of information.

(3) Papers will be considered in any form, but authors of papers accepted for publication will be expected to provide a final copy conforming to the general style of the Journal as outlined in notes 4 through 13 below.

(4) Manuscripts should be 1,5 spaced, with wide margins, and printed on one side of the paper only. All pages should be numbered in sequence. Titles and subtitles should be short. References, tables, and captions for the figures should be printed on separate pages.

(5) The first page of the manuscript should contain the following information: (i) the title; (ii) the name(s) and institutional affiliation(s) of the author(s); (iii) an abstract of not more than 100 words. A footnote on the same sheet should give the name, address, and telephone and fax numbers of the corresponding author [as well as an e-mail address].

(6) The first page of the manuscript should also contain at least one classification code according to the Classification System for Journal Articles as used by the Journal of Economic Literature; in addition, up to five key words should be supplied.

The classification system used in JEL can be found at:

 $http://www.aeaweb.org/journals/jel\_class\_system.html.$ 

(7) Acknowledgements and information on grants received can be given in a first footnote, which should not be included in the consecutive numbering of footnotes.

(8) Footnotes should be kept to a minimum and numbered consecutively throughout the text with superscript Arabic numerals.

(9) Displayed formulae should be numbered consecutively throughout the manuscript as (1), (2), etc. against the right-hand margin of the page. In cases where the derivation of formulae has been abbreviated, it is of great help to the referees if the full derivation can be presented on a separate sheet (not to be published).

(10) References to publications should be as follows: 'Smith (1992) reported that...' or 'This problem has been studied previously (e.g., Smith et al., 1969)'. The author should make sure that there is a strict one-to-one correspondence between the names and years in the text and those on the list. The list of references should appear at the end of the main text (after any appendices, but before tables and captions for figures). It should be double spaced and listed in alphabetical order by author's name. References should appear as follows:

For monographs

Sen, A., 1970, Collective Choice and Social Welfare, San Francisco: Holden Day.

#### For contributions to collective works

Kornai, J., 1991, Stabilization and Economic Transition in Hungary: The Next Two Years, in J. de Melo and A. Sapir (eds.), *Trade Theory and Economic Reform: North, South and East*, Oxford: Basil Blackwell, 307-326.

For periodicals

Magdalinos, M., 1990, "The Classical Principles of Testing Using Instrumental Variables Estimates", *Journal of Econometrics*, 44, 241-279.

Note that journal titles should not be abbreviated.

(11) Illustrations will be reproduced photographically from originals supplied by the author; they will not be redrawn by the publisher. Please provide all illustrations in quadruplicate (one high-contrast original and three photocopies). Care should be taken that lettering and symbols are of a comparable size. The illustrations should not be inserted in the text, and should be marked on the back with figure number, title of paper, and author's name. All graphs and diagrams should be referred to as figures, and should be numbered consecutively in the text in Arabic numerals. Illustration for papers submitted as electronic manuscripts should be in traditional form.

(12) Tables should be numbered consecutively in the text in Arabic numerals and printed on separate sheets.

(13) Accepted papers should be submitted in electronic form in a storage media (i.e. CD, DVD, USB, etc.) with accompanying manuscript. Electronic manuscripts have the advantage that there is no need for re-setting of text, thereby avoiding the possibility of introducing errors and resulting in reliable and fast delivery of proofs. The preferred format is either .doc or .docx. Make absolutely sure that the file on the disk and the printout are identical.

(14) Page proofs will be sent to the corresponding author. Proofs should be corrected carefully; the responsibility for detecting errors lies with the author. Corrections should be restricted to instances in which the proof is at variance with the manuscript. There are neither submission fees nor page charges.